#### SYSTEMS COMMUNICATIONS MODULE

	<b>H</b> I	nnan	aanaa	<b>uuu uuu</b>
n	<b>д д</b>	<sup>а</sup> щ	<b>ti</b> 1	
uu n n n n		Ħ	<b>1</b> 1 11	а цал п
a	a	ш	H H	n n
anana	. <b>д</b>	annua	<b>HHHHHH</b>	<b>11 11</b>

EURROUGHS CORPORATION COMPUTER SYSTEMS GROUP SANTA BARBARA PLANT

E1800/B1700 Series Systems

Software Operational Guide

Preliminary

:

:

:

1

•

:

:

. . . . . .

:

:

Company Confidential

13 September 1978

Title: SYSTEMS COMMUNICATION MCDULE Identity: Software Operational Guide Burroughs Corporation believes the information described in this document to be accurate and reliable, and much care has been taken in its preparation. However, the Corporation cannot accept any responsibility, financial or otherwise, for any consequences arising out of the use of this material. The information contained herein is subject to change. Revisions may be issued from time to time to advise of such changes and/or additions. Release: Level Number - 7.0 Revision Level - Original Date of Release - 13 Sep 1978 Date of Printing - 12 Sep 1978 I/D Number - Preliminary

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page – iii
Preliminary Documentation	(Original Version)
TABLE OF CONTENTS	

## PREFACE

MANUAL	08.	JECI	TVES	AND	RE	. A I	ÉE⊁	<b>{</b> }	AS	SUM	191	r' <b>i</b> o	) N S	<b>.</b>	•	•	•	•	•	•	•	•.	•	1
STRUCTU	JŔE	0F	DOCUN	MENT	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	2

## SECTION I

## INTRODUCTION TO SYCOM

GENERAL DESCRIPTION	5
SCOPE AND OBJECTIVES	6
FUNCTIONS PERFORMED	6
BASIC FEATURES	7
LINE THROUGHPUT	8
ERROR DETECTION AND RECOVERY	8
LIMITATIONS	9
NOTICE	9
ASSUCIATED DOCUMENTATION	10

## SECTION II

## CONCEPTS AND FACILITIES

GENERAL DESCRIPTION	12
THE SYSTEM CONFIGURATION	12
SYSTEM COMMUNICATIONS	14
Point-to-Point Switched Network	14
Point-to-Point Non-switched Network	16
Communications Procedure	17
User System - Server System Relationship	17
SYCOM AND ITS ENVIRONMENT	19

31800/B1700 SYSTEMS COMMUNICATIONS MODULE	12	Sep 1978
Software Operational Guide	Page	; - iv
Preliminary Documentation	(Original	/ersion)
TABLE OF CONTENIS (continued)		
HARDWARE REQUIREMENTS		• 19
MEMORY REQUIREMENTS		. 20
SOFTWARE REQUIREMENTS		- 20
Master Control Program (MCP)	· • • • • • •	• 21
SYSTEM/MAKEUSER Program 4	• • • • • •	• 21
DMPALL Program	• • • • •	. 21
Network Controller program	• • • • •	• 22
SOFTWARE OPTIENS	• • • • • •	• 22
Supervisory Message Control System (SMCS)	• • • • • •	• 23
Application Programs	• • • • • •	• 23
CANDE Program		• 24
FUNCTIONAL DESCRIPTION		• 24
STANDARD RJE FUNCTIONS	· · · · · · · · · · · · ·	• 25
FILE TRANSFERS	• • • • •	. 26
Maintaining Integrity of File Characteris	tics	. 28
VIRTUAL TERMINALS	• • • • • •	. 29
Terminal-to-Program Communications	• • • • • •	. 29
the user system		. 32
the server system	• • • • •	. 33
Remote Supervisory Console (RSC)	• • • •	• 34
Program-to-SYCOM Communications	• • • • • •	. 37
PROGRAM-TO-PROGRAM COMMUNICATIONS		• 39
NETWORK CONTROL		- 40
Conversational Communications Procedure .		- 4C
Point-to-Point Contention Procedure	• • • • •	• 41
Transmission of Packed Data		. 42

B1800/B1700 SYSTEMS COMMUNICATIONS	MODULE			12	Sep 1	978
Software Operational Guide	•		•	Pac	]e -	• •
Preliminary Documentation	• •		(Origi	nal	Versi	on)
TABLE OF CONTENTS (continued)						
NETWORK CONTROLLER	• • • •	• •	• • • •	• (	•	43
CONFIGURATIONS AND SOFTWARE ASSOCI	ATIONS.	• •		• .	• •	44
Entry Level System	• • • . •	• •	• • • •	•	• •	44
Extended Level System		• •		•	• •	46
Ultimate Level System	· · · ·	• •		•	• •	48
MESSAGE PATH ASSIGNMENTS	• • • •	• •	• • • •	•	• •	50
For the Entry Level System		• •	• • • •	•	• •	50
For the Extended Level System .		• •	• • • •	•	• •	54
For the Ultimate Level System .	• • • •	• •	• • • •	•	• •	55
QUEUE DEPTH CONTROL			• • • •	•	• •	58
SYSTEM FILES		•		•	•	59
MASTER CONTROL PROGRAM (MCP) FIL	ES	• •	• • • •	•	• •	59
(SYSTEM)/USERCODE File						59
NETWORK CONTROLLER FILES				_		60
Network Information (NIE) File						60
CVCAM ETIES	••••					6.0
			• • • •	•	• •	
Peripheral Files	• • • •	• • •	• • • •	•	• •	6.0
Remote Files	• • • •	• • •	• • • •	•	• •	61
Queue Files	• • • •	• • •	• • • •	•	• •	62
RESTRICTIONS	• • • •	• • •	• • • •	) <b>•</b>	•. •	63

## SECTION III

## COMMANDS

GENERAL .	•	•	•	•	•	•	•	•	•	•	••	•	•	•	•	•	• •	•	•	•	•	•	•	•	•	65
CUMMANDS.	•	•	• .	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	65

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page – vi
Preliminary Documentation	(Original Version)
TABLE OF CONTENTS (continued)	
COMMAND SYNTAX CONVENTIONS	
Notation used in model format	
Punctuation Rules	67
Command Related Error Messages	63
COMMAND SET	
Terminal Commands	68
• ABORT• • • • • • • • • • • • • • • • • • •	69
•BYE••••••••••••	70
.CONNECT	71
.COPY	
DISCONNECT	76
•FETCH••••••••••••••••••••••••••••••••••••	77
•LIST • • • • • • • • • • • • • •	
•PUT•••••••	• • • • • • • • • 81
-USER	• • • • • • • • • 83
- WHAT	• • • • • • • • • • 84
* <smcs command="" string=""> (User Syste</smcs>	em)85
*? <smcs command="" mcp="" string=""> (User</smcs>	System)
<x><smcs command="" string=""> (Server S</smcs></x>	System)87
<x>?<smcs command="" mcp="" string=""> (Ser</smcs></x>	rver System) 88
System SPO/Frogram Commands	89
•CL • • • • • • • • • • • • • • •	• • • • • • • • • 90
•CP • • • • • • • • • • • • • •	• • • • • • • • • 91
• DEBUG••••••••••••••••••••••••••••••••••••	••••
•E0J•••••••••••••••	

B1800/	B1700	SY	S T E	MS	CC	4 M C	101	(10	: A 1	ΓIά	INS	5 M	100	UL	E							12	S	ep	1978
Softwa	re Ope	era	tio	na	i e	iui	ide	9							•							Pa	ge	-	vii
Prelim	inary.	Do	c น ก	en	tal	tic	on'											(	0r	<b>i</b> ç	, i n	al	Ň	er	sion)
TABLE	OF COM	NTE	NTS	()	c or	nt i	i nu	Jec	;) 														-		
System	SP071	° <b>r</b> o	gra	m I	Cor	nma	800	d s		cor	nti	inı	ièc	1)											
	.LIST	•	••	•	•	•	. •	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	94
-	.LOG.	•	• •	•	٠	•		•	•	•		•	٠	•	•	•	•	•	•	•	•	•.	•	•	96
	•LP •	•	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	•	. • ,	•	97
	•OFFL	INE	•	•	•	•	•	•	•	•	•	1. •	. •	•	. <b>•</b> •	•	•	•	•	•	•	•	•	•	98
. · ·	ONL I	NE	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	÷	•	•	•	•	•	99
	-PHON	Ε.		•	•	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	•	•	•	•	100
	•QS •	•	••	•	•	•	, •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	• .	•	101
	.READ	•	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	102
	.RY .	•	• •	•	•	•	•	•	•	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	•	103
	•SIZE	•	• •	•	•	•	•	•	•	•	•	•		•	•	•	•	•	•	•	•	•	•	•	104
	.STOP	•	••	•	. •	•	•	•	•	٠	•	. •	•	•	•	•	•	•	•	•	• 4	•	•	•	105
	.sv .	•	• •	•	•	•	•	•	•	•	•	•	•	•	• .	•	•	•	•	•	•	•,	•	•	106

## SECTION IV

## OPERATING PROCEDURES

INTRODUCTION	109
SYSTEM OPERATING PROCEDURES	109
PRELIMINARY OPERATIONS	110
Program Switches	110
Preparation of Parameters Card File	110
Setup MCP Program Name Table	111
Assigning Program Priorities	112
EXECUTION OF SYCOM * WITHOUT SMCS	112
EXECUTION OF SYCOM * WITH SMCS • • • • • • • • • • • • • • • • • • •	114

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page – viii
Preliminary Documentation	(Original Version)
TABLE OF CONTENTS (continued)	
COMMON SYSTEM OPERATIONS	••••• 116
SMCS Signal Characters	••••••
Entering the Parameters	•••••
SYCOM Commands From The SPD	•••••• 119
Systematic SYCOM Shutdown	••••• 120
Reactionary Hints and Pointers	••••••• 122
TERMINAL PROCEDURES	•••••• 123
TERMINAL FUNCTIONS	123
TERMINAL OPERATIONS (SYCOM CONTROLLED)	• • • • • • • 123
Example Interactive Terminal Session	123
TERMINAL OPERATIONS (SMCS CONTROLLED)	••••• 126
Example Interactive Terminal Session	126

# SECTION V

## GENERATING THE NETWORK CONTROLLERS

GENERAL	130
DETERMINING THE SPECIFICATIONS	130
OPERATIONAL PROCEDURES	131
Preparation of Input	131
Compilation Process	131
NDL Compiler Modifications	132
Print The USER/NCNIF File	133
SAMPLE NDL SOURCE FILE (LOCAL USER SYSTEM)	134
SAMPLE NOL SOURCE FILE (REMOTE SERVER SYSTEM)	138

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page - ix
Preliminary Documentation	(Original Version)
TABLE OF CONTENTS (continued)	

SECTION VI -

## APPENDICES

APPENDIX A: ERFOR MESSAGES	142
DESCRIPTION	142
Command Error Messages	143
Command Informational Messages	157
System Parameter Error Messages	158
System Warning Messages	160
System Condition Messages	163
System Fatal Messages	166
APPENDIX B: CHARACTER SET	167
APPENDIX C: RESERVED WORDS	168
APPENDIX D: COMMAND SUMMARY	169
APPENDIX E: COMMUNICATIONS CONTROL MESSAGES	171
GENERAL • • • • • • • • • • • • • • • • • • •	171
MESSAGE TYPES	171
MESSAGE FORMATS	172
Basic Data Communication Messages	172
System Control Hessage	173
SPO Messages	173
Card Reader/Punch Message	174
Line Printer Message	174
File Transfer Message	175
Virtual Terminal Message	175
File Separator Messages	175

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE		12	Sep	1978
Software Operational Guide		Pag	]e -	X
Preliminary Documentation	(Origi	nal	Vers	ion)
TABLE OF CONTENTS (continued)				
Program-to-Program Messages	• • • •	• (	• •	176
Program-to-SYCOM Messages	• • • •	•	• •	176
APPENDIX F: RESTRICTED MCP COMMAND SUMMARY .		•	•	177
Library Maintenance Instructions	• • • •	•	• •	177
Program Control Instructions	• • • •	•	• •	177
Program Control Instruction Attributes	• • • •	•	الدينية. إذا • 1 •	178
System Control Instructions	• • • •	1997 1997 1997	• •	178
Job Spawning Control Attributes	• • • •	•	• •	179
APPENDIX G: PARAMETERS FILE ATTRIBUTES	• • • •	•	••	180
Record Format	• • • •	•	• •	180
APPENDIX H: SUMMARY OF FILES		•	• •	182
APPENDIX I: SAMPLE APPLICATION PROGRAMS	• • • •	•	• •	183
Program-to-Program (User System)	• • •	n ₹	• •	183
Program-to-Program (Server System)	• • • •	•	• •	185

SECTION VII

GLOSSARY

TERMS	•		• •	•	•	• • • • •	í š	2 • •	• •	• •	• •	•	•	•	• •	•	•	•	187
ACRONYMS.	•	•	•	•	• .			i di ji	• •	• •	•	• •	•	•	•	•	•	•	202

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page - xi
Preliminary Documentation	(Original Version)
LIST OF ILLUSTRATIONS	

#### SECTION 11

#### CONCEPTS AND FACILITIES

Figure	11-1:	User	System	Conf	i gur a	tion.	• • •	• •	•	•	•	•	13
Figure	11-2:	Pcint	-To-Poi	nt S	witch	ed Neti	acrk .	•	•	•	•.	•	15
Figure	11-3:	Point	-To-Poi	nt N	on-sw	itched	Netwo	rk.	•	•	•	•	16
Figure	11-4:	User	System	- Se	rver	System	Schem	atio	•	•	• '	•	18
Figure	11-5:	Virtu	al Term	inal	s Sch	ematic	• • •	• •	•	•	•	●	30
Figure	II-6:	RSC I	nput/Ou	tput	Mess	age Flè	ом	• •	• •	•	•	•	35
Figure	11-7:	Progr	am-to-S	YCOM	Sche	matic.	• • •	. •	•	•	•	•	38
Figure	II-8:	Progr	am-to-P	rogr	am Sc	hemati	C	• •	• •	•	•	•	39
Figure	11-9:	User	System	Sche	matic	• • •	: • • •	•	•	•	•	•	45
Figure	11-10:	User	System	Sche	matic	(term	inals)	•	•	•	•	•	47
Figure	11-11:	User	System	Sche	matic	CSMCS	; term	inal	ls)	•	•	•	49

LIST OF TABLES

## SECTION III

#### COMMANDS

12 Sep 1978

Software Operational Guide

Page - xii

÷

(Original Version)

Preliminary Documentation

REVISION HISTORY

Original Release - September 1978

#### DOCUMENTATION NOTES

\*\*\*\*\* Do to the information available when this document Has 4 written, descriptions of functions and facilities for a 4 remote host system assumes a B1800/B17CO series system. When information becomes available for implementations of the Systems Communications Nodule (SYCOM) on "other" Burroughs systems, this Software Operational Guide will 4 be updated to reflect those capability differences that + + are provided by B4800/B4700 and B680C/B6700 systems. 4 + ŧ 4 Although this document describes the "Program-to-SYCOM" ٠ and "Program-to-Program" features, these capabilities ٠ are not yet implemented in this system release level. \* 

1

Software Operational Guide

Preface

(Original Version)

Page -

#### PREFACE

## MANUAL OBJECTIVES AND READER ASSUMPTIONS

The intent of this software operational guide is to enable its user to successfully operate the Systems Communications Module (SYCOM) under control of the B1800/B1700 Master Control Program (MCP). As a means to this objective, this cocument defines scope and objectives, describes inherent and command driven functions, and presents basic implementation characteristics. Additionally, provides operating procedures at both system and terminal levels. (Termiral operating procedures are limited to those capabilities within the scope of SYCOM.)

The Systems Communications Module can be used for a multitude of applications and by users with varying degrees of proficiency in computer operations. Although this guide has a tutorial slant, it does not claim to train operators. The reader is assumed to have a basic knowledge of the B1800/B1700 system and of the Master Control Program (MCP). The reader is also assumed to be familiar with B1800/B1700 operating procedures, to have had experience operating the system Supervisory Console (SPO) and peripheral devices, and to have a basic vocabulary of computer related terms. The user need not however, need any knowledge of programming or of the Network Definition Language (NDL).

This guide is designed to be self-contained. However, if the user wishes to have a deeper insight into the system's construction or precise details as to the various internal system functions, a reading of the "Network Definition Language Reference Manual" and the "System Software Operational Guide" is strongly recommended.

Software Operational Guide

Preface

12 Sep 1978 2

Page -

(Original Version)

STRUCTURE OF THE DOCUMENT 

Section I presents an overview of the Systems Communications Module along with the functions it performs. This section also identifies any limitations which might be applicable to the usefulness of the system and lists associated documentation.

Section II describes concepts and implementation characteristics employed by the Systems Communications Module. This includes system configurations, functional specifications, relationships in inter-system communications, and intra-system relationships with other system software elements. Pictorial overviews are presented where applicable to elaborate points of interest. Presentation of the Functional Description progresses from basic introductions, to what is involved, thru detailed "how it works" explanations.

Section III describes directive commands available to the user from either a Work Station (terminal) or the User SPO. With these commands, the user directs SYCOM to perform specific functions to satisfy a user's processing requirements. Examples are offered for each command which includes both commanc input and reply messages. The commands are presented in alphabetical order for ease of reference. The command language construction is presented in detail; including keyword/parameter semantics.

Section IV contains operating procedures for a SYCOM system (operating independently of in conjunction with the Supervisory Message Control System (SMCS)) and for local system terminals. Under the heading "System Operating Procedures", useful SYCOM operating procedures, commands, and operational hints/pointers are presented. Under the heading "Terminal Operating Procedures", terminal operational commands and functions are covered which assume either a TD820 or TD830 Video Display Terminal (VDT). However, the system utilizes any Burroughs supported VDT, (see "Limitations").

Section V presents instructions for the preparation of a Network Controller program required by the SYCOM system. It is basically comprised of a sample NDL coded input file for the Controller's generative process. Various comments are included throughout the sample listing to explain variable parameter data which could be different for each installation. The procedure used to invoke the generative process for an operational Network Controller program tailored to the needs of individual systems is included.

12 Sep 1978

Software Operational Guide

Page – 3

Preface

(Original Version)

Section VI contains the appendices to this cocument, each of which is presented below:

Appendix A provides lists of error messages and their meanings. These lists are broken down into two major categories; those error messages which may be displayed in response to commands and those error messages which are displayed on the User SPD which identify a processing exception condition within SYCOM.

Appendix 8 contains the "character set" defined for terminal input/output and data transmissions.

Appendix C lists the "reserved words" recognized by SYCOM either as a command or command option keyword.

Appendix D contains a "command summary" which lists each command and briefly describes its purpose.

Appendix E provides a list of all control messages used by the communications protocol and includes their associated format attributes.

Appendix F presents a list of restricted MCF commands which are permitted from a terminal or SPO on the User System.

Appendix G describes the PARAMETERS file attributes and details the parameters record format.

Appendix H provides a summary of system files; including names and purposes.

Appendix I contains sample application programs which show the mechanics of interfacing with SYCOM and the system's terminals for program-to-program and terminal-to-program communications. These sample programs can also be utilized in the system test procedures once the appropriate Network Controllers are created and the systems installed.

Section VII is a glossary of terms and acronyms used throughout this document.

SECTION I

INTRODUCTION TO SYCOM

12 Sep 1978

5

Software Operational Guide

Page -

(Original Version)

Section I Introduction to SYCOM

SECTION I

INTRODUCTION TO SYCOM

#### GENERAL DESCRIPTION \_\_\_\_\_\_

The Systems Communications Module (SYCOM) program provides the B1800/B1700 user with 1) the capability to submit jobs to run on another system, 2) to transfer files between two systems, and 3) in general, to interactively obtain services and information from another system with the ability to communicate with interactive programs in a remote system. Job run output can be returned to the submitting system (upon request) or directed to a peripheral device on the processing system itself. Interactive output data is returned to the User System directly, via Network Controllers.

SYCOM along with its associated Network Controller is responsible for all data communications with other systems. This includes performing the necessary functions to establish line connections, receive/transmit data, and execute the correct line termination procedures.

The Systems Communications Module maintains a secondary level of control (MCP has primary control) over local peripheral devices such as the card reader, card punch, line printer, and magnetic tapes. It reads and transfers card, tape, and disk files, accepts input from and displays messages to the Supervisory Console (SPO) and receives data files/messages for printing, punching, display, or writing to either tape or disk. It provides the mechanism by which a terminal or apolication program in one system is able to communicate with specific application programs in another system. Program selection is based on the servicing system which dictates the processing requirements (programs) optioned to run in that system.

12 Sep 1978

Software Operational Guide

Page - 6

(Original Version)

Section I Introduction to SYCOM

## SCOPE AND OBJECTIVES

The purpose of the Systems Communications Module is to increase the use of data processing facilities (systems) either local or remote to the target system. The scope of SYCOM is to provide all of those features previously afforded by the Remote Job Entry (RJE/TERMINAL) system and additionally provide enhancements with the following expansive communications features: file transfer, terminal-to-program, program-to-program, and program-to-SYCOM. The first two functions assume inter-system communications with the program-to-SYCOM function assuming intra-system communication which is provided by the MCP. The Systems Communications Module with these encompassing capabilities, considerably expands the flexibility of a RJE Station; henceforth referred to as a "User System".

In addition to the above design objectives, several mandatory objectives have been specified. They are: 1) Retain compatibility with the standard Burroughs data communications line discipline and message format conventions and 2) maintain standard Remote Job Entry (RJE) command compatibility with existing RJE program products for Burroughs small, medium, and large systems.

With this software system, the objectives are also to provide the user with the most benefits in terms of flexibility and ease of use. Flexibility in that the user is free to design and install system configurations suited to the requirements of the company for distributed processing. For example, a user can install a B1800/B1700 in the same installation to increase/share resources with an existing system; in a remote installation to place some resources in a remote facility; or in a network of systems to provide distributed processing services. Ease of use in that all capabilities of the system can be selectively utilized by any user thru either command initiated or inherent system functions.

FUNCTIONS PERFORMED

Since SYCOM applications imply transmission of large data files between two or more systems, line throughput maximization, error detection, and error recovery are standard features. Following is a comprehensive list of the functions performed; including those that are under user control. They are presented in three groups, Basic, Line Throughput, and Error Detection and Recovery.

12 Sep 1978

#### Software Operational Guide

Page - 7

Section I Introduction to SYCOM

(Original Version)

BASIC FEATURES

For SYCOM to accomplish the functions generally described for the system, the following basic features are provided; full details are presented in Section II, "Concepts and Facilities".

- A. File control: SYCOM verifies all file transfers between two systems.
- B. Input: Introduction of programs from an input device on the local system for compilation and/or execution on a remote system. Introduction of data from an input device on the local system for processing on the remote system.
- C. Command Input: SYCOM commands as well as SMCS/MCP type commands are permitted from the system SPO, terminals, and specifically tailored application programs running in that system.
- D. Output: Dispersion of data and/or programs produced by the remote system to the local systems peripheral output devices, SPO, terminals, and application programs.
- E. Monitoring: Monitoring and controlling programs on the remote system via a local system's Remote Supervisory Console (RSC); terminal, SPD, or application program (requires SMCS on remote system).
- F. File Transfers: The transfer of certain types of files between a local system and a remote system; in both directions.
- G. Program-to-Program: Support of communication between a program running in a local system and a program running in a remote system. Also, support a local application program communications link to SYCCM itself.
- H. Virtual Terminals: Provide a means by which a terminal, SPO, or application program in a local system can become a virtual terminal entity to a remote system. By design, virtual terminals are automatically assigned the status of a Remote SPO when connected to a SMCS in the remote system.
- I. Communications: Supports a full line of communications; asynchronous, synchronous, and binary-synchronous data transmissions (including broadband up to 50,000 bbs).

12 Sep 1978

Software Operational Guide

Page - 8

(Original Version)

Section I Introduction to SYCOM

LINE THROUGHPUT

Maximization of line throughput is assured by a point-to-point message transfer procedure which:

- A. Specifically acknowledges receipt of a message with the transmission of either another message block or a "numbered acknowledgement" (#ACK) message. This reduces both line time and turnaround time.
- B. Blocks records for transmission, (i.e., sends multiple data records per message block.
- C. Compress and expand blanks in data records. It is common for card input and printer output to have a large number of imbedded blanks. Eliminating transmission of these blanks significantly improves line throughput.

ERROR DETECTION AND RECOVERY

The following features are provided in error detection/recovery areas to assure reliable service to the system user:

- A. A specific numbered acknowledgement (either in the next message or in a #ACK) guards against the possibility of a lost message by providing an unambiguous acknowledgement for the transmitted message block.
- B. The use of modulo 2 by the transmission numbering facility for the detection of duplicate message blocks.
- C. Dynamic reduction (by command) of the transmission message block size provides the potential for reducing error rates and retransmissions.
- D. The convention of different timeout values for local and remote systems alleviate line collision problems.
- E. The message block transfer procedure eliminates the ambiguous use of certain ASCII control characters (for example, "ENQ" is used only during "establishment"; and no where else).
- F. Data transmission error detection is provided thru the use of vertical and horizontal parity in the systems hardware and software.

12 Sep 1978

#### Software Operational Guide

Page - 9

Section I Introduction to SYCOM

(Original Version)

LIMITATIONS

The Systems Communications Module (SYCOM) is intended to provide a combined system of RJE, file transfer, program-to-SYCOM message transfer and terminal-to-program/program-to-program communication capabilities. To provide these services and still maintain a certain degree of compatibility with existing Remote Job Entry (RJE) systems, certain limitations become inherent to the system. Therefore, applicable limitations are defined as follows:

- A. SYCOM does not, and will not, support Cassette tapes as an optional peripheral device.
- B. In reference to "file transfers", the implementation currently in effect does not save, or forward, symbolic file headers. Therefore, certain file characteristics would be lost on file transfers between unlike systems.
- C. Program-to-Program limitations involve an application program being assigned a "virtual terminal" position which reduces the number of positions available for terminal attachments.
- D. Interactive application programs in a remote system serving virtual terminals on a local system cannot make use of "tallies" or "toggles". This prevents CANDE from from being a viable application program in this system.
- E. The implemented version of SYCOM which this document describes, does not have the "file transfer recovery" capability. Therefore, when the cocasion arises when an inter-system connection is lost, any in-process file transfer operations are aborted and must be restarted from the beginning.
- F. The most noticeable limitation is the fact that SYCOM only supports a single line connection for RJE and File Transfer data communications; at any one time.

## NOTICE

The specifications of the software system described in this document are subject to change without notice. The availability or performance of some features may depend on a specific configuration of equipment such as larger memory or communication line speed. Customers should consult their Burroughs Sales Representative for details.

B1800/B1700 SYSTEMS COMMUNICATIONS	MODULE 12 Sep 1978
Software Uperational Guide	Page - 10
Section I Introduction to SYCOM	(Original Version)

ASSOCIATED DOCUMENTATION

Some of the details of the Systems Communications Module system and its composite software elements are contained in other system manuals. Following is a list of associated documents referenced in this guide. Readers and users of this document may refer to any or all of these listed documents for additional information.

Company Confidential

Document Title		Identi	ficatio	on Nu	nber
B1800/B1700 SUPERVISORY	MESSAGE CONTROL	SYSTEM	P.S. 2	2219	0482
MESSAGE CONTROL SYSTEM	INTERFACE		P.S. 2	2212	5447
B1700 NETWORK DEFINITIO	N LANGUAGE		P.S. 2	2212	5223
B1800/B1700 NDL LIBRARY			P.S. 2	2212	5215
B1700 DATA COMM AUDIT			P.S. 2	2212	5421

Corporate Released

Document Title	Form Number
B1800/B1700 SYSTEMS SYSTEM SOFTWARE OPERATIONAL GUIDE	1068731
B1700 SYSTEM NETWORK DEFINITION LANGUAGE (NDL) Reference Manual	1073715
B1700 SYSTEMS DATA COMMUNICATIONS Information Manual	1089992
BURROUGHS CORPORATE DATA COMMUNICATIONS STANDARDS	1284 9006
B1800/B1700 SYSTEMS REMOTE JOB ENTRY (RJE) Reference Manual	1090602
B1800/B1700 SUPERVISORY MESSAGE CONTROL SYSTEM	Preliminary

SECTION II

CONCEPTS AND FACILITIES

B1800/B1700 SYSTEMS COMMUNICATIONS I	MODULE	12 Sep 1978
Software Operational Guide		Page - 12
Section II Concepts and Facilities	(Orig	inal Version)

SECTION II

CONCEPTS AND FACILITIES

## GENERAL DESCRIPTION

To elaborate upon the design concepts applied to SYCOM, the full range of system facilities are presented in logical progression. First to be covered are actual system hardware configurations. This provides the base for further concept elaborations. Next, descriptions of the data communication networks involved and the relationship of a User System to a Server System are identified. Following this, implementation concepts and functional aspects of the system are described.

## THE SYSTEM CONFIGURATION

A User System oriented to SYCOM utilizes a hardware configuration similar to the one shown in Figure II-1 (next page). A system is usually comprised of the following hardware elements: processor, memory, SPO, disk, plus standard communication controls/adapters. Optionally it can be equipped with an assortment of peripherals as per a user's needs (e.g., card reader, card punch, disk packs, line printer, magnetic tapes, etc. and terminals).

The SYCOM program is responsible for managing availability and use of peripheral devices which are used for SYCOM functions. It also controls all input/output operations for those peripheral devices as well as transmitting and receiving data files. (Data file is a general term used here to encompass all file types acceptable to SYCOM such as data, program source/object, printer backup, etc..)

12 Sep 1978

Software Operational Guide

Page = 13

(Original Version)

Section II Concepts and Facilities

With this hardware configuration, "all" SYCOM capabilities for a User System are initiated thru the system SPO. An "extended" User System (configuration with local terminals) adds a new dimension to the SYCOM system in that it's capabilities are no longer confined to initiation from the SPO but are now also available to local terminals as well. Commands entered from a system SPO can initiate any SYCOM function; including establish the connection for virtual terminal communications. Terminals are, however, restricted to an explicit command repertoire. (Details of this concept are presented later in this section.)



Software Operational Guide

Page = 14

(Original Version)

Section II Concepts and Facilities

SYSTEM COMMUNICATIONS

Described here are the communications network aspects of a User System and its relationship to a Server System. SYCOM and its associated Network Controller supports User Systems using either a point-to-point non-switched or point-to-point switched network to a Server System.

The data communications procedure utilized is a subset of the Burroughs conversational point-to-point line procedure. It is defined in a document entitled "Burroughs Corporate Data Communications Standards"; refer to "associated documentation". That document fully describes the characteristics and functional attributes for a non-transparent half-duplex mode of data communication to which this system subscribes.

Point-to-Point Switched Network

In a switched network environment, a User System communicates with a Server System without requiring dedicated communication lines; see Figure II-2 (on the following page). The computers are connected by telephone lines to the common carrier exchange serving their respective locations. A temporary connection (data path) is established between the computers only for the period of time in which data transmission is actually required. The connection is established by dialing the telephone number of the Server System. Usually the dialing procedure is manual, however, SYCOM can cause this function to be done automatically upon command directive.

In switched Line networks, a line refers to a discrete data path between a Server System and the User System (including terminals) thru the common carrier exchange. The services provided by this common carrier are usually on a time-used basis.

12 Sep 1978

Software Operational Guide

Page - 15

Section II Concepts and Facilities

(Original Version)



12 Sep 1978

Software Operational Guide

Page = 16

Section II Concepts and Facilities

(Original Version)

Point-to-Point Non-switched Network

A point-to-point non-switched network, as shown in Figure II-3 (below), consists of a private or leased communications line that connects the User System with a Server System. In this network, the computers are physically connected, that is, the circuits making up the communications line are permanently established, and data may be transmitted over the line at any time. No dialing is required for this type of network.

Communication lines in a non-switched network are alternately known as private, leased, or dedicated lines. In this document they are referred to as "dedicated lines". These lines may be privately owned (e.g., internal direct connection in the same physical location), or provided by a common carrier on a contract basis. This dedicated connection between two specific locations would provide a line for exclusive system use.



Figure II-3: Point-to-Point Non-switched Network

12 Sep 1978

Software Operational Guide

Page - 17

Section II Concepts and Facilities

(Original Version)

Communications Procedure

With the conversational point-to-point communications procedure assumed by SYCOM, inherent provisions include the capability to "reverse" message block transmissions without having to invoke separate establishment and termination procedures. The essential characteristic of this capability is its ability to specifically "acknowledge" receipt of the previous message block. This is accomplished by including a special acknowledgement character in the header of the "next" message block transmitted. This special character is the Acknowledge Received Message (ARM) character.

The usefulness of the ARM character is twofold: First, it provides a specific acknowledgement to the receipt of a previous message block by utilizing the "numbered ACK" facility. Secondly, when the receiving system receives a message block that it cannot accept (no buffer space or a specific peripheral is not-ready), the ARM character permits an implicit NAK of that message block. This is achieved by transmitting the NAK in the next outgoing message block.

User System - Server System Relationship

The concept of a User System - Server System relationship is quite fleeting, especially when both systems are in the same physical location. The important point to remember, however, is the basic principles of this relationship (described below) and how it applies to this system.

Referring to Figure II-4 (following page), the User System is viewed as the system in the network which serves as a point of data collection and data distribution. For example, RJE type card input, card output, line printer output, ard/or terminal input/ output messages.

A Server System on the other hand is generally considered to be the data processing system; often referred to as a Host System or Central System. It normally receives large volumes of data from a User System to be processed or program source files to be compiled. In the case of program compilations, the printout and/ or program object code files can be transmitted back to the User System (upon request) for printing or possible program execution. Interactive output is returned directly to the initiating program or terminal via the Network Controllers.



Figure II-4: User System - Server System Schematic

12 Sep 1978

Software Operational Guide

Page = 19

Section II Concents and Facilities

(Original Version)

The actual physical size of a computer however, has relatively little importance when considering a system as a Server System. The B1800/B1700 system can be either a User or Server System; depending on the angle of view and the services required at the time. It can also be, under certain circumstances, both a User System and a Server System.

SYCOM AND ITS ENVIRONMENT

## HARDWARE REQUIREMENTS

The following list of equipment must be present for SYCOM system operations. However, the listed equipment is not dedicated to SYCOM (or supporting system software) and is available for use by other programs.

Hardware Type	Usage
B1800/B1700 Central Processor	System Control
98 KB Main Memory (minimum)	System Software
Secondary Memory (disk storage) (head-per-track, pack)	Program/Data Files
120 or 132 Position Printer	Line printer output
80 or 96 Column Card Reader	Control/RJE Type Input
80 or 96 Column Card Punch	Control/RJE Type Output
System SPD	MCP/SYCOM/SMCS Operations
Single or Multi Line Control	System/Terminal Communications
Line Adapters as Required (leased, switched, direct)	System/Terminal Communications
Data Sets and Cables (as required)	System/Terminal Communications
TD820/TD830 Terminals	As Required

Software Operational Guide

Page - 20

(Original Version)

Section II Concepts and Facilities

MEMORY REQUIREMENTS

Provided here is a list of System Software elements and their associated memory requirements. They can be used to approximate memory requirements for an interactive B1800/B1700 SYCOM system; operating with or without a Supervisory Message Control System (SMCS):

64.0 KB Master Control Program (MCP).
15.0 KB Network Controller (basic).
3.9 KB Each additional line in the Controller.
16.0 KB Supervisory Message Control System (SMCS).
18.0 KB Systems Communications Module (SYCOM).
22.0 KB Command AND Edit system (CANDE).
2.0 KB Each additional terminal Logged-on CANDE.

After determining the basic memory requirements for the system, additional memory must be included for any other programs that are to be executed in conjunction with SYCOM. (See appropriate reference manuals for those memory requirement estimates.)

#### SOFTWARE REQUIREMENTS

In order for SYCOM to become operational, certain system software elements are required. Although these elements support the operation of SYCOM, they are general purpose and in fact support the entire system. Each software element identified here is individually documented for its capabilities and requirements. Following is the list of those pertinent system software elements required to support SYCOM operations.

- Master Control Program (MCP)
- SYSTEM/MAKEUSER program
- DMFALL program
- Network Controller program

12 Sep 1978

Software Operational Guide

Page - 21

Section II Concepts and Facilities

(Original Version)

Master Control Program (MCP)

The MCP is an operating system which manages the demands and resources of B1800/B1700 systems. It reduces programming efforts by providing a family of commonly needed furctions and services. The MCP requires strict user adherence to program structures and procedures in order that there may be communication between the different programs, interpreters, and the MCP itself.

Its primary function is to optimize the productivity of a system. External intervention is held to a minimum, and maximum throughput is achieved by incorporating into the MCP the primary tasks of Input/Output (I/O) control, file handling, multiprocessing, interrupts, memory allocation, application programs, and operator interfaces.

SYSTEM/MAKEUSER Program

A utility program that is used to create, modify, and access the system's "usercode" file. This file contains all valid system usercode/password combinations. These usercodes-passwords are maintained in a file named "(SYSTEM)/USERCECE". With this file, the MCP can support, and enforce, a file security mechanism for disk files and programs/users that access them.

DNPALL Program

The DMPALL program is a general media conversion facility that allows reproduction of files from one medium to another. It also can list files from any medium with byte, digit, or combined format. Along with these abilities it can start or stop its reading of these files at a specified record. Specifications for these functions may be passed to DMPALL from a SPO, Card Reader, or local system terminal.

Software Operational Guide

Page - 22

Section II Concepts and Facilities

(Original Version)

## Network Controller

The Network Controller is a Network Definition Language (NDL) type program whose prime function is to process and supervise the flow of messages between systems, terminals, and programs. It performs all data communication line discipline functions which are necessary for proper line/terminal control. The Network Controller handles the initiation and completion of communication I/O's and services exception conditions. In order to optimally provide these functions, the Network Controller is configured, by NDL generation, to the specific needs of the system on which it is to be run, (i.e., the number and type lines, number and type terminals, etc.). A product of this generative process is the Network Information File (NIF). This file contains all terminal and communication line attributes as well as logical aspects of the communications network.

Basically, this Controller enables messages to pass between it and SYCOM as well as other terminal oriented programs executing in the system. This is accomplished via MCP supported Remote Files and Queue's along with the Controller's message switching capability. Each program that is to be part of the message flow must contain a "remote file" for input/output message traffic and it must be known to the Network Controller; this includes SYCOM.

SOFTWARE OPTIONS

The following software elements are not required but may be used in conjunction with SYCOM to service a particular user's needs.

- Supervisory Message Control System (SMCS)
- Application Programs
- CANDE Program (multiple-user text editing)

Although the Supervisory Message Control System is not a required system software element for the operation of a SYCOM system, it is definitely suggested for those systems which have terminals. A Terminal in a SYCOM system environment is permitted to initiate "file transfers" or become a "virtual terminal" entity on a Server System. Additionally, these terminals may be utilized for other applications such as data entry and program development. To maintain overall control of the terminals in these varied system environments, the SMCS becomes the controlling element.

Software Operational Guide

Page - 23

Section II Concepts and Facilities

(Original Version)

Supervisory Message Control System (SMCS)

The Supervisory Message Control System (SMCS) is intended to be the "supervisor" for a data communications software system which includes such Burroughs software elements as "SYCGM", CANDE, and other online programs of either Burroughs or customer origin.

The SMCS provides to an interactive system, many functions and special features. To fully explore its capabilities, please consult the "SMCS Software Operational Guide". However, the highlights of some of the more pertinent functions are presented here. Relevant to the SYCOM system, SMCS provides some basic message switching (terminal to terminal, terminal to SPO, and SPO to terminal), attachment of terminals to application programs via a remote file, and information about the system's network.

Additionally, it controls terminal log-on security (checking of user security codes and granting access to user files) and approves (or denies) all remote file opens to terminals under its control. Moreover, SMCS provides a certain degree of resource allocation. By this is meant, thru security, job control, remote file control, and error handling, the SMCS can allocate some of the systems resources. For example, for every terminal controlled by the SMCS, it has final control over that device as a resource. It can therefore, temporarily "attach" terminals to SYCOM by means of the SMCS "SIGN ON" facility.

Application Programs

Effectively there are three types of application programs useful to the SYCOM system user, they are: 1) Interactive data entry, 2) system-to-system, and 3) command input to SYCOM. Each program type is briefly described here with details of how they actually work presented in the following "Functional Description".

A user can elect to develop an "interactive" application program tailored to specific needs of an installation. These specific needs could include such applicational functions as data entry, information retrieval, or any application desired.

If there is a requirement for an application program in a User System to communicate directly with another application program in a Server System, the capability is there. This communication link is directly thru the Network Controllers and SYCOM has no role in this message traffic. SYCOM does however, initially

12 Sep 1978

Software Operational Guide

Page = 24

Section II Concepts and Facilities

(Original Version)

approve a "virtual terminal" isn for identification and control purposes. This is a residual capability based on requirements for Network Controllers by SYCOM.

The third type of application program provides the source for a string of SYCOM commands to be performed on an automatic basis. Normally this would be for off-hours operation but could also be utilized for common everyday occurrences of a work flow pattern.

CANDE Program

The B1800/B1700 CANDE program provides generalized file updating and preparation capabilities in an interactive, terminal oriented environment. CANDE performs these file updating and text editing functions in program development and other such environments. For this system however, CANDE can only be utilized when SYCOM is not involved (i.e., for the creation and maintenance of "NDL source" files).

The CANDE system has been designed to run in a self-sustaining operation or in conjunction with the SMCS. (In either case, it requires a Network Controller for data communication functions.) It operates with a basic usercode-password type of security that is available to CANDE, and SMCS, from the MCP.

FUNCTIONAL DESCRIPTION

The operation of SYCOM is dependent upon support from a Network Controller program. This Network Controller provides interfacing links between SYCOM and the communications hardware (terminals and other systems). The controller furnishes those services and capabilities which are outlined in general with the preceding subject, "Software Requirements". Throughout this description, Network Controller functions are elaborated upon where it is necessary to maintain continuity of subject matter. Concluding this section, the functions of a Network Controller is outlined under the heading "Network Controller".

The following features of the system are functionally described in the order in which they are identified below: Standard RJE, File transfers, Virtual terminals, Program-to-program, and Network control. The virtual terminal capability contributes terminal-to-program, program-to-SYCON, and Remote Supervisory Console (RSC) communications.
12 Sep 1978

Software Operational Guide

Page - 25

Section II Concepts and Facilities

(Original Version)

STANDARD RJE FUNCTIONS

The SYCOM program controls, at a secondary level, all typical RJE peripheral devices of a User System. It reads and punches cards, prints output, accepts SPO input, displays SPO output messages, and transfers file/control messages between the User System and a Server System.

Input can be directly from a card reader, from tape or disk thru file equates, or from the SPD. Card reader input is generally associated with RJF batch data transmissions; whereas magnetic tape and disk input more directly relates to file transfers. However, these operations in SYCOM are considered synonymous.

Commands can be alternately input from a specifically designed application program thru the "Program-to-SYCOM" facility. With this mechanism, the RJE input as well as associated output could be automatically controlled on a daily basis without requiring the system operator to enter a long series of commands to SYCOM.

Butput can be made directly to a card punch, line printer, disk, magnetic tape, or SPD. However, in the case of a printer or card punch, the data received can be in the form of a "backup file" which is first written to disk. Subsequent MCP commands are entered to cause printing or punching as appropriate.

Furthermore, SYCOM gives certain directives to the Network Controller such as requesting it to establish or re-establish a connection to a Server System. SYCOM also causes the controller to enter an "idle" state while waiting for an incoming connection to be established and to terminate a connection when appropriate.

The transmission of RJE data files to/from a Server System occurs as follows: A user of the User System prepares a card deck for transmission to the Server System. The cards are read, compressed and blocked by SYCOM (upon command) and written into the Network Controller's reacte file. (See writeup on "Queue Depth Control".) The controller begins immediately to transmit these card images (message blocks) to the Server System. This process continues until all cards have been read and transmitted. The card file, known as a "job stream", is then under control of the Server System's MCP which processes that job stream as per directives included in that file (compile and/or execute).

12 Sep 1978

Software Operational Guide

Page - 26

(Original Version)

Section II Concepts and Facilities

All SPO type messages which bertain to that job (80J, EOJ, and SPO messages) are transmitted back to the User System for display on the User SPO. Messages can be entered via the SPO to control jobs executing on the Server System or manipulate data files as required; providing a MCS is running in that system. SYCOM accepts these messages and writes them to the Network Controller for transmission to the Server System and subsequent processing.

When the job has finished, any associated line printer and card punch output created at the Server System is transmitted back to the User System. (This is dependent upon the software system in the Server System; HOST/RJE does this automatically where SYCOM does not.) The Network Controller receives these incoming message blocks and forwards them to SYCOM via SYCOM's remote file (input queue). SYCOM then appropriately prints and/or punches this data on a User System peripheral device.

The afore described process describes the flow of data between the User System and a Server System; often referred to as a "job stream". Any number of job streams may be initiated at the same time (multiple Server System MCP directives contained in one job file) from a User System. The number of jobs which can or will run on a Server System is limited only to its capabilities to process them.

FILE TRANSFERS

File transfers are supported for both disk and magnetic tape devices. Files may be transferred between any combination of these devices. Magnetic tape transfers are restricted however, in that only one file per tape may be transferred. Library tapes created by the "LGAD.DUMP" function are not supported by SYCOM. (These tape files are first loaded to disk and then transferred.) Between "unlike" systems, only program code files and 8-bit data files (includes text files) are permitted. On the other hand, "like" systems can transfer other files such as printer backup, punch backup, interpreter, intrinsic, and dump files. The data communications protocol allows transfer of only one file at a time; in each direction.

The user initiates a file transfer by entering a "COPY" command to SYCOM (PUT and FETCH are acceptable alternates). The command syntax allows source and cestination systems to be specified (PUT and FETCH excluded) for the transfer via attributes. These source and destination system identifiers must be the same as those specified in the "system-id" control message at connection initialization time.

12 Sep 1978

### Software Operational Guide

Page - 27

Section II Concepts and Facilities

(Original Version)

If the specified source system is the User System, the transfer is a "PUT"; which is a transfer of a file "to" another system and is based on SYCOM's "SYSTEMID" parameter. When the source system is a Server System, then the transfer is a "FETCH"; which is the transfer of a file "from" another system, also based on SYCOM's "SYSTEMID" parameter.

The actual transfer of a file is initiated by exchange of control information between systems; this process is often referred to as "handshaking". All file transfers occur as "Put's". That is, if one system wishes to do a "Fetch", it sends a "Fetch message" to the other system. This causes the other system to initiate a "Put" sequence by sending a "Put Message" to the initiating system. Put and Fetch messages completely specify both source and destination file attributes. The Put message elicits a "Put Reply Message", signifying that data transfer may begin. The sender system then sends one or more data message blocks which are eventually followed by an end-of-file message.

Upon completion of the file transfer, the system that received the file informs the originating system of the completion. This notification is made with a "user-destined control message".

If either system wishes to abort an in-process file transfer (for any reason), that system sends either an "Abort Put" or "Abort Fetch" system control message to the other system. An abort is initiated by the appropriate command entered to SYCOM. Alternate "device not ready; suspend" and "device ready; resume" messages are used for flow control at any time for temporary stoppages.

A file transfer request that cannot be processed when entered, because another file transfer is in progress, is queued to disk and processed when next possible. Although there is a maximum depth to this disk queue (see "Restrictions"), it is sufficient to handle all but the most extreme cases. This feature prevents request rejections due to the fact that only one file transfer (in each direction) can be in-process at any given time. Requests are queued on a first come, first serve, basis. This mechanism assures an orderly processing sequence.

An important point to note, all file transfer requests are queued as "Puts". This is, if a user requests a data file "Fetch", that request is sent to the other system's SYCCM where it is queued as a "Put" on that system.

12 Sep 1978

Software Operational Guide

Page - 28

Section II Concepts and Facilities

(Original Version)

Maintaining Integrity of File Characteristics

A goal of the file transfer mechanism is to maintain as many of a file's characteristics as possible throughout the transfer. For example, if a COBOL source program is transferred, it should be capable of being compiled, without further modification, on the receiving system. Likewise, if a program code file is transferred to a "like system", it should be executable without modification.

Specifically, if a file is transferred between two like systems, even if via one or more unlike systems, all file characteristics must be preserved. To accomplish this, a symbolic version of the file's header (understandable to all systems) is transferred with the file. This symbolic file header is the "PUT-DR-FETCH-RECORD" as defined for SYCOM file transfers. It is prefixed in the data stream by a beginning-of-header control message.

In the receiving system, when a symbolic file header is received, as many as possible of the file characteristics are incorporated into the file being created/received. If the file is transferred to an "unlike" system, the symbolic file header and the original system type are saved with the file, to be forwarded with the file whenever it is again transferred. If the file is transferred to its original system type, the entire symbolic file header is incorporated to create a file identical in characteristics to the original file, and the symbolic header is not saved.

When a system sends one of its own files, it must include in the symbolic file header, enough information for it to recreate that file if returned. If this file is returned, it must be made to look identical to the original file, with the exception of "file name" (this is conveyed in the symbolic file header).

An unlike system created file may be changed or replaced locally, thus invalidating the saved symbolic file header. In this case, the symbolic file header describes the file according to its new creation system type.

NOTE OF IMPORTANCE: The "symbolic file header" discussed above is not implemented in this version of the system software. It will however, be available in a later release of the system.

12 Sep 1978

### Software Operational Guide

Page - 29

Section II Concepts and Facilities

(Original Version)

# VIRTUAL TERMINALS

To the SYCOM system, a "virtual terminal" can be either a local terminal, the system SPD, or an application program. Therefore, virtual terminals is a logical capability whereby a message path can be established between a terminal, SPD, or program on the User System and an appropriate interactive application program in a Server System. Additionally, utilize the inherent capability of controlling jobs in a Server System with a special MCP command repertoire. Basically, on the Server System end, this becomes a reality thru the capabilities of the SMCS. However, the main inter-system connections are established by SYCOM which makes it possible. Refer to Figure II-5 for a pictorial overview of a User System with virtual terminals. (Interactive application programs can be any program using a "remote file" but not using "tallies" or "toggles".)

The terminals are virtual in that all User System facilities as well as all Server System facilities are available to the system user. User System commands are local and are therefore, processed within the User System by either SYCOM, SMCS, MCP, or application program. Server System input messages are accepted by the User System and transmitted to the Server System (via the Network Controllers) for processing and subsequent reply messages.

In view of a "virtual terminal" in an interactive environment, Server System commands must be forwarded as soon as possible. To effect this requirement, SYCOM has the capability to "interleave" a "user destined control message" into the data transmission stream immediately if a file transfer is in progress. If SYCOM is not presently transmitting a file, the control message which contains the virtual terminal message is immediately transmitted on an individual basis to the Server System.

# Terminal-to-Program Communications

When SYCOM is utilized in a larger system development or data entry environment (with terminals), then the User System depicted in Figure II-11 is applicable. In that hardware/software configuration, the complete spectrum of SYCOM capabilities are available to the system user. To be more specific, in addition to the capabilities previously identified, a terminal-to-program capability is available to the terminal user.

1800/B1700 SYSTEMS	COMMUNICATIONS MODULE	12 Sep 1978
oftware Operationa	Page - 30	
ection II Concepts	and Facilities	(Original Version)
	· · · · · · · · · · · · · · · · · · ·	• SPO/RSC • • Virtual • • Terminal •
he User System		
	l . Network Controller	I I 
• • • • • • • • • • • • • • • • • • •		• • • • • • • • • • • • • • • • • • •
• SYCOM • •	A A A A A A A A A A A A A A A A A A A	pplication Program . [ Virtual Terminal) . [ 
		The Server System
SMCS	Network Controller	MCP
- Sycom		Application Program . I

Figure II-5: Virtual Terminal Schematic

12 Sep 1978

Software Operational Guide

Page - 31

(Original Version)

Section II Concepts and Facilities

In this environment, SYCOM still communicates directly with the Network Controller for normal communications activity. However, the SMCS maintains elementary control over SYCOM and control over a system's terminals. To clarify this, the SMCS directly causes the execution of SYCOM when a terminal (or RSC) "signs-on" to SYCOM thru the SMCS. A terminal user can then direct commands to SYCOM in the User System or the SMCS in either the User or Server System. Exactly which commands and how the directives are given is covered in Section IV, "Terminal Operations". An important note; these capabilities are also available to the user who optionally elects not to utilize a SMCS in the system. (Refer to Figures II-9 and II-10 for configuration cverviews.)

With the introduction of the SMCS software element into a system, all of its commands, functions, and capabilities become available to the terminal user. (Refer to the "SMCS Software Operational Guide" for details and capabilities.)

Before proceeding, it might be helpful to this discussion for the terms User System and Server System to again to clarified. A User System is the computer on which there are terminals/users wishing to engage in interactive computing sessions. The Server System is the computer that supplies those interactive services on a demand basis. (Refer to Figure II-4; back a hit.)

The communications protocol assumed by SYCOM supports User System terminals connected (logically) to an interactive application program on a Server System. Using this protocol minimizes the visibility of interfacing software elements so that to a user, it appears as though the terminal is connected directly to the application program. The terminal is then used as if it was locally attached to that program (see "Message Path Assignments" further on in this section).

Presently available interactive application programs may be used or the user can optionally elect to develop a special application program for terminal and/or other program interaction.

The method for conveying terminal data between two systems makes use of the "device address" field in the control message. This message specifies the terminal with which the message text is associated. The device address in the cortext is termed "virtual address", and is the name used by both User and Server Systems to reference a particular terminal. This addressing scheme follows:

A. The device address is comprised of two characters; one upper case and ore lower case alphabetic letter.

12 Sep 1978

Software Operational Guide

Page - 32

(Original Version)

Section II Concepts and Facilities

- The User System "transmits" a virtual terminal address
- B. The User System "transmits" a virtual terminal address as lower case/upper case letters. When it "receives" a response message from the Server System, the terminal address letters are reversed; upper case/lower case.
- C. The Server System "receives" a virtual terminal message with a device address of lower/upper case letters (as transmitted by the User System). The virtual terminal addressing scheme then requires the Server System to "reverse" these device address characters prior to its transmission back to the User System. (These transmit/ receive device addresses are specified in the Network Controller's NDL generation process; "both" systems.
- D. Assume that a User System supports a connection for five terminals to a Server System. The User System refers to these terminals by virtual terminal (transmit) address:

aA, bA, cA, dA, and eA

The Server System also refers to these same terminals by virtual terminal (transmit) address; but, as the Server Systems role is reversed, its transmit address is also reversed. For example:

Aa, Ab, Ac, Ad, and Ae

The following events represent the terminal-to-program procedure implemented by SYCOM. These events portray an interactive session between a virtual terminal on a User System and an application program on a Server System.

the user system

At initialization time, SYCOM issues a "remote file information" request to obtain, from the Network Controller, the Logical Station Numbers (lsn's) for the virtual terminals assigned to the system. These numbers are placed into a table which maps the actual "lsn" of each virtual terminal into a virtual "lsn" with which SYCOM can reference the corresponding virtual terminal.

When a user wishes a connection to a Server System, a ".CONNECT" command is entered to the User System's SYCOM. SYCOM finds the "next" available entry in its "virtual lsn table" and transmits a "connect message" to the specified Server System. This control message contains a virtual terminal address. Upon receipt of the "connect reply message", SYCOM replies with the results.

12 Sep 1978

#### Software Operational Guide

Page - 33

Section II Concepts and Facilities

(Original Version)

During the life of a connection, all data messages entered from a virtual terminal are sent to the Server System for processing. (Excluded are any local SMCS commands which are prefixed with the assigned signal character.) This is accomplished by doing a remote file write to the Controller using the terminal's virtual terminal "lsn". In processing this message, the Controller maps the virtual terminal lsn into the assigned virtual terminal address (two characters). The message is then transmitted to the Server System, according to SYCOM protocol, with the virtual terminal address. The terminal user can then "sign on" to any application program that is supported in the Server System; providing the capability is inherent in the form of a MCS.

Any incoming message to the Network Controller from a SYCOM line which contains a virtual terminal address (e.g., "Aa", "Ab", etc) is sent to SYCOM. These messages are written into SYCOM's remote file (LINK) by the Controller with a device address aporopriate for the terminal. SYCOM upon reading the message, searches the virtual terminal address table for the actual system "lsn" for that terminal. SYCOM then puts the actual "lsn" into the reply message and writes it into the Network Controller's remote file (station queue) for distribution to the originating terminal.

In the case where the virtual terminal is actually an application program, the reply message is sent to that program via the queue mechanism for "program-to-SYCOM" communications.

When the user wishes to disconnect from the Server System, a ".DISCONNECT" command is entered to local SYCOM which transmits a "disconnect message" to the Server System, specifying the virtual terminal address to be disconnected. The terminal is notified of this action and is then free to establish another connection or work pattern as desired. (Virtual terminals are automatically disconnected when the system operator enters the ".EOJ" command.)

the server system

The Server System's Network Controller intercepts "connect" and "disconnect" messages (unlike all other control message which are routed to SYCOM). Un receipt of a "connect message", the Network Controller verifies that the addressed virtual terminal exists, checks to see if it can support another connection, and that some interactive application program is ready to accept input from it. If all is okay, the Network Controller returns a "connect reply message" indicating successful completion; else the Controller sends back an error response in a "connect reply message".

Software Operational Guide

Page - 34

Section II Concepts and Facilities

(Original Version)

If the connection is successful, the Server System makes this virtual terminal appear as a terminal local to that system (but not yet "signed-on").

During the life of the connection, any input data received by the Server System's Network Controller for an enabled virtual terminal is conveyed to the Waiting application program thru its remote file (could be the SMCS). This message transfer transpires in exactly the same manner as for local terminals. Any output data generated by the Network Controller is transmitted to the User System, with the virtual terminal address specified.

The Server System upon receipt of a "disconnect message", takes necessary action to detach that virtual terminal and respond to the User System with an appropriate "disconnect reply message".

Remote Supervisory Console (RSC)

It is an inherent capability of the system to utilize the SPO, an application program, or a terminal in a User System as a "Remote SPO" to a Server System. This only occurs when, as Figure II=6 shows, a copy of the SMCS is running in the Server System. For this system, the SPO is also referred to as a "Remote Supervisory Console", or simply the RSC. This term was selected to imply capabilities which transcend normal capabilities of a SPO. This interface infers communication (via the SMCS) with a Server System's MCP, hence the term, Remote Supervisory Console.

Messages input from the RSC to SYCOM are grouped into two basic types; Local SYCOM and Remote MCP. Local SYCOM input includes commands to initiate file transfers, request SYCOM status data, and other such functions. Additionally, respond to conditions within SYCOM's domain that cannot be preprogrammed; such as card reader not ready, clear peripherals, etc..

Remote MCP commands permitted from a RSC are basically directives used for control over the execution of jobs in the Server System; all via the Server System's SMCS. Only certain MCP commands are permitted from a RSC (reference Appendix F, "Restricted MCP Command Summary") for processing on a B1800/B1700 Server System. Should the Server System be other than a B1800/B1700 series computer, refer to the appropriate operating system reference manual for a list of MCP commands permitted from a Remote SPD.

12 Sep 1978

Page = 35

Software Operational Guide

Section II Concepts and Facilities

(Original Version)

1 . . . . . . . . MCP SPOIRSC 1 \_\_\_\_\_ 12 1 . . . . . . . . . . A 1 1 1 A 1 12 1 1 . . . . . 11 1 >. AP/RSC . . 00000000 . 1 000000000 . 1 111 1 L 1 1 1 1 1 1 1 V v 1 3 . . . . . . . . . >. Network . SYCOM . . Controller . 1 The User System 1 10 1 1-1 1----1 1 VDT/RSC I DIS I ----1 The Server System -1 9 .<---. SYCOM .--->. Network I. 1 --. Controller .--1 D/S 1 • 1 - - - - - | 1 . . . . . . . . . 51 A A A 1 1 8 V 1 1 1 1 SMCS 1 1 1 . ŧ 1 1 Note: The numbers 1-12 1 1 represent the 1 . . . . . . . . . . 6 1 1 sequence of data 1 . **A** L V 1 7 ۷ movements. v MCP

Figure II-6: RSC Input/Output Message Flow

12 Sep 1978

Software Operational Guide

Page = 36

(Original Version)

Section II Concepts and Facilities

Output messages for RSC display can be grouped into four types; they are: Local SYCOM, Local MCP, Remote SMCS, and Remote MCP. Local SYCOM output messages are normally responses to commands but can also be error type messages reporting SYCOM problems. Local MCP output would include such messages as "NO PRINTER AVAILABLE" and other such information messages which pertain to operations of a SYCOM system. Remote SMCS messages would be replies to SMCS commands entered for processing by the Server System's SMCS. Remote MCP output messages include SPO type output resulting from jobs run on a Server System. These messages are routed back to the User System's SYCOM program for display.

In the case where the RSC is actually an application program, the response message is directed to that program via SYCOM's "program to SYCOM" queue mechanism.

Referring again to figure II=6, assume for example, that a user wanted to execute a job on the Server System. Assume also that the subject program and all data requirements reside in the Server System. The appropriate command string is entered to the User System's SYCOM, via the SPO/RSC. SYCOM accepts this message from the MCP, formats and transmits it to the Server System; via the Network Controller. (The command string could also be entered from either a local terminal or application program. Although these message paths are slightly different, the end results are the same.)

In the Server System, the SMCS receives the MCP command message (from the Network Controller) and "zips" it along to the MCP. The Server System's MCP processes the request as asked. All output messages pertaining to that job are given to the SMCS which in turn transmits them to SYCON in the User System. User System SYCOM then causes these messages to be displayed its SPO/RSC.

From this we can see SYCOM'S involvement and the processing flow involved in this type of operation. All preceding functions described for the RSC are over and above the standard User SPO input/output messages used for normal User System operations.

Since SYCOM is symmetrical, both systems can perform an identical set of functions; providing of course, both systems are of the same series. Actual input and output messages are shown and exemplified in Section IV, "System Operating Procedures".

12 Sep 1978

Software Operational Guide

. .

Page = 37

Section II Concepts and Facilities

(Original Version)

Program-to-SYCCM Communication

The same range of functions are possible for the program-to-SYCOM facility as for terminal-to-program communications. This feature is made possible by an adaptation of the terminal-to-program scheme. A mechanism (of implementation dependent nature) is provided which allows an application program in the User System to masquerade as a "terminal". This program formats messages to appear as if they were terminal keyboard input. These messages are then written into SYCOM's "INQ" queue for processing.

From this point on, at least until a reply message is processed, processing is the same as terminal-to-program commands. When SYCOM receives a reply message, it is written to its "OUTO" queue where the application program in turn requests a "read" on this queue to obtain the message. These reply messages must also be formatted exactly as those created for terminal display; because they could be destined for terminal output. The application program can then either print, store, ignore, parse, or display select message text as appropriate for the program.

The schematic on the following page (Figure II-7) provides an overview of the mechanics for "Program-to-SYCOM communications": Communication between the application program and SYCOM is thru a pair of simple "queue files". The program gives a command to SYCOM by "writing" into SYCOM's "DUTQ" queue file. It obtains a response message by "reading" from SYCOM's "INQ" queue file.

To convey certain basic information with these input/output messages, a "header" is required which must contain specific information. This header is defined in Section VI; "Appendix E, Message Formats". There are no command restrictions enforced for this method of SYCOM/system control. However, this feature is controlled by the SYCOM parameter "PROGRAM". For this feature to be available, that parameter "must be" specified when the SYCOM parameters file is created.



The Server System

MCP 1 1 1 1 1. 1 1. 1 1 1 1 . ] . ]----1 1 1 . SYCOM .<-->. Network • Controller •----I D/S I 1 . 1 . . . . . . . . . 1----1 1 1 1 1 A SNCS .<-----AP

Figure II-7: Terminal-to-SYCOM schematic

12 Sep 1978

Software Operational Guide

Page - 39

Section II Concepts and Facilities

(Original Version)

PRCGRAM-TO-PROGRAM COMMUNICATIONS

The program-to-program capability employs an entirely different concept for inter-system communication. As previously discussed, this feature is an inherent by-product of the system. SYCOM itself is in no way involved with the program-to-orogram message traffic and therefore, does not provide any services to either program. The only exception to this is that SYCOM does "approve" a virtual terminal position for the program communications path.

As shown in Figure II-8 (below), communications are directly thru the Network Controllers via a special "remote file" declared in each Controller; explicitly for that purpose. Both application programs must have mirror images of these remote files (one each) declared as well. This remote file is shown in both Controller examples (Section V) as well as both sample application programs (Appendix I).

Figure II-8 can be used to visualize how an application program in one system can communicate with a program in another system. The input/output path is for the program to accept input messages from the Network Controller and return output reply messages for transmission to the other system/program. It would behave the developer of these application programs to design a specific set of messages which exclude any terminal connctations. This could greatly simplify development efforts extended for these programs.

The User System The Server System MCF . 1 . . . . . . . . . . . . . . . . . . 1 1 1 MCP . 1 ....... . . . . . . . . . I . AP .. - SYCOM - I 1 1 1 . . 1 ł 1 1 . . . . . . . . . . . . . . . . . . 1 Α 1 1 1 1 AP 1 1 1 1 1 1 ł ŧ A ----- Network Ŧ 1 Network i . Controller .<----I . 1 \ 1 . . . . . . . . . ->. Controller 1 1 . . . . . . . . . . . . . . . 1

Figure II-8: Program-to-Program Schematic

12 Sep 1978

Software Operational Guide

Page - 40

Section II Concepts and Facilities

(Original Version)

To convey certain required basic information with these input/ output messages, a "header" is required and must contain specific information. This header is defined in Section VI, "Appendix E, Message Formats". There are no restrictions as to the text in these messages but the length is controlled to a degree be the "BUFFERSIZE" parameter. This parameter is in SYCOM's "parameters file" and can be specified at initialization time.

The usefulness of this feature is to be evaluated and decided upon by the system user. The capability is there to be utilized for whatever purpose.

### NETWORK CONTROL

The SYCOM User System is normally considered to be the remote RJE Station and must initiate the transmission of data to the Server System. In a point-to-point non-switched network, this is accomplished automatically by enabling the line adapter and then exchanging handshaking control messages with the Server System.

In a point-to-point switched network, the User System must call the Server System. This is accomplished by either dial-out (automatic dialing by the computer) or by manual dial-up (dialing performed by the system operator). The dialing is done "after" the adapter is enabled. When the line connection is established, the exchanging of handshaking control messages with the Server System is carried out. In a point-to-point non-switched network utilizing a direct connection, no dialing is done; or possible.

Once this communication link has been established with a Server System, line communication is identical for the switched and nonswitched networks. (The User System and the Server System must be compatible as to transmission rate and other such optional hardware capabilities.)

## Conversational Communication Procedure

The following notes refer to the basic conversational line discipline employed by SYCOM. They apply to a point-to-point circuit, either dedicated or switched:

12 Sep 1978

Software Operational Guide

Section II Concepts and Facilities

(Original Version)

Message Transfer: All message blocks are delineated by special characters. Messages begin with a "SOH" character and end with a "BCC" character. The message control format is as follows: SOH, ARM, IM#, SIX, DA1, DA2, message text (blocked records), ETX, and finally the BCC.

Retransmission of Messages: If the transmitting system does not receive a reply to its last transmission within a given period of time, the unacknowledged message block is re-transmitted.

Parity or Block-Check Error Handling: If the receiving system receives a recognizable message, but it contains either a parity or block-check error, it replies with a "NAK". The transmitting system should retransmit the message. This is repeated "n" number of times ("n" being user specified), and if the error persists, either system can initiate error recovery.

Message Idle State: If the receiving system receives a "good" message, but has no message traffic to send, it responds with an "ACK". If the sending system has more message traffic, it sends another message block. If the sending system has no more message traffic, it proceeds to the "message idle state".

Normal Termination: The normal termination of a conversation occurs when one system sends an "ACK" (indicating that it has nomore message traffic), and the system receiving the "ACK" also has no more message traffic. The system receiving the "ACK" proceeds to the "message idle state".

Point-to-Point Contention Procedure

The essential characteristic of this procedure is that after the end-of-message transfer, as indicated by the transmission of the "DLE-EOI" sequence, both systems are in an idle state. Either system may then bid for master status by sending the "DLE-ENQ" sequence. This is normally initiated by the entry of a command to SYCOM.

Master Status: If either system is in the idle state and wishes to begin a message transfer, it may bid for master status by sending the "DLE-ENQ" sequence. If the same system had just terminated a message transfer by sending a "DLE-EUT" sequence, then there is a delay created before this system again sends the "DLE-ENQ" sequence.

12 Sep 1978

Software Operational Guide

Page - 42

(Original Version)

Section II Concepts and Facilities

Begin Message Transfer State: If the system bidding for master status receives an "ACK" (indicating ready to receive), it may commence message transfer immediately. If the response is "NAK" (not ready to receive), the "DLE-ENO" sequence may be repeated up to a retry limit of "n" ("n" being user specified), and if no valid response is received, commence error recovery. If the system bidding for master status receives the "DLE-ENO" sequence (indicating that the other system is bidding at the same time), it abandons its own bid and replies in accordance with its readiness to receive. In order to avoid any line collisions when both systems are bidding at the same time, the no-response timeout values must be set to different values. This is true whether not the systems are "like" or "unlike".

Termination: The "DLE-EDT" sequence is transmitted by the system directed to terminate to indicate the end of its session, and if using a switched line, to disconnect the modem.

### Transmission of Packed Data

As in previous versions of this protocol, data records within a message block are separated by "RS" characters. Repeated occurrences of blank characters are compressed out using the "ESC-integer" mechanism. The method of representing non-printable characters has changed with SYCOM's implementation. Printable characters are those listed in "Appendix B, Character Set", with the exception of the "Line Feed" and "BELL" characters.

If the data records contain only printable characters, then the first character of the record "is not" a "DLE" character. If the data record contains any character not in the "character set", then the first character of the record must be a "DLE" character, and the remainder of the record contains translated data. The original data is translated such that every 4-bits is mapped into an 8-bit printable character. This mapping is accomplished by internal translations so that each 4-bits of data is prefixed with a 4-bit hexadecimal value "4". (Actual data transmissions occur in ASCII character mode.)

12 Sep 1978

### Software Operational Guide

Page = 43

Section II Concepts and Facilities

(Original Version)

NETWORK CONTRULLER

The Network Controller provides interfaces with the User System's terminals and Server Systems at the hardware level. It is responsible all for direct communication with those terminals and Server Systems. It performs the actual data transmission (both transmitting and receiving) functions between two systems. The Network Controller accepts data message blocks and system control messages from SYCCM and certain application programs (program-toprogram) and transmits them to the Server System. In reverse, it receives all messages from a Server System and forwards them as required for handling control and distribution. The SYCOM program relies upon the MCP for the basic function of "message queuing" tetween it and the Network Controller and certain application programs. This facility provides the message paths between SYCOM, the SMCS, application programs, and the Network Controller.

Following is a normal sequence of events which transpires between SYCOM, the Network Controller, and the system's terminals:

In response to a "poll" by the Network Controller, a terminal transmits a message back to the Network Controller. The Network Controller accepts the message and appends a header, supplying pertinent information about the message, (e.g., terminal type, lsn, etc.). The Network Controller then determines which program to send this message to and writes it into the appropriate program's remote file (input queue). (Assume in this case it is SYCOM.) SYCOM reads its remote file (input queue) and obtains an input message. Should the message be a SYCOM command, it is processed and a reply message created. The reply message is then started down the line by SYCOM when it writes the output message into its remote file to the Network Controller. The Controller in turn reads the reply message and transmits it (devoid of header of course) to the originating terminal.

The description on the relationships of the SMCS and the Network Controller with SYCUM, which is also considered an application program, apply to any application program. Thus, any application program can enjoy the benefits of a sophisticated terminal network with message/terminal/program management thru the SMCS and Network Controller. In summation, the SMCS is the message controller for the system. It also provides a myriad of functions which direct Network Controller functions as well as special purpose MCP functions.

12 Sep 1978

Software Operational Guide

Page - 44

(Original Version)

Section II Concepts and Facilities

CONFIGURATIONS AND SOFTWARE ASSOCIATIONS

The Systems Communications Module has been designed to operate in either of three modes; 1) as an individual entity "without" local terminals, 2) as an individual entity, "with" local terminals, and 3) in conjunction with the Supervisory Message Control System (SMCS); also with terminals. Referring to Figures II-9, II-10, and II-11 (which follow), system configurations with supporting software elements are shown; and describes.

Entry Level System

This system configuration provides all of the SYCOM capabilities previously defined which includes the SPO as a virtual terminal. This is accomplished with the smallest memory size commitment. Although the command repertoire is generally confined to entry from the system SPO, special application programs running in the system can alternately inject commands to SYCOM for processing.

In the User System schematic which follows (Figure II=9), the communication links between system software elements are shown. Each of these links, "queues" and "remote files", have specific purposes which are briefly identified as follows:

- A. "INQ" is a queue file used by SYCOM as output to an application program utilizing the "program-to-SYCOM" facilities.
- B. "OUTQ" is also a queue file used by SYCOM as input from an application program utilizing the "program-to-SYCOM" facilities.
- C. "LINK" is a remote file which is used in communications with a remote Server "host" System. All RJE and virtual terminal input/output message traffic moves thru this file. This remote file is directly related to the SYCOM communications line to a remote host.
- D. "RFUSER" is a user defined name for a remote file used for program-to-program I/D message traffic, primarily directly thru the Network Controllers. All messages for this remote file move thru the same communications line as those identified for the "LINK" file.

BIBOOVBITOO SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page - 45
Section II Concepts and Facilities	(Original Version)

		•		•	
800 or B1700 User	System		/		
	aster Control	Program (MCP	•••••	• • • • • • • •	•
A 1	•••••	•••••	•••	• • • • • • • •	A 1 1
1 . Progr 1 SYC 1 I .	am to . DM .	• Program to • Program	•		
I (INQ) I I V V	••••• { !	A	••	• • • • • • • •	1 V
	Q)	(RFUSER	·)->. ·>.	Network Controll	• ler •
• • • • • • • • • • •			••	• • • • • • • •	

1--1 Communications 1 line to Server System.

1

12 Sep 1978

Software Operational Guide

Page - 46

(Original Version)

Section II Concepts and Facilities

Extended Level System

This system configuration also provides all SYCOM services as outlined for the Entry Level System and extends this to include local terminals. With terminals, SYCOM functions can be invoked from outside the computer room and away from the system SPO. Terminals on the system increase memory usage in proportion to the number of terminals attached to the system and the line or lines for terminals. The memory increase however, will be in the Network Controller program and not SYCOM itself.

Local terminals are exclusively owned and controlled by SYCOM. Although there is no operational SMCS required in a User System, it is "required" in a B1800/B1700 Server System when Remote SPO type activities are needed. The introduction of terminals in this system requires another remote file declared in both SYCOM and the local Network Controller:

A. "TERMINALS" is the remote file used by SYCOM for local terminal input/output message traffic between it and the Network Controller. As the controller handles all necessary terminal interfacing, SYCOM only reads input from and writes output to this remote file.

12 Sep 1978

Software Operational Guide

Page - 47

Section II Concepts and Facilities

(Original Version)

SPOZESC B1800 or B1700 User System Master Control Program (MCP) Δ 1 Ł ł Program to . · Program to · 1 SYCOM . Program 1 Ì. 1 (INQ) ..... L 1 1 v ۷ t . 1 ---(RFUSER)->. .<-(OUTQ)--. Network <-----(TERMINALS)-----SYCOM ---->. Controller . ----(LINK)-----Communication line to Server System. VDT's

12 Sep 1978

Software Operational Guide

Page - 48

Section II Concepts and Facilities

(Original Version)

Ultimate Level System

Using this configuration, not only are all of the capabilities provided as outlined for both Entry Level and Extended Level systems, it now provides the SYCOM terminal user with an access to all of the capabilities afforded by the Supervisory Message Control System (SMCS). Memory increases in this configuration are directly related to the SMCS and the Network Controller.

Local terminals in this system configuration are under direct control of the SMCS. Upon command (SMCS "sign on"), a terminal is temporarily "attached" to SYCOM for functional directives. When SYCOM activities are complete, the terminal is "detached" from SYCOM with the ".BYE" command and is again under the control of the SMCS. As such, an additional terminal I/O communications link is needed. Also, in addition to this link, the remote file "TERMINALS" is effected somewhat. Both are described as follows:

- A. "MCSREMDIE" is a remote file used by the SMCS as a link to the system's terminals. All terminal I/O message traffic between SMCS and the Network Controller passes thru this file. During the time a terminal is "signed" on" to SYCOM, certain messages are "switched" to the SMCS for processing (those messages prefixed with a SMCS "signal character").
- B. "TERMINALS" is SYCOM's remote file for terminal input/ output message traffic between it and the Controller. The slight difference being that terminal input/output only passes thru this file for those terminals that are "signed-on" to SYCOM.

Software Operational Guide

Page - 49

Section II Concepts and Facilities

(Original Version)

	tan an a		. SPO/RS	с.
31800 or 8170	n User System		······ / /	••
<ul> <li>I</li> <li>I&lt;</li></ul>	Master Co	ontrol Progra	am (MCP)	• • • • • • • • • • • • • • • • • • • •
	• • • • • • • •	• • • • • • • • • • •	• •••	A A I I V I
	<ul> <li>Program</li> <li>to</li> <li>SYCOM</li> </ul>	<ul> <li>Program</li> <li>to</li> <li>Program</li> </ul>	>. S	• • • • • • • • • • • • • • • • • • •
1 1 (INQ) 1 1 1 1 V 1			(MCSREMOTE)	v
.   .   . Sycum   .	.<-(UUTQ) .<(1	(RFI	USER)>.	Network . Controller .
. 	•	_ I N K )	•	· -///
			/ //	// / /
			/ / 	Communication line to Serve System.
		•	••••••	

Figure II-11: User System Schematic (SMCS; terminals)

12 Sep 1978

Software Operational Guide

Page - 50

Section II Concepts and Facilities

(Original Version)

MESSAGE PATH ASSIGNMENTS

In order to describe the potential message communication paths opened up by the virtual terminal capability, a special message path syntax was developed to pictorially present this myriad of possibilities. The message path syntax is easy to understand as one can visualize these logical paths thru the User and Server Systems. System software elements are easily recognized by the widely used acronyms (all defined in the "Glossary"). Although the directional flow arrows "==>" indicate a single direction, communications are two-way. The boxed in syntactical elements to the right indicate an alternate message path; also available thru the "virtual terminal" capability.

For the Entry Level System

SPO input:

For an "Entry Level System" (Figure II-9) there are three message paths for SPO input. The following examples explore each of these possibilities:

		User S	ystem	Server	System	
SP0> MCP>	SYCOM	:	NC			:
		:	-	NC	> SYCOM	:
. <xxx>&gt; <xxx></xxx></xxx>	SYCOM			   > syco	M> SP0	

As shown in this diagram, all input command strings are given to User SYCOM by the MCP. Messages prefixed with a period "." are processed by User SYCOM. All other input messages are sent to Server System's SYCOM which displays them on that SPD. When the "virtual terminal" capability is invoked (by using the ".CONNECT" command), then the following message paths are established:

12 Sep 1978

Software Operational Guide

S

Page - 51

Section II Concepts and Facilities

(Original Version)

				User	System	I S	erver	System	1	
	- CUNNECT	>	SYCOM			+     				
מח	> NCU		SYCOM	:	NC -		• • • • •		• • • • • • •	•••
ru.	> MCF		SICUM	:	NC -	/	NC	>	SMCS	:
	. < X X X >	>	SYCOM	• • • • •	• • • • • •	   	• • • • •	••••	• • • • • •	•••
	< X X X > ? < X X > ? < nnn >					>	SMCS SMCS	> >	SMCS MCP MCP	
						1				

If there is no SMCS currently running in the Server System, then the message string is intercepted by the Network Controller which causes an error message to be displayed on that system's SPO.

When a SMCS is running in the Server System, the SPO operator can either "sign=on" to a program in that system or initiate RSC type commands; thru the SMCS. When the virtual terminal is signed=on to the specified application program, it is "attached" to that program. All further non-prefixed input is sent to that program for processing. (In the example below, the "\$" character is assumed to be the Server System's SMCS "signal character"; it could be another character.)

				<u>User</u>	Syste	m 1 +	Server	System	
S	IGN ON	AP	> SYC	<b>G</b> M	•	\$			
SPO -	-> MCP		SYCOM	:	NC		• • • • • •	•••••	•••••••••••••••••••••••••••••••••••••••
				:			- NC	> \->	AP : SMCS :
•	< X X X > : X X X >	>	SYCOM	••••				>	ар.
5 5 5	<xxx> ?<xx> ?<xx> ?<nnn></nnn></xx></xx></xxx>					>	SMCS SMCS	>	SMCS MCP MCP

B1800/B1700 SYSTEMS COMMUNICA	TIGNS MODULE		12 Sep 1978
Software Operational Guide			Page - 52
Section II Concepts and Facil	ities	(Origin	al Version)

Program-to-SYCOM input:

For "Program-to-SYCOM" communications, the following message path assignments are available:

			an an the second se	User	System	l Server	System	
						1		
AP		>	SYCOM -	:	NC			:
				:		/ NC	> SYCOM	:
•	. <xxx></xxx>	>	SYCOM	• • • •		••••••••••••••••••••••••••••••••••••••	• • • • • • • • • • • • •	••
	< X X X >		ann ann ann ann ann ann ann ann ann	1 446 446 446 446 146 1	* * * * * * * * * *	> SYCO	M> SPO	

All input command strings are given to User SYCOM by the MCP via the "OUTO" queue. Messages that are prefixed with a period "." are processed by User SYCOM. All other input messages are sent to Server System's SYCOM which displays them or that SPO.

When the "virtual terminal" capability is invoked (be entry of a "CONNECT" command), the following message paths are established:

User System 1 Server System

.CONNECT --> SYCOM

١P		>	SYCOM -	•••••• : > N	IC		••••		• • • • • • •	••
						/	NC	>	SMC S	:
	• < X X X >	>	SYCOM			• • • • •   		••••		
	< X X X > ? < X X > ? < nnn>	******				>	SMCS	>	MCP MCP	

12 Sep 1978

Software Operational Guide

Page - 53

rage 33

(Original Version)

Section II Concepts and Facilities

If there is no SMCS currently running in the Server System, then the message string is intercepted by the Network Controller which causes an error message to be displayed on that system's SPO. When a SMCS is running in the Server System, this program can "sign-on" to a program in that system or initiate RSC commands; thru the SMCS. When this virtual terminal (program) signs-on to a specific application program, it becomes "attached". All nonprefixed input is sent to that Server program for processing. (In the example below, the "\$" is assumed as the Server System's SMCS "signal character"; it could be something else.)

User System | Server System SIGN ON AP ----> SYCOM 1 1 : : SYCOM ----> NC AP : ---> : 1 ----> AP NC : ----2 1-> SMCS 2 : I ---> SYCOM •<XXX> ł < X X X > > AP \$<XXX> > SMCS \$ ? < X X > ---> MCP \$?<nnn> -----> SMCS ---> MCP

# Program-to-program input:

The message path for program-to-program communications is limited to the following predetermined path. Messages can be any format and contain any text agreed upon by both application programs. The only prevailing restrictions are maximum "buffersize" and all messages must be prefixed with a "header" ("Appendix E").



. . . 1

12 Sep 1978

Software Operational Guide

Page -54

Section II Concepts and Facilities

(Original Version)

For the Extended Level System

In this configuration (Figure II-10), the SPO and all terminals attached to, and owned by, SYCOM normally communicate with local SYCOM. When the virtual terminal capability is utilized, these devices are permitted to communicate with either the system software programs or appropriate application programs in a Server System.

The SPO, program-to-SYCOM, and program-to-program input message paths for this software configuration are the same as those defined for the "Entry Level System". The below message path assignments are for local terminals attached to this system:

User System 1 Server System VOT NC SYCOM --->

• < X X X > SYCOM ---->

By entry of the ".CONNECT" command to SYCOM the virtual terminal capability is invoked. The following message paths are then established for the terminal:

. CONNECT ----> SYCCM

				• • • • • • • •	• • • • •				
VDT> NC	>	SYCOM	>	NC					
			:		/	NC	>	SMCS	:
• < XX X >	· · · · · · · · · · · · · · · · · · ·	SYCOM			1	• • • • • •	• • • • • •	••••	•••
< X X X> ? < X X > ? < nnn >						SMCS SMCS	>	MCP MCP	

12 Sep 1978

#### Software Operational Guide

Page = 55

Section II Concepts and Facilities

(Original Version)

If there is no SMCS currently running in the Server System, then the message string is intercepted by the Network Controller which causes an error message to be displayed on that system's SPO.

When a SMCS is running in the Server System, the terminal user can either "sign-on" to a program in that system or initiate RSC type commands; thru the SMCS. When a virtual terminal signs-on to a specified application program, it is "attached" to that program. All further non-prefixed input is sent to that program for processing. (In the example below, the "\$" character is assumed to be the Server System's SMCS "signal character"; it could be another character.)

	User System I Server System
	<b>I</b>
JIGN UN AF	I
VDT> NC> SYCOM -	: > NC :
	: / : NC> AP :
	: \-> SMC2 :
. <xxx>&gt; SYCOM <xxx>&gt;</xxx></xxx>	I AP
\$ < X X X >	SMCS> SMCS
\$? <nnn></nnn>	SMCS> MCP

## For the Ultimate Level System

In this configuration (Figure II-11), all local terminals are attached to, and owned by the SMCS. Therefore, the normal message communication path is to the User System's SMCS. When the virtual terminal capability is utilized, these terminals are permitted to communicate with either system software programs or appropriate application programs in the Server System. Additionally, these same terminals are permitted to communicate with any appropriate application program in the User System.

12 Sep 1978

Software Operational Guide

Section II Concepts and Facilities

The SPO, program-to-SYCOM, and program-to-program input message paths for this software configuration are the same as those defined for the "Entry Level System". The following message path assignments are for all local terminals attached to the User System's SMCS. First the normal message paths for terminal input within the User System are presented:

User System | Server System | VDT --> NC --> SMCS |

<XXX> ----> SMCS
?<XX> ----> SMCS --> MCP
?<nnn> ----> SMCS --> MCP

In order for the terminal user to gain access to the system's SYCOM, the terminal user must enter the SMCS "SIGN ON" command as shown below. (Although this example is for a sign-on to SYCOM, it could be any appropriate application program in the User System.)

1

1

ŧ

SIGN ON SYCOM ----> SMCS

The following message path assignments are then in effect for all terminal input.

VD T NC SYCON ----> NC -----1-> SMCS : 1 : ---> SYCOM 1 -----NC 1 . < X X X > SYCOM 1 < X X X > SYCOM ---> SPO ---> \* < X X X > --> SMCS Ł \* ? < X X > ----> SMCS MCP ---> 1 \*?<nnn> ----> SMCS MCP

Page = 56

(Driginal Version)

12 Sep 1978

Software Operational Guide

Page - 57

(Original Version)

Section II Concepts and Facilities

The "virtual terminal" capability is invoked when the terminal user enters the ".CONNECT" command to SYCOM. Following are the message paths now established for the terminal; if a SMCS is not currently running in the Server System:

				Use	r System	1.	Server	Syste	m 	
	• CONNEC	T	> SYCOM			! ! !				
VDT	> NC	> \->	SYCOM SMCS	:	> NC	/	- NC	>	SMCS	•••
	• <xxx> <xxx> * <xxx> *?<xx> *?<xx> *?</xx></xx></xxx></xxx></xxx>	> > >	SYCOM SMCS SMCS SMCS		 мс р мс р	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		>	SMCS	••

If there is no SMCS currently running in the Server System, then the message string is intercepted by the Network Controller which causes an error message to be displayed on that system's SPO.

When the "virtual terminal" capability is invoked and there is a SMCS currently running in the Server System, then the following message paths are established for the terminal:

			· .			1 <b>1</b>				
VD T	> NC	>	SYCOM SMCS	••• : 	> NC	• • • • • • • •		• • • • • • • • • • • • •		•••
- - - -		\->		:			- NC	>	SMCS	
				• • •						
	• < X X X > < X X X >	>	SYCOM			<b> </b>	SMCS			
	? < X X > ? < nnn >						SHCS SHCS	>	MCP	
	* < X X X > * ? < X X >	>	SMCS	>	MCP	1				
	*? <nnn></nnn>	>	SMCS	>	MCP	1				

81800/81700 SYSTEMS	COMMUNICATIONS	MODULE		12 Sep 1978
Software Operational	Guide			Page - 58
Section II Concepts	and Facilities		(Origin	al Version)

Section II Concepts and Facilities

With the SMCS running in the Server System, a terminal user can either "sign-on" to a program in that system or initiate RSC type commands; thru the SMCS. When a virtual terminal is signed-on to a specified application program, it is "attached" to that program until the terminal user "signs-off". All non-prefixed input is sent to that application program for processing. (In the examples below, the "\*" character is assumed to be the User System's SMCS "signal character" and the "\$" character is assumed to be the Server System's SMCS "signal character"; they can be different.)

> User System I Server System 1

· · · · · · · · · · · · · · · · · · ·	
an a	
/DT> NC> SYCOM> NC N-> SMCS : /	:
: NC> AP : \-> SM	: : : :
• <xxx>&gt; SYCOM 1</xxx>	
* <xxx>&gt; SMCS   *?<xx>&gt; SMCS&gt; MCP   *?<nnn>&gt; SMCS&gt; MCP  </nnn></xx></xxx>	
<xxx>&gt; AP \$<xxx>&gt; SM</xxx></xxx>	CS
\$? <xx>&gt; SMCS&gt; MC \$?<nnn>&gt; SMCS&gt; MC</nnn></xx>	P P

QUEUE DEPTH CONTROL \_\_\_\_\_\_

In the SYCOM operational environment, there is a MCP controlled data path which exists between SYCOM and the Network Controller. This data path is referred to as the "remote file" (or message queue) mechanism. Due to differences in data communication line transmission rates and peripheral input speeds, this remote file Without some kind of control on the message becomes a buffer. traffic thru this file, it could grow beyond reasonable limits or be jammed full.

12 Sep 1978

Software Operational Guide

.

(Original Version)

Page = 59

Section II Concepts and Facilities

To provide a control over the number of message blocks that can be queued up in this file at any given time, a "marker message" system is employed. As SYCOM is putting message blocks into this file, a count is maintained. When this count reaches the halfway point of the queue depth limit, a marker message is put into the file. The marker message is a specially prepared data message with a code in the message header. This special code ensures recognition by SYCOM when this message is echoed back to from the Network Controller.

Once a marker message is placed into the file, SYCOM continues putting message blocks into the file until the predetermined queue depth limit is reached, at which time peripheral input operations are suspended. When the Network Controller encounters the marker message during the transmission of the queued message blocks, it acknowledges receipt of a marker message by returning a "flag" in the corresponding "ACK" message to SYCOM. When SYCOM receives this acknowledgement, it resumes peripheral input, again places a marker message into the file and continues putting message blocks into the file until the predetermined queue depth is reached or until an eng-of-file condition, whichever occurs first.

SYSTEM FILES

In the total system picture, there are many files and file types. However, as this system overview pertains only to SYCOM and its relationships, only those pertinent files which are required for SYCOM operations are presented. Following are file descriptions, by software element, that are necessary for SYCOM operations:

MASTER CONTROL PROGRAM (MCP) FILES

(SYSTEM)/USERCODE File

This file contains all valid usercode-passwords that are known to the system. It is used by the MCP to enforce file security disk files and programs/users that access them. With this file, the MCP can service any requests by programs regarding validity of specific usercode-password combinations.

Software Operational Guide

Section II Concepts and Facilities

(Original Version)

12 Sep 1978

Page = 60

NETHORK CONTROLLER FILES

Network Information (NIF) File

This file contains tables that describe the physical and logical attributes of the network. The initialization values for lines, terminals, and file tables of Network Controller are contained in this file. This file is created by the NDL compiler when it is in the process of generating a Network Controller.

## SYCOM FILES

The Systems Communications Module utilizes many types of files in the management of its domain. There are, peripheral (includes disk), remote communications, and queue files. These files are briefly described here, under separate headings, with detail descriptions for major files available in the appendices.

### Peripheral Files

These files are generally used to input/output RJE type data and include the card reader, card punch, line printer, magnetic tape, and disk files. Also there is a disk file which is used by SYCOM to manage the various options and settings. Following are general descriptions of the SYCOM files which fall into this category:

### A. PARAMETERS File

The PARAMETERS file is a single record file containing required operational parameters for SYCOM. It is static and can only be changed by removal and recreation. This file is created by SYCOM upon the entry of appropriate parameters from either the card reader or the User SPO.

### 8. Card Input File

This file is used to input RJE card data files from the system's card reader. It can be either a 80-column or 96-column device.
12 Sep 1978

Software Operational Guide

Page = 61

Section II Concepts and Facilities

(Original Version)

C. Card Output File

This file is used to punch RJE card output data files to the system's card punch. It can be either a 80-column or 90-column device, as appropriate.

D. Printer File

This file is used to print all print files that are being, or have been, received from a Server System. The line printer device can by any one of the various models which can be attached to the system. The device can be either a 120-position or 132-position line printer.

Remate Files

The remote files are used in communications between SYCOM and the Network Controller and SYCOM and either the terminals or SMCS. There are two of these files which are described as follows:

A. LINK File

The "LINK" file is used for all input/output messages relating to RJE and file transfers; specifically, station types for Control, SPD, Cards, Printer, and File records.

8. TERMINALS File

The "TERMINALS" file is the communication link between SYCOM and the Network Controller for local terminal messages. This remote file is however, only used when there are local terminals; controlled by either the SMCS or SYCOM. B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 62

Section II Concepts and Facilities

(Original Version)

Queue Files

The queue files are utilized by SYCOM for buffered communications between itself, the MCP, and the Network Controller. The purpose of each of these cueue's is described as follows:

A. RJE Queue

Used to save printer/punch data received from the other system when the User System's printer or punch is busy.

B. LINK Queue

- Used to save input messages from the other system during the "initialization" phase; else they may be lost in the shuffle.
- Used to save output messages for the other system when "messages queued" exceeds "MAX-MESSAGES".

#### C. INQ Queue

This queue is the mechanism by which SYCOM sends reply messages to an application program dealing in programto-SYCOM communications. SYCOM "writes" into it and the application program "reads" from it to obtain messages.

## D. OUTQ Queue

This queue is the mechanism whereby SYCOM receives input command messages from an application program that deals in program-to-SYCOM communications. The application program "writes" into it and SYCOM "reads" from it to obtain input messages.

12 Sep 1978

Software Operational Guide

Page - 63

Section II Concepts and Facilities

(Original Version)

RESTRICTIONS

The following restrictions are valid for the system based on a logical evaluation of the system's requirements:

- Maximum message block size = 1950 bytes
- Minimum message block size = 78 bytes
- Maximum simultaneous system connections = 1
- Maximum message block queue cepth = 255
- Maximum number events queued to process = 99
- Maximum number of terminals supported = 15
- Maximum number of card readers supported = 1
- Maximum number of card punches supported = 1
- Maximum number of line printers supported = 1
- LOAD.DUMP tapes are not supported for file transfer operations.
- Between unlike systems, only file transfers of code files and 8-bit data files are allowed.
- Only "data type files" are file transferable between disk/tape and tape/disk (codes files are invalid).
- Multi-point networks are not supported by SYCON; at this time.
- Message blocking is prohibited. (A message block already contains multiple records or data images.)
- Monitoring and controlling of programs in the Server System requires a SMCS executing in that system.
- For systems using "binary-synchronous" transmission lines, the NDL generated Network Controller requires certain changes in the area of "NAK" conversions.

These restrictions may be changed during subsequent development cycles, but are however, currently in effect.

## SECTION III

COMMANDS

### Software Operational Guide

Section III Commands

Page - 65

(Original Version)

## SECTION III

\_ \_ \_ \_ \_ \_ \_ \_ \_ \_ \_ \_

COMMANDS

# GENERAL .

The commands available with the Systems Communications Module (SYCOM) are described in this section. These commands are used to give SYCOM directions as to specific actions which must be executed in order to perform the required tasks. The commands are presented alphabetical order within two groups, those allowed from terminals and those restricted to entry from the system SPO or program. This way, the terminal user need not be encumbered with the details of SPO commands; which are not permitted.

To ensure that each command is physically separated, a unique format presentation, which is described below, has been developed for this document. This scheme of formatting presents commands in modular format which allows for expansion and change without significant disruption of existing syntax.

## COMMANDS

\_\_\_\_\_\_

Each statement in the Command Language consists of a command and, in some cases, a list of parameters. The parameters in the list are separated from the command and each other by blanks. In the following examples single spaces are used although the use of multiple spaces between command elements could be used.

When the SYCOM has finished processing a command, it displays a reply message either confirming successful completion of the command or indicating an error. Errors terminate the processing of a command; the entire command must be rementered correctly. B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page = 66 Section III Commands (Original Version)

COMMAND SYNTAX CONVENTIONS

The notation conventions that are described below are used throughout this section to describe the syntax of each command.

Notation used in model format

First, a model of a command is presented. This gives the user an overview of the potential of the command structure.

.

•

:

BRACKETS: Information that is enclosed within brackets [] represents an option which may be used or omitted by the user. In the model example given above, the options USER, AS or ONTO, and "file-name-2" indicate three possible choices when using these options, or the choice of not using any option.

BROKEN BRACKETS: Left and right broken <> brackets are used to indicate that the enclosed variable within the brackets must be replaced by digits and/or letters whose value is determined by the context of the command.

UPPER CASE LETTERS: Words that appear in upper case letters indicate that the word is a literal and must be substituted verbatim, when used. In the model given above, the word .COPY would be entered first when constructing a command statement.

LOWER CASE LETTERS: Words that appear in lower case letters indicate that a value must be supplied. In the model above, the word "file-name-1" indicates that a character string representing the name of the file to be copied must be entered.

12 Sep 1978

Software Operational Guide

Page - 67

Section III Commands

(Original Version)

COMMAND STRING SYNTAX CONTINUATION: A few of the commands have a considerable number of syntactical items which overflow a printed line into two or more command string lines in the presentation. To provide a clear line of continuation for the command string, the use of line direction arrows are used. A line direction arrow appears as ----> and always flows left to right; to continue the command string. It is easy to follow a command string in this format as it is logical in rature.

END OF STATEMENT: The end of a command statement is indicated by the appearance of a % (percent sign). It is only a syntactical string terminator and is not required text for command input.

OPEN-CLOSE PARENTHESES (): When these characters appear in a command syntax statement they are required input. See the "COPY" command for a use reference.

SLASH /: When this character appears in a command syntax statement it is required input. A space or spaces on either side of this character are not allowed.

Punctuation Rules

The space, the comma, and the period are considered punctuation marks (except in literals) and are used as follows:

- The space (blank) is a separator, required after words and literals in the absence of any other separator. A space may precede or follow any other separator, and many spaces are the same as one (except in literals).
- 2. A comma is a separator that is legal only where it is specifically indicated in the command syntax. The commawhere it is legal, can also serve as a terminator for words and numeric literals. The comma is never required, except to separate criteria in a command string.
- 3. The period is a required prefix to local SYCOM commands. It is never required within any command string syntax. Its only acceptability would be as an embedded character in a file name, system identifier, or usercode/password.

Software Operational Guide

Page = 68

Section III Commands

(Original Version)

COMMAND RELATED ERROR MESSAGES

Error messages specific to the following SYCOM commands appear in Appendix A (Command Error Messages).

COMMAND SET

The command set which follows, is used to direct SYCOM operations from either the SPO, a system terminal, or an application program running in the system. Non-SYCOM commands are also permitted to direct User and Server System operations via respective SMCS's. This command set is separated into two subsets; 1) commands which are permitted from any device and 2) commands restricted to a SPO or special application program.

Terminal Commands

The following commands are permitted from terminals. They direct file transfer and remote Server System operations. Although these commands are for terminals, they are also permitted from the SPO and special application programs used for system control. However when these commands are entered from the SPO, the command string must be prefixed with the "job number" assigned to SYCOM followed by an "AX" and then the command string.

12 Sep 1978

Software Operational Guide

Page - 69

ABORT .

Section III Commands

(Original Version)

• A BOPT

The ABORT command is used to force SYCOM to "abort" Function: an in-process file transfer, either receiving or transmitting. If "fetching" (receiving), the file created up to the point of aborting is either removed or purged as appropriate. .ABCRT FETCH % Syntax: PUT Semantics: FETCH = causes the FETCH in-process to be aborted. PUT = causes the PUT in-process to be aborted. De fault: None. (Kill off transmitting the current file.) Example 1: .ABCRT PUT Input: ---Reply: --- PUT OF FILE <file name> ABORTED

Example 2: (Stop receiving of incoming file transfer.) Input: --- .AEGRT FETCH

Reply: --- FETCH OF FILE <file name> ABORTED

Note: If a file transfer is not in-process when the command is entered, an appropriate error message is returned.

12 Sep 1978

Software Operational Guide

Page - 70

Section III Commands

(Original Version)

•8YE

Function: This command completely logs the current user off SYCOM. It is valid from all stations including the system SPO and application programs. The following sequence of events transpire when it is entered. A "BYE" text message is sent to the other system to cause a "log-off" from that system, the usercode is removed, an automatic "DISCONNECT" is initiated, and a "detach" message is sent to the local SMCS; if the terminal is not owned by SYCOM.

Syntax: .BYE %

Semantics: None.

Default: None:

Example:

Input: --- .BYE

Reply: --- GCODBYE: (usercode) LOGGED OFF SYCOM 11:45:32.1

Note: The initiating station need not be "logged-on".

12 Sep 1978

Software Operational Guide

Page - 71

Section III Commands

(Original Version)

• CONNECT

Function:

The CONNECT command is used when a terminal user wishes to be logged—on to a Server System. Upon entry, the User System goes into "online" mode, attempting to establish communications with a Server System.

If a connection currently exists to a host Server System, the existing connection is utilized to attach the terminal to that system.

Syntax:



Semantics:

message string = this optional message string can be either a text character string or a valid usercode/password for a program's log-on.

Default: When no "message string" is specified, the user's usercode/password is forwarded to the other system if available; else blanks are sent.

Example: (Request attachment to the Server System.) Input: --- .CGNNECT US MY/USER Reply: --- ".CGNNECT" INITIATED

Note: If the Server System is accessed via a "switched" line and its telephone number has been previously established by entry of the "FHONE" command, then this command initiates an automatic dial-out.

CONNECT

Software Operational Guide

Section III Commands

Page = 72

(Original Version)

• C O P Y

Function: This is the command which is used to initiate "file transfers". The command syntax provides for the transfer of files "from" the initiating system to another system (a PUT) and from another system "to" the initiating system (a FETCH). The files can be in either magnetic tape or disk form and the only restrictions that apply are those identified in "Restrictions", Section II.

Syntax: .COPY <file=name=1> [ USER=<usercode=1>, ----->

---> PASSWORD=<password-1> ) ] -----

---> [ AS <file name-2> [ USER=<usercode-2>, --> ONTO

---> PASSWORD=<password-2> ] ] ----->

---> FROM [<volume-1>] ( [KIND=PACK ,] ---TAPE TAPE7 TAPE9 PETAPE

---> HOST=<system name-1> ) TO [<volume-2>] -----

---> (CKIND=PACK ,1 HOST=<system name-2>) % TAPE TAPE7 TAPE9 PETAPE

Note:

An important point to remember is that the "parens" are REQUIRED syntactical elements as well as the keyword "FROM"...

12 Sep 1978

73

Software Operational Guide

Section III Commands

.CCPY (continued)

(Original Version)

Page -

Semantics: file-name-1 the actual name of the file to be = transferred; including multifile identifiers, if any. usercode-1, the usercode/password values which = identify the user for purposes of password-1 verifying access rights to the specified file. means create a file with the name AS specified as "file-name-2". ONTO = means write over any existing file with the name specified as "file-name-2". file-name-2 = a name to be given to the file that is transferred, at its destination. usercode-2> the usercode/password values which · = password-2 specify the security to be placed on the transferred file once the transfer has been completed. the "pack-id" from where the file volume-1 = to be transferred can be found; providing the following KIND option is equal to "FACK". This option is not required for files on magnetic tape or system disk. KIND one of the keywords shown which is used to describe the peripheral device from where the file is to be transferred "from"; if other than a system disk. specifies the "system-id" from system-id-1 =

from.

COPY

where the file is to be transferred

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide 74 Page -Section III Commands (Original Version) .COPY (continued) \_\_\_\_\_\_ the "pack-id" to where the file is transferred; providing the Semantics: volume-2 -(continued) following KINC option is equal to "PACK". This cption is not required for files on magnetic tape or disk. KIND <u>.</u> one of the keywords shown is used to describe the peripheral device to which the transferred file is to be written; if other than a system disk. the "system-id" of the system which system-id-2 🚔 is to receive the file. Defaults: The syntactical item "VOLUME" defaults to "system disk" if "KIND" is also defaulted. The syntactical item KIND defaults to "KIND=PACK". The default for "usercode-1" and "password-1" are those of the logged-on user. The destination usercode/password (usercode=2 and password-2) are defaulted to those of the source usercode/password.

> The destination file name (file-name-2) defaults to the original name of the file transferred and the "AS"/"ONTO" option defaults to "AS".

Options:

The syntactical abbreviation "DISK" may be used in lieu of "KIND=PACK".

The syntactical abbreviation "USER=<usercode>/ <password>" may be substituted for the specified command syntax elements so named.

12 Sep 1978

Software Operational Guide

Page - 75

Section III Commands

(Original Version)

•COPY (continued)

Example 1: (Transfer file "XY7" from system A to system B.)

Input: --- .COPY XYZ (USER=<usercode>/<password>) FROM (DISK, HOST = USER1) TO (DISK, HOST=SERVER1)

Peply: --- FILE XYZ START TRANSMISSION --- FILE XYZ END TRANSMISSION

Example 2: (Copy file from another system changing file name.)

Input: --- .COPY DMPALL AS DUMPALL FROM (HOST = SERVER1)

Reply: --- FILE DMPALL START TRANSMISSION --- FILE DUMPALL RECEIVED --- FILE DMPALL END TRANSMISSION

Notes:

 In the 7.0 implementation for file transfers from either tape or disk to tape in a receiving system, if a scratch tape is not mounted and/or ready, SYCOM will "hang" until the tape is made available; no notification is made of this condition and no input messages/commands are processed.

2. If a "usercode" is not wanted as a multifile-id to the file name in the receiving system, enter "file-name-1" prefixed with an asterisk "\*".

12 Sep 1978

Software Operational Guide

Page - 76

Section III Commands

(Original Version)

• D I SCONNECT

Function: This command must be used when the terminal user wishes to terminate a session with a host Server System. The terminal is logged-off of the Server System when a "BYE" text message is sent to the other system which is followed by a "disconnect" message. The terminal reverts back to its previous state of being signed-on to SYCOM. The terminal user is then free to enter more commands to SYCOM or sign-off (.BYE).

Syntax: .CISCONNECT %

Semantics: None.

Default: None.

Example:

Input: --- .DISCONNECT

Reply: --- STATION DISCONNECTED

Note: The terminal is logically disconnected but may or may not be physically disconrected. To the terminal user however, the net effect is the same.

-----

12 Sep 1978

77

FETCH

Software Operational Guide

Section III Commands

(Original Version)

Page -

•FETCH

Function: This command is the shorthand version of the "COPY" command for use in specifying explicit "FETCH" file transfer requests. The command syntax provides for transferring files from another system "to" the initiating system (a FETCH). Files can be either magnetic tape or disk (in any combination) and the only restrictions that apply are those identified in Section II, "Restrictions".

syntax:	•FE1UH	<110	e-name-1> L(U)	ER= <usercode=1< th=""><th><b></b></th><th></th></usercode=1<>	<b></b>	
	> P	ASSWO	RD= <password=1< th=""><th>&gt;)]</th><th></th><th></th></password=1<>	>)]		
	•					
	> [	AS Onto	<file name="2"></file>	[(USER= <user< td=""><td>code-2</td><td>!&gt;,</td></user<>	code-2	!>,
				- <b>-</b> -		
	> P	ASSWU	RD= <password=2< td=""><td>&gt;)]]</td><td></td><td></td></password=2<>	>)]]		
	> (	FROM	[ <volume=1>]</volume=1>	ECKIND=PACK TAPF	)]]	
				TAPE7		
				TAPE9		
				PETAP	E	
						5
		TO	[cuolumon()]		111	<b>4</b>
	>	.10	LING COMB - 273		/ 1 1	
				TAPE7		
· · · · · ·				TAPE9		
				PETAP	F	
			· · · · · · · · · · · · · · · · · · ·	- Park	-	

to the

to the file name on the receiving system, then prefix "file-name-1" with an asterisk "\*".

B1800/B1700 S	YSTENS COMMUNICATIONS MO	DULE 12 Sep 1978
Software Oper	ational Guide	Page - 78
Section III C	ommands	(Original Version)
•FETCH (conti	nued) =====	
Semantics:	Refer to the "COPY" com	mand semantics.
Defaults:	Refer to the "COPY" com	mand defaults.
Options:	Refer to the "COPY" com	mand options.
Example 1:	(Cbtain a copy of the " Server System.)	DMPALL" program from the
Input:	.FETCH DMPALL	
Reply:	FILE DNPALL START TRANS FILE DMPALL RECEIVED FILE DMPALL END TRANSMI	MISSICN

Note:

In the 7.0 implementation for file transfers from either tape or disk to tape in a receiving system, if a scratch tape is not mounted and/or ready, SYCOM will "hang" until the tape is made available; no notification is made of this condition and no input messages/commands are processed.

12 Sep 1978

Software Operational Guide

Page - 79

LIST .

Section III Commands

(Original Version)

•LIST

Function: This command's purpose is to cisplay information about the system and its users. Command options provide lists of those users currently "signed-on", terminals that are currently "online", and file transfers that are currently "scheduled" to occur.

Syntax:	.LIST	USERS	76
		TERMINALS	
		SCHEDULE	

Semantics:USERS= requests a list of all users currently<br/>"signed-on" to SYCOM.TERMINALS= requests a list of all terminals that<br/>are currently "online".SCHEDULE= requests a list of all file transfers<br/>currently scheduled.

Default: Currently there is no default..

Example 1: (List the current signed-on users of the system.)

Input: --- .LIST USERS

Reply: --- [<uc index>] DENNIS ONLINE --- [<uc index>] JEANNE OFFLINE --- [<uc index>] CARLYE ONLINE --- 3 LOGGED ON

Example 2: (List the currently "online" terminals.) Input: --- .LIST TERMINALS Reply: --- [<rsn>] <SPO> <v.lsn> Aa aA --- [<rsn>] <lsn> <v.lsn> Ab bA B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 80 Section III Commands (Original Version)

Example 3: (list all file transfers "scheduled" to occur.)

Input: --- .LIST SCHEDULE

Reply: --- SCHEDULE LIST NOT IMPLEMENTED [<number>/<count>]

Notes: 1. The above example lists are displayed in reverse order on the SPO unless the MCP option "DIRN" is set to the value "1".

2. Example 1, the "<uc index>" is Usercode Index.

12 Sep 1978

Software Operational Guide

Page - 81

Section III Commands

(Original Version)

•PUT

Function: This command is a shorthand version of the "CUPY" command and is used to specify explicit file "PUT" transfer requests. The command syntax provides for transferring files "from" the initiating system to another system (a PUT). The files can be either magnetic tape or disk (in any combination) and the only restrictions that apply are those identified in Section II, "Restrictions".

.PUT <file-name-1> [(USER=<usercode-1>, -----> Syntax: ---> PASSWORD=<password-1>) ] <file name=2> [ USER=<usercode=2>, ---> ---> [AS ONTO ---> PASSWORD=<password=2>) ] ] -15K ---> [FROM [<volume-1>] [ (KIND=PACK ) ] ] TAPE TAPE7 TAPE 9 PETAPE ---> [TO [<volume-2>] [ (KIND=PACK ) ] ] ] % TAPE TAPE7 TAPE 9

PETAPE

PUT .

Note:

If a "usercode" is not desired as a multifile-id to the file name on the receiving system, then prefix "file-name-1" with an asterisk "\*".

81800/81790	SYSTEMS COMMUN	ICATION	S MODULE		12 Se	ep 1978
Software Ope	rational Guide				Page	- 82
Section III	Commands			( )	riginal Ve	ersion)
.PUI (contin	ued) ====					
Semantics:	Refer to the	*COPY*	command	semant i	C S •	
Defaults:	Refer to the	"COPY"	command	defaul t	S•	
Options:	Refer to the	*COPY*	command	options	en ● Normal State	
Example 1:	(Send the pr execution.)	ogram "	TEST16"	to the S	erver Sys	te <b>n</b> for
Input:	.PUT TEST16					

Reply: --- FILE TESTIG START TRANSMISSION --- FILE TESTIG END TRANSMISSION

Example 2: (Send a printer-backup file to the "hub" system.)

Input: --- .PUT (SP0)/#2121

Reply: --- FILE (SPO)/#2121 START TRANSMISSION --- FILE (SPO)/#2121 END TRANSMISSION

- Notes:
- In the 7.0 implementation for file transfers from either tape or disk to tape in a receiving system, if a scratch tape is not mounted and/or ready, SYCOM will "hang" until the tape is made available; no notification is made of this condition and no input messages/commands are processed.
- 2. Referring to example 1; Upon completion of the "PUT" file transfer, the following message is displayed on the SPO of the "other" system:

% SYCOM =<job #> FILE TEST16 RECEIVED

This is also true under the reverse circumstances.

12 Sep 1978

Software Operational Guide

Page = 83

Section III Commands

(Original Version)

USER

Function: This command is valid from any source. Once the statement is entered, the "usercode/password" is stored; without validation. This data is forwarded to the other system for "CONNECT" commands and is used for file transfer purposes; as the default usercode/password when none is specified.

Syntax: .US[ER] <usercode>/<password> %

Semantics: usercode = a predefined user identification.

password = a predefined user security code.

Default: None.

Example: (Log-on to SYCOM.)

Input: --- .US MY/USER

Reply: --- USER MY/USER LOGGED ON

Note: If a "usercode"/"password" is already logged-on, the current log-on attempt is still allowed and takes precedence.

12 Sep 1978

Software Operational Guide

Page - 84

Section III Commands

(Original Version)

.WHAT

Function: The .WHAT command solicits the return of a reply message indicating the current "status" of SYCOM's processing activity. The reply message is variable in length and content to reflect as many conditions as possible. If any file is in the process of "transfer", its name and the number of records thus far transferred is disolayed.

Syntax: .WHAT %

Semantics: None.

Default: None:

Example 1: (Assume at the time this command is entered, SYCOM is connected and waiting for something to do.)

Input: --- .WHAT

Reply: --- SYCOM ONLINE; IDLE THIS STATION OFFLINE, LOCAL HOST = USER1, REMOTE HOST = SERVER1

Example 2: (The command is entered during a file transfer.)

Input: --- .WHAT

Reply: --- SYCOM ONLINE; OO FILES QUEUED; SENDING: DMPALL(70/82); NOT RECEIVING

Note: The context of a WHAT reply is extremely fluid in that it is representative of what is happening at the time the command is entered. Because of the number of combinations of events that could be in progress at any given time, it is not feasible to attempt to example them all. However, the examples above will give the user some idea as to what can be expected.

WHAT .

12 Sep 1978

85

Software Operational Guide

Section III Commands

(Original Version)

Page -

\*<command>

\*<SMCS Command String> (User System)

Function: This command syntax allows a terminal user on the User System to give commands to the system's SMCS for processing. Commands entered from a terminal that are prefixed with a signal character of "\*" are intercepted by the Network Controller and are immediately sent to the SMCS for processing. Consequently, the entire range of SMCS commands are available to a SYCOM user with this command syntax.

Syntax: \*<SHCS command> [<attributes>] %

Semantics:

SMCS command = any SMCS command which is valid from a User System terminal.

attributes = any data elements that are required by the command syntax.

Default: None.

Example:

Input: --- \*FILE (JEANNE)/GUIDE

Reply: --- '(JEANNE)/GUIDE' IS A PUBLIC FILE OF 188 RECORDS (90 BYTES, 20 PER BLOCK) IN 10 BLOCKS (10 PER AREA, 1 AREA) CREATED ON 04/24/78, LAST ACCESSED ON 06/21/78 (ACCESS SECURITY IN INPUT-OUTPUT) AND CURRENTLY IN USE BY O USERS

Note:

For a complete list and descriptions of the SMCS commands available, refer to the "SMCS Software Operational Guide".

12 Sep 1978

Software Operational Guide

Page - 86

\*? <XX> .

Section III Commands

(Original Version)

\*? <SMCS/MCP Command String> (User System)

Function: This command is used to give "safe" MCP commands to The User System's MCP for processing. This command string is message switched to the system's SMCS by the Network Controller which in turn "zips" the command string to the MCP for processing. The SMCS appends an appropriate usercode, password, and session number to the message "if" the user is currently "logged-on" to the SMCS; otherwise the command message string is zipped as entered.

Syntax: \*? <XX> <mcp command attributes> %

Semantics: XX = any MCP command which is valid from a work station. mcp command attributes = any data elements required by the command syntax.

Default: None.

Example:

Input:	*2 1	PD =/=	
Reply:	PD=	"DCHCODE"	112
	PD=	"DCHSOURCE	-
	PD=	"#DUMP"	
	PD=	"PROGRAMX"	
	END	PD	

Notes:

- The "mcp command string" is made secure, that is, in the above example, only the user's directory is returned.
  - Refer to "Appendix F, Restricted MCP Command Summary" for a list of those MCP commands which are permitted from a terminal.

12 Sep 1978

Software Operational Guide

Page - 87

Section III Commands

(Original Version)

<x><SMCS Command String> (Server System)

Function: This command syntax allows a terminal user on the User System to give commands to the Server System's SMCS for processing. Each command entered must te prefixed with the current signal character for that SMCS; it must be different than that of the User System's SMCS for obvious reasons. All messages with this signal character are intercepted by the Network Controller and immediately sent to the SMCS for processing. Consequently, the entire range of SMCS commands are available to a SYCOM user. (This command is only valid when a SMCS is running in the Server System.)

Syntax:	<x><smcs comm<="" th=""><th>and&gt; [<attributes>] %</attributes></th></smcs></x>	and> [ <attributes>] %</attributes>
Semantics:	<b>X</b>	<pre>= the Server System's current SMCS "signal character".</pre>
	SMCS command	= any SMCS command which is valid from a User System terminal.
	attributes	= any data elements that are required by the command syntax.

Default:

None.

Example:

Input: --- \$FILE (JEANNE)/GUIDE

Reply: --- '(JEANNE)/GUIDE' IS A PUBLIC FILE OF 188 RECORDS (90 BYTES, 20 PER BLOCK) IN 10 BLOCKS (10 PER AREA, 1 AREA) CREATED ON 04/24/73, LAST ACCESSED ON 06/21/78 (ACCESS SECURITY IN INPUT-OUTPUT) AND CURRENTLY IN USE BY 0 USERS

12 Sep 1978

Software Operational Guide

Page = 88

Section III Commands

(Original Version)

<x>? <SMCS/MCP Command String> (Server System)

Function: This command format allows certain MCP commands to be sent to a Server System for processing. They are permitted from either the User System's SPO or from one of its terminals. All replies to these commands are returned to the User System SYCOM which causes them to be displayed on a appropriate SPO/terminal. Certain MCP commands require a "usercode/password", prefixed to the desired MCP command, which is known to the Server System. To do these commands however, requires the SMCS operating in the Server System. See "Appendix F" for the list of safe MCP commands for a B1800/B1700 Server System. All commands that require a usercode/password are so indicated.

-------Syntax: <x>? [US <usercode>/<password>] ---------> <XX> <mcp command attributes> z Semantics: = the Server System's SMCS X signal character. = any MCP command code or XX iob # that would be valid from a terminal or SPO. mcp command attributes = any data elements required by the command syntax. Default: None. (Request a "PD" list for user currently logged-on.) Example: Input: ---\$? PD =/= Reply: ---PD= "TESTER" PD= "OPENS" END PD In the above example, the "\$" Note: character is assumed to be the Server System's SMCS "signal <x>? <XX> . character".  $= T_{p_{\rm eff}} \sim 10^{-11} {\rm eV}$ 

12 Sep 1978

Software Operational Guide

Page - 89

Section III Commands

(Original Version)

System SPO/Program Commands

The following SYCOM commands are restricted to entry from either the system SPO or a special application program. These commands are generally of the type used by a system operator to control functions relevant only to operation of the system. They are generally functions a terminal user cannot logically direct without current knowledge as to the state of the system and its peripherals. By enforcing this restriction, the system is assured that control is at the system level and not at a terminal level.

Program command input does not require the command string to be prefixed with SYCOM's "job number" and an "AX". Command input from a program should be in the same format as command input from terminals.

B1800/B1700 S	SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Oper	ational Guide	Page - 90
Section III (	Commands	(Original Version)
<b>C1</b>		
• • • • • • • • • • • • • • • • • • •		
Function:	The .CL command is used to "cl device in the event the device when receiving a data file. I	ear" a peripheral becomes inoperable t clears (discards)
	the buffers associated with th and the receiving process is t	e specified device erminated.
Syntax:	•CL <udi> X</udi>	
Semantics:	uci = the Unit Device Identif one of the following mn	ier is restricted to emonic codes:
	LP = Line Pri CP = Card Pun	nter ch
Default:	None:	
Example 1:	(Clear the line printer becaus	e its down.)
Input:	<sycom #="" job=""> AX.CL LP</sycom>	
Reply:	(no response)	
Example 2:	(Clear the card punch because continue.)	there is no need to
Input:	<sycom #="" job=""> AX.CL CP</sycom>	
Reply:	(no response)	
Notes:	None.	

•

12 Sep 1978

Software Operational Guide

Page - 91

CP

Section III Commands

(Driginal Version)

•CP

Function: The purpose of this command is to inform the user System that the systems Card Punch has fallen into a "not ready" state. SYCOM in turn informs the Server System (if such connection is established) of this hardware deficiency condition. The Server system would "suspend" transmitting any card punch data stream messages; if it was in this process.

> It is a responsibility of the User System operator to inform SYCOM (via the appropriate command) when the card punch is again "ready" to receive output (see notes below).

Syntax: .CP NR %

Semantics: NR = Not Ready

None.

Default:

Example: (Inform SYCOM that the Card Punch is not ready.) Input: --- <sycom job #> AX.CP NR

Reply: --- (no response)

Notes: 1. To inform SYCOM that the "card punch" is now ready to receive output data, the ".RY CP" is provided.

> If the card punch has become "inoperative", then the ".SV CP" command is more appropriate.

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 92 Section III Commands (Original Version)

•DEBUG

Function: The "DEBUG" command provides the facilities for a printer trace of all data messages read or written by SYCOM. The "ON" option will initiate the DEBUG printer tracing, the "OFF" option will terminate tracing, and when no option is specified a request for the current status of the DEBUG facility is assumed.

Syntax: •DEBUG <default option> % ON OFF

Semantics: Self explicit.

Default: If no option is specified or the option is invalid, (not ON or OFF) a "status request" is assumed.

Example 1: (Turn printer debugging "ON" for all I/O messages.)

Input: --- <sycom job #> AX.DEBUG ON

Reply: --- % SYCOM =<job #> DEBUG TURNED ON

Example 2: (Obtain current DEBUG status.)

Input: --- <sycom job #> AX.DEBUG

Reply: --- % SYCOM =<job #> DEBUG ON

Example 3: (Turn debugging OFF.)

Input: --- <sycom job #> AX.DEBUG OFF

Reply: --- X SYCOM =<job #> DEBUG TURNED OFF

Notes: None.

12 Sep 1978

Page - 93

#### Software Operational Guide

Section III Commands

(Original Version)

•E0J

a de la compañía de l

Function: The ".EOJ" command provides the capability for an orderly shutdown of SYCOM. Any further commands are prevented (by rejection) from being processed. A message ("PLEASE SIGN OFF") is broadcast to all terminals currently "logged-or" to SYCOM. A "close message" is sent to the other system's SYCOM after the completion of any in-process file transfers and when there are no longer any terminals signed-on to SYCOM. The SYCOM program then sends a "disconnect message" to the Network Controller and goes to "end-of-job".

Syntax: .EOJ %

Semantics: None to specify.

Default: None.

Example: (Entering the command to shutcown SYCOM.)

Input: --- <sycom job #> AX.EOJ

Reply: --- % SYCOM =<job #> PLEASE SIGN CFF --- % SYCOM =<job #> EDJ INITIATED --- % SYCOM =<job #> SYCOM NOW CFFLINE

Note: All file transfers which are scheduled to occur (queued) are permitted to transpire.

B1800/B1700 SYSTEMS	COMMUNICATIONS	MODULE 12 Sep 1978
Software Operational	Guide	Page - 94
Section III Commands		(Original Version)

•LIST

Function: This command is valid from any terminal or the SPO. Its purpose is to display information about the system and its users. Command options provide lists of current system "parameters", users currently "signed-on", terminals that are currently "online", and file transfers currently "scheduled" to occur.

Syntax: .LIST PARAHÉTERS Z USERS TERMINALS SCHEDULE

Semantics: PARAMETERS =	requests a list of the parameters SYCOM is currently operating with.
USERS =	<pre># requests a list of all users currently #signed=on" to SYCOM.</pre>
TERMINALS	<pre>= requests a list of all terminals that are currently "online".</pre>
SCHEDULE :	requests a list of all file transfers currently scheduled.

Default: Currently there is no default..

Example 1: (Display the list of SYCON's parameters.)

Input: --- <sycom job #> AX\_LIST PARAMETERS

Reply:	% %	SYCOM Sycom	= <job =<job< th=""><th>#&gt; #&gt;</th><th>TERMINAL RECORD SIZE Debugging</th><th>.=</th><th>180 OFF</th><th></th></job<></job 	#> #>	TERMINAL RECORD SIZE Debugging	.=	180 OFF	
	z	SYCOM	= <job< td=""><td>#&gt;</td><td>PROGRAM INPUT</td><td></td><td>ALLOWED</td><td></td></job<>	#>	PROGRAM INPUT		ALLOWED	
	%	SYCOM	= <job< td=""><td>#&gt;</td><td>FILE TRANSFER</td><td></td><td>ALLOWED</td><td></td></job<>	#>	FILE TRANSFER		ALLOWED	
	7	SYCOM	= <job< td=""><td>#&gt;</td><td>FILE SEPARATORS</td><td></td><td>SENT</td><td></td></job<>	#>	FILE SEPARATORS		SENT	
	X	SYCOM	= <job< td=""><td>#&gt;</td><td>MAXIMUM BUFFER SIZE</td><td>· -</td><td>400</td><td></td></job<>	#>	MAXIMUM BUFFER SIZE	· -	400	
	2	SYCOM	= <job< td=""><td>#&gt;</td><td>WAIT</td><td></td><td>FALSE</td><td></td></job<>	#>	WAIT		FALSE	
	%	SYCOM	= <job< td=""><td>#&gt;</td><td>TIMEOUT INTERVAL</td><td>=</td><td>25</td><td></td></job<>	#>	TIMEOUT INTERVAL	=	25	
	%	SYCOM	= <job< td=""><td>#&gt;</td><td>SYSTEMID</td><td>=</td><td>"USER1</td><td></td></job<>	#>	SYSTEMID	=	"USER1	

LIST

12 Sep 1978

Software Operational Guide

Page - 95

LIST .

(Original Version)

Section III Commands

•LIST (continuec)

Example 2: (List the current signed-on users of the system.)

Input: --- <sycom job #> AX.LIST USERS

Reply: --- % SYCOM =<job #> [<uc index>] DENNIS ONLINE --- % SYCOM =<job #> [<uc index>] JEANNE DFFLINE --- % SYCOM =<job #> [<uc index>] CARLYE UNLINE --- % SYCOM =<job #> 3 LOGGED ON

Example 3: (List the currently "online" terminals.)

Input: --- <sycom job #> AX.LIST TERMINALS

Reply: --- % SYCOM =<job #> [<rsn>] <SPO> <v.lsn> Aa aA --- % SYCOM =<job #> [<rsn>] <lsn> <v.lsn> Ab bA --- % SYCOM =<job #> 2 TERMINALS IN USE

Example 4: (list all file transfers "scheduled" to occur.)

Input: --- <sycom job #> AX.LIST SCHEDULE

Reply: --- % SYCOM =<job #> SCHEDULE LIST NOT IMPLEMENTED [<number>/<count>]

Notes: 1. The above example lists are displayed in reverse order on the SPC unless the MCP option "DIRN" is set to the value "1".

2. Example 2, the "<uc index>" is Usercode Index.

DI BAA / DI TAA SYSTEMS	COMMUNICATIONS	40010 C		070
DTOAN DTUAN 2121FW2	CUMMUNICATIONS	MUUULL	IS 26D I	710
C	0			
sortware operational	. Sulde		rage -	70
Section 111 Lommands			curiginal versi	on)
영영훈계 전통 영향관 감정 않았다. 또는			동안 이 있는 것이 있는 것이 있는 것이 있다. 2013년 1월 19일 - 19일 - 19일 - 1	

.L.OG ====

Function: The purpose of this command is to display counts of specific data communication line errors and certain message transmission statistics. These counter are "not" zeroed out after they are displayed. The log effect is produced when these counters (identifiers included) are printed in the "DEBUG trace" listing (if the DEBUG option was "ON").

.LOG Syntax: 2

Semantics: None.

Default: None.

Example: (Request a display of the status counters.)

<sycom job #> AX.LOG Input: ---

Reply: ---

```
% SYCOM =<job #> TIMEOUT = <ccunt>,
                   BREAK = <count>,
                   END-OF-BUFFER = <count>,
                   LOSS-OF-DSR = \langle count \rangle,
                    LOSS-OF-CARRIER = <count>.
                    PARITY = <count>,
                    BUFFER OVERFLOW = \langle count \rangle_{\bullet}
                    MEMORY PARITY = <count>,
                    NUMBER OF TRANSMISSIONS = <count>.
                    NUMBER OF BYTES SENT = <number>.
```

Note:

Expect the above reply message to be displayed as a single message string.
12 Sep 1978

Software Operational Guide

Page - 97

LP

Section III Commands

(Original Version)

•LP

Function:

The purpose of this command is to inform the User System SYCOM that the line printer has become "not ready". SYCOM in turn informs the Server System of this temporary condition. The Server System then suspends transmitting of the line printer data stream messages until notified to resume by the User System's SYCOM.

It is the responsibility of the User Systems operator to inform SYCOM when the line printer is again ready to receive output data (see notes).

Syntax: .LP NR 7

Semantics: NR = Not Ready

Default: None.

Example: (Inform SYCOM of a "not ready" Line Printer.)

Input: --- <sycom job #> AX.LP NR

Reply: --- (no response)

Notes:

- To inform SYCOM when the line printer is ready to accept output data, the ".RY LP" command is provided.
- 2. If the line printer has become "inoperative", then the command ".SV LP" is more appropriate.
- 3. This command is only relevant to RJE functions. In particular, when the line printer is in use printing a RJE printer backup file.

#### Software Operational Guide

Page - 98

#### Section III Commands

(Original Version)

•OFFLINE

Function:

The purpose of the ".OFFLINE" command is to inform a Server System that the User System is going to "treak" the existing connection and go into standby mode; waiting for the Server System to call back. This command can be used for either a "switched" or "non-switched" line connection. Since the Network Controller actually performs this action (upon SYCOM directive), it then continuously monitors the "data set ready" signal. When this signal comes "true", the Network Controller re-establishes the connection with the system and normal operations then resume.

Syntax: .OFFLINE %

Semantics: None.

Default: None.

Example:

Input: --- <sycom job #> AX.OFFLINE Reply: --- % SYCOM =<job #> OFFLINE INITIATED --- % SYCOM =<job #> SYCOM NOW OFFLINE

Notes:

None.

12 Sep 1978

Page - 99

Software Operational Guide

Section III Commands

(Original Version)

.ONLINE (establishment)

Function:

The purpose of this command is to cause the Network Controller to attempt to re-establish the line connection to the Server System. Upon satisfactory completion of this task, a response message is displayed (see example below).

This command may be used in conjunction with the retry function. The error message "RETRIES-UP" is displayed when the current message buffer being sent to the Server System is not being received. By entering this command, the message tuffer being sent is discarded and an attempt is made to reestablish the line connectior.

Syntax: .CNLINE Z

Semantics: None.

Default: None.

Example:

Input: --- <sycom job #> AX.ONLINE

Redly: --- % SYCOM =<job #> ".ONLINE" INITIATED --- % SYCOM =<job #> SYCOM NOW ONLINE to REMOTE HOST

Notes:

None.

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 100 Section III Commands (Original Version) -PHONE ===== The PHONE command is used to either "establish" a Function: phone number to be used for the dial-out to the predefined Server System or to display an existing number. The automatic dial-out (with an Auto Call Unit) to a specific Server System requires a phone number, hence the need for the command. • PHONE <phone number > Syntax: Z <null option> phone number = the phone number of a target Server Semantics: System. null option = no option specified. = clears existing phone number. Default: The omission of a valid phone number in the command string causes any existing phone number previously entered to be "displayed". Example:1: (Establish the auto dial phone number of a specific Server System.)

Input: --- <sycom job #> AX.PHONE 1-805-964-6881

Reply: --- (no response)

Note: If an auto-dial unit is supplied, the specified phone number is used the next time dial-out is initiated to establish a connection to a Server System. (Dial-out is automatically initiated by the ".ONLINE" command.)

12 Sep 1978

Software Operational Guide

Page - 101

QS.

#### Section III Commands

(Original Version)

•QS ====

Function: This command allows a user to set a maximum number of message buffers that can be queued at any one time between SYCOM and the Network Controller. The value range for this parameter is 4 thru 15. (For a description on the use of this parameter see the writeup on "Queue Depth Control" in Section II.)

Syntax: .QS=<queue size> %

Semantics: queue size = a 1 or 2 digit integer representing the new Queue Depth limit. A suggested value for this parameter would be 10 message buffers.

Default: None.

Example: (Change the queue size to 10.) Input: --- <sycom job #> AX.QS=10 Reply: --- % SYCOM =<job #> QUEUE SIZE CHANGED TO 10

Note: A specified value outside the defined value range produces an error response.

12 Sep 1978 Page - 102

Software Operational Guide

(Original Version)

Section III Commands

•READ

Function: The ".READ" command causes SYCOM to open an input card file and commence transmitting those card data images to the Server System; providing a connection has been established. When end-of-file is reached, the file is closed automatically.

> This command is only processed when the User System is properly logged-on to a Server System which is defined as a "host". (It is possible to override this feature by using the ".RYCR" command instead.)

Syntax: .READ %

Semantics: None.

Default: Assumes the input card file is labeled "RJE/CARDS".

Example: (Start transmission of an RJE card data file.)

Input: --- <sycom job #> AX.READ

Reply: --- % SYCOM =<job #> CARD READER OPENED --- % SYCOM =<job #> CARD READER CLOSED

Note: An input card file is defined to be any group of cards with the first card in the deck containing "<I> STREAM RJE/CARDS" and the last card containing "<I> TERMINATE RJE/CARDS". (The <I> symbol represents an invalid character in column 1 as required for MCP control. B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 103 Section III Commands (Original Version) .RY (ready). \_\_\_\_\_ This command is used to inform SYCOM that the Function: specified peripheral device is currently ready to send/receive data. SYCOM in turn takes the appropriate action to advise the Server System. Syntax: .RY <udi> % Semantics: udi = the Unit Device Identifier is restricted to one of the following mnemonic codes: CR = Card ReaderCP = Card PurchLP = Line PrinterDe faul t: There are no defaults. Example: (Ready the line printer.) Input: --- <sycom job #> AX.RY LP Reply: --- (no response) Notes: None.

RY

이 같은 것이 같은 것이 같아요.	그는 것 같은 것 같은 것은 것을 많을 것 같아요. ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ?	아이는 물건에 전철하는 방법에 전화했다. 성격 전철 것
B1800/B1700 S	YSTEMS COMMUNICATIONS MODU	LE 12 Sep 1978
Software Oper	ational Guide	Page - 104
Section III C	ommands	(Original Version)
•SIZE		
Function:	This command allows the s maximum message buffer si receiving purposes. SYCD System of this change wit type message.	ystem operator to set the ze fcr transmitting and M then informs the Server h a special system control
Syntax:	.SIZE= <buffer size=""> %</buffer>	
Semantics:	buffer size = is a three limited to "400" bytes	digit integer with a value a rarge of from "078" thru •
Default:	None.	
Example:	(Reduce the send/receive to 320 characters.)	message block buffer size
Input:	<sycom #="" job=""> AX.SIZE=320</sycom>	
Reply:	% SYCOM = <job #=""> ".SIZE"</job>	INITIATED
Note:	This command applies only the SYCOM system.	to the RJE functions of

12 Sep 1978

Page - 105

#### Software Operational Guide

Section III Commands

.STOP

(Original Version)

Function:	This command provides the ability for an immediate shutdown of SYCOM. No further input messages are accepted as SYCOM goes to "end-of-job" immediately. Terminals are not logged-off, virtual terminals are not disconnected, nor is an inter-system connection gracefully disconnected.
Syntax:	STOP %
Semantics:	None to specify.
Default:	None -
Example:	(Entering the immediate shutdown command.)
Input:	<sycom #="" job=""> AX.STOP</sycom>
Reply:	% SYCOM = <job #=""> STOP INITIATED % SYCOM =<job #=""> SYCOM NOW OFFLINE</job></job>

12 Sep 1978

#### Software Operational Guide

Page - 106

#### Section III Commands

(Original Version)

• S V

Function: This command directs the User System to inform the Server System to cease transmitting the card punch or line printer data stream and to "save" the entire file at the Server System. The command permanently inhibits the transmission of any further card punch or line printer data streams until notified by the User System (see notes).

Syntax: .SV <udi> X

Semantics: udi = the Unit Device Identifier is limited to one of the following mnemonic codes:

> CP = Card Purch LP = Line Printer

Defaults: None.

Example: (Terminate the output to the Card Punch.)

Input: --- <sycom job #> AX.SV CP

Reply: --- (no response)

Notes: 1. To re-establish the card punch or line printer output data stream, the command ".RY CP" (for the card punch) or ".RY LP" (for the printer) must be used. This is used to inform the Server System to start sending any card punch or line printer data streams that are available for transmission.

2. All card punch and line printer data streams start at the beginning of the file.

12 Sep 1978

Page - 107

### Software Operational Guide

Section III Commands

(Original Version)

	I Local System I Input		I Remote System I Process		
1 Command	NO SMCS	SMCS	I Source	I NO SMCS	I SMCS 1
	,	ŧ = = = = = = = = I	+	t	+1
ABORT	Ves	l ves	1 TSP		
-BYE	yes -	l ves	I ISP		
I .CL	ves	l ves	I SP		
I .CONNECT	ves	lves	I TSP		1 00 1
I .COPY	yes	lyes	I TSP	l no	
I .CP	yes	lyes	I SP	l no	
I .DEBUG	yes	lyes	I SP	l no	I no I
I .DISCONNECT	yes	lyes	1 TSP	l no	l no l
I .EDJ	yes	lyes	I SP	l no	l no l
I .FETCH	yes	lyes	I TSP	l ves	l ves 1
I .LIST	yes	lyes.	I TSP	l no	I no I
I .LOG	yes	lyes	I SF	l no	1 no 1
1 .LP	yes	lyes	I SP	l no	I no I
I .OFFLINE	yes	l yes	I SP	l no	l no l
I .ONLINE	yes	lyes	I SP	l no	I no I
I .PHONE	yes	lyes	I SP	l no	I no I
I .PUT	yes	lyes	I TSP	l no	I no I
1.95	yes	lyes	I SP	l no	t no l
I .READ	yes	1 yes	I SP	l no	I no 1
1 .RY	yes	lyes	I SP	l no	ina i
I .SIZE	yes	l yes	I SP	l no	I no I
I .STOP	yes	l yes	I SP	l no	l no l
1 • SV	l yes	l yes	I SP	l no	I no I
I .USER	yes	lyes	I TSP	l no s	I no. I
I _WHAT	l yes	lyes	TSP .	l no	I no I
I * <user smcs=""></user>	l rio	l yes	I TSP	l no	t no t
I *? <user mcp=""></user>	no	l y e s	I TSP	l no	I no 👘 1
I <x><server smcs=""></server></x>	l no	l no	I TSP	l no	lyes l
I <x>?<server mcp=""></server></x>	no	l no	I TSP	l'no -	lyes I
1		t i se se se	1 ·	han an a	1 · · · · · · · · · · · · · · · · · · ·

Note: The above commands are restricted to entry from the source indicated in the table above.

Legend: T = Terminal S = SPO P = Program

SECTION IV

OPERATING PROCEDURES

B1800/B1700 SYSTEMS COMMUNICATIONS	MODULE	12 Sep 1978
Software Operational Guide		Page - 109

(Original Version)

SECTION IV

OPERATING PROCECURES

Section IV Operating Procedures

# INTRODUCTION

This section provides system and terminal operating procedures used for the operation of a SYCOM system with and without local terminals; and/cr in conjunction with the Supervisory Message Control System (SMCS). System operating procedures are slanted toward operation of SYCOM for the purpose of Remote Job Entry (RJE) type functions and the associated system control functions. These procedures present all aspects of SYCOM operations from initialization thru controlled shutdown.

Local terminal operational procedures are separated into two (2) categories; terminals under SYCOM control and terminals under the control of the SMCS. These procedures provide coverage of all terminal operations within the scope of SYCCM) from sign-on to sign-off. Actual input/output message text is used in the examples to provide a feel for the ease of operation and what can be expected by a user of the system.

# SYSTEM OPERATING PROCEDURES

The system operational procedures presented here are intended to provide a User System operator with sufficient information for operation of an interactive system using SYCOM. Operation of the associated Network Controller is briefly discussed. In addition, suggestions on the use of certain basic SYCCM commands and their options are exemplified.

12 Sep 1978

#### Software Operational Guide

Page - 110

Section IV Operating Procedures

(Original Version)

PRELIMINARY PROCEDURES

Before the SYCOM system is executed, the user needs to be aware of a few system variables that must be considered. First there are "program switches" which are used to invoke certain optional features. Second there is the "parameters" file which is used to maintain SYCOM's variable control functions and processing attributes. Each of these subjects in covered in turn starting with the program switches.

Program Switches

Program Switch 2 serves two purposes: 1) If "set" to the value of "2" at execute time, the initialization phase causes a display, on the SPO, of SYCOM's "compile time and date"; else no display. 2) If the switch is "set" after BOJ time, then the purpose is to cause any excess length I/O DEBUG trace lines to be truncated to exactly one print line to reduce printing; if the DEBUG option is "set" of course.

Program Switch 3 serves to expand the information message that is displayed at initialization time with a current "run date" (both time and date). Because this switch is used in conjunction with Program Switch 2, it must also be "set" prior to SYCOM execution time for switch 3 to be effective.

Preparation of Parameters Card File

Because SYCOM automatically uses an existing "parameters" file from disk, to enter changes to any existing file requires that the current "SYCOM/PARAMETERS" file be removed prior to program execution. This causes SYCOM to effect the process which creates a new parameters file based on; 1) a predetermined set of default parameters and 2) any parameters entered supersede the associated default parameter.

These parameters can be entered either from the card reader or the SPO. However, an important point to note, SYCOM will first attempt to accept parameters from a "card" file and failing to find this file, enters the mode whereby it accepts parameters from the system SPO. Therefore, if card parameter input is to be used, the card reader "must be readied" prior to execution.

12 Sep 1978

Software Operational Guide

Page - 111

Section IV Operating Procedures

(Original Version)

A simple way to create a parameters file is to punch a card deck; providing the target system has a card reader in its hardware configuration. The external name of this parameters card file is "CARDS". Following is an example of a SYCOM system card file: (The <I> in the example below represents an "invalid" character as required for MCP type commands.)

<I> DATA CARDS DEBUG BUFFER = 400 RECORD = 240 SYSTEMID = USER1 PROGRAM NOFS NOFTP WAIT TIMEOUTS = 25 <I> END

The above example shows "all" of the available SYCOM parameters. However, in a real life situation, parameters are selected to fit the requirements of the individual system. Refer to the procedure "Entering the Parameters" of descriptions of these parameters.

If the system does not have a card reader, then the procedure "Entering The Parameters" must be utilized "after" SYCOM has been executed. When SYCOM is ready to accept parameters from the SPO it is so indicated by the display of "ENTER SYCOM PARAMETERS". (Refer to "Execution of SYCOM" procedures.)

Setup MCP Program Name Table

In order for SYCOM's Network Controller to be automatically executed by the system, it is necessary to establish the identity of this Controller in the "MCP Program Name Table". To do this is simple, enter the following MCP command:

--> CH C USER/NC(etx) NETWORK CONTROLLER CHANGED FROM "" TC "USER/NC"

The controller name "SYCOM/NC" is of course a user optional name and is therefore only representative. The actual Controller name selected by the user would be substituted as required.

12 Sep 1978

Software Operational Guide

Page = 112

Section IV Operating Procedures

(Original Version)

To verify that the correct Network Controller I/D is in the MCP Name Table ("C" slot), the following MCP command can be entered:

--> WWC(etx) C = "USER/NC"

Assigning Program Priorities

For optimum system response, it is necessary that the system's Network Controller and SYCOM run at a "high" priority. Therefore, to ensure that the MCP automatically executes these programs at their optimum priorities, the following MCP commands are used to permanently assign a priority of "15" to the Network Controller and a priority of "12" to SYCOM:

--> MO USER/NC PR 15(etx) "USER/NC" MODIFIED

--> MO SYCON PR 12(etx) "SYCOM" MODIFIED

EXECUTION OF SYCOM \* WITHOUT SMCS

To use the SYCOM system without the benefit of a SMCS, the system operator simply enters the execute SYCOM command as shown below. The MCP automatically executes the associated Network Controller program (providing one is in the "C" slot of the MCP Name Table) when the first remote file open is initiated for the controller. Both SYCOM and the Network Controller should be run at a priority which ensure adequate response times to the user of the system (see suggested priority values below).

It is the option of the user whether or not SYCOM is executed with a "privileged usercode". If accessing of usercode files is required, then it must be executed "with" a privileged usercode; otherwise execute SYCOM as shown below:

--> EX SYCOM(etx)

SYCOM =< job #> BOJ. PP=12, MP=12 TIME = 08:14:51.2

12 Sep 1978

Software Operational Guide

Page = 113

Section IV Operating Procedures

(Original Version)

During the beginning of the initialization chase if SYCOM cannot locate the "SYCOM/PARAMETERS" file on disk, it attempts to obtain new parameters from the card reader ("CARDS" file). Should the card reader be "not ready" or if there is no card reader, then SYCOM will lapse into the mode of accepting new parameters from the SPO. When SYCOM is in this mood, the following messages are displayed on the SPO:

> % SYCOM =<job #> ENTER SYCOM PARAMETERS SYCOM =<job #> ACCEPT.

The system operator can then enter the appropriate parameters to alter the system's "default parameters" (refer to Appendix G, "Parameters File Attributes"). The entry procedure is presented with "Entering The Parameters" in this section. Although the "SYSTEMID" parameter is not required, in some instances, it is strongly suggested that a representative "system identifier" be entered with the parameters at this time.

Once this file is established on disk, it reed not be recreated unless it is lost or requires changes. Changes to the file can made using either of the two methods identified. However, again, it is necessary to "remove" the existing parameters file from disk. Additionally, this requires SYCOM to be shutdown and remexecuted if it is currently running; else all is in order.

When SYCOM opens its Remote Files, the MCP knows whether or not a Network Controller is running and takes the appropriate action. That is, if a Cortroller is running, then the remote file opens take place and the system becomes operational. If a Controller is "not" currently running, then if the Controller's name is in the "C" slot of the MCP Program Name Table, the MCP automatically executes the named Controller program and the following messages are displayed at the SPO:

> EX SYCOM/NC PT; SYCOM/NC =<job #> BOJ. PP=15, MP=15 TIME = 08:15:08.7

If there is no Controller name entered in the MCP Program Name Table, then the following message is displayed:

REMOTE FILE "MCSRENOTE" NEEDS A CONTROLLER

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 114

Section IV Operating Procedures (Original Version)

The system operator must then execute the appropriate Network Controller by entering the following command:

--> EX SYCOM/NC(etx)

SYCOM/NC =<job #> BOJ. PP=15, MP=15 TIME = 08:16:01.2

The MCP displays the above "BOJ" message, permits the Remote File opens to occur and the system is operational.

Further on during the initialization phase, if Program Switch 2 is set to a value of "2", the following message is displayed on the system SPO:

% SYCOM =<job #> COMPILE DATE = FRIDAY 24 JUL 78 AT 12:07

If Program Switch 3 is set to a value of "3", then the following message is also displayed in conjunction with the above message:

RUN DAY = SATURDAY 19 AUG 78 at 10:13

Upon completion of SYCOM's initialization phase and a successful connection established to a default Server System, the following messages will be displayed on the User SPO:

% SYCON =<job #> \*\*\*\* SYCON (07/24/78 12:08) \*\*\*\* % SYCON =<job #> SYCOM NOW ONLINE TO REMOTE HOST = SERVER1

EXECUTION OF SYCON \* WITH SNCS

When SYCOM runs in conjunction with the SMCS, it is automatically executed by SMCS thru its "job spawning" facility. This occurs when the "first" terminal user "SIGNS-DN" or "PASS's" to SYCOM from a local terminal. Therefore, the system operator must not execute SYCOM from the system SPD; else SMCS just zip executes another copy of SYCOM which causes a few operational problems.

12 Sep 1978

Software Operational Guide

Page - 115

Section IV Operating Procedures

(Original Version)

If during the initialization phase, SYCOM cannot locate the file "SYCOM/PARAMETERS", it attempts to obtain new parameters from the card reader ("CARDS" file). Should this card file not be present or if there is no card reader, then SYCOM commences to accept new parameters input from the SPO. To signal the system operator of this requirement, the following messages are displayed:

> X SYCOM =<job #> ENTER SYCOM PARAMETERS SYCOM =<job #> ACCEPT.

The system operator must then enter the appropriate parameters to override the system's "default parameters" (refer to Appendix G, "Parameters File Attributes"). The parameter entry procedure is presented with "Entering The Parameters"; this section. Although the parameter "SYSTEMID" is not required, for some instances, it is strongly suggested that a representative "system identifier" be entered with the parameters at this time.

Also during the initialization phase, if Program Switch 2 is set to the value "2", than the following SPO message is displayed:

% SYCOM =<job #> COMPILE DATE = FRIDAY 24 JUL 78 AT 12:07

If Program Switch 3 is set to a value of "3", then the following message is also displayed in conjunction with the above message:

RUN DAY = SATURDAY 19 AUG 78 at 10:13

Upon completion of initialization and the successful connection to a default Server System, the following messages are displayed on the User SPO:

> % SYCOM =<job #> \*\*\*\* SYCOM (07/24/78 12:08) \*\*\*\* % SYCOM =<job #> SYCOM NOW ONLINE TO REMOTE HOST = SERVER1

12 Sep 1978

Software Operational Guide

Page - 116

Section IV Operating Procedures

(Original Version)

COMMON SYSTEM OPERATIONS

Once SYCOM has been executed, from either the SPO or spawned by the SMCS, most operating procedures are common to both modes of operation. These common operating procedures are oriented to SPO input in the form of SYCOM parameters, SYCOM commands, and MCP directives. Following then are the common operating procedures, presented in the sequence they are most likely to occur; under individual headings.

SHCS Signal Characters

In order for a SYCOM systems network to function effectively, it is a requirement that the SMCS "signal character" for each system in the network to be a different character. This can be easily accomplished by mutual agreement when the SYCOM system network is designed. Each system's SMCS should have a specific character which is made known to all system users. Obviously, if the SMCS is not utilized, this is not necessary.

If during actual system operations it is found that a SMCS signal character is undesirable for any reason, it can be changed while online. This can be accomplished by entering a "signal character change command" from either a local or remote source. For actual details, refer to the "SMCS Software Operational Guide".

Entering the Parameters

The parameter definitions listed below can be entered to SYCOM from the system SPO. To do so, they must be entered with one line using the following "Accept" process:

--> <sycom job #> AX <parameter keyword> [=<value>] [<parameter keyword> [=<value>]....] (etx)

Each parameter is validated and the system operator is informed of any illegal parameters encountered during the parsing of the parameter data.

12 Sep 1978

Software Operational Guide

Page - 117

Section IV Operating Procedures

(Original Version)

DEBUG parameter

This parameter indicates that the printer tracing mechanism is to be utilized to produce a record of SYCDM's I/D message activity.

Syntax: DEBUG

The default for this parameter is no "printer trace".

BUFFER parameter

This parameter defines the size, in bytes, of the used in SYCOM communications with the other system. The buffer size should be large enough to handle the largest message, and possibly achieve a blocking factor for multiples of a record.

Syntax: BUFFER = <integer>

The default buffer size is 400 and the maximum allowable size is 1950 bytes. (This parameter directly relates to file transfer buffer size and not RJE buffer size.)

RECORD parameter

This parameter defines the maximum record size. This value limits the record size of files, and the length of program-to-program messages transmitted by SYCOM to another system. Thus, if SYCOM is to be used to transfer code files only, the record size could be set to 180 bytes.

Syntax: RECORD = <integer>

The default record size is set to the value of 180. The maximum record size permitted is 1950 bytes.

B1800/B1700 SYSTEMS COMMUNICATIO	INS MODULE 12 Sep 1978
Software Operational Guide	Page - 118
Section IV Operating Procedures	(Original Version)
	· · · · · · · · · · · · · · · · · · ·

SYSTEMID parameter

This parameter is used to specify the name of the User System. A system-id can be comprised of from one to seventeen characters.

Syntax: SYSTEMID = <system=id> (e.g., USER1)

PROGRAM parameter

This parameter is specified when SYCOM is to perform program-to-SYCOM functions. The "record size" parameter limits the maximum length of these messages.

Syntax: PRCGRAM

This parameter defaults to "no" program-to-SYCON input permitted.

TIMEOUTS parameter

This parameter is used to establish an "interval" for the display of "NUMBER OF TIMEOUTS = 25" error messages on the system SPO. A value of "O" inhibits the message display entirely. Other values cause the message to be displayed when the timeout count reaches the specified value.

Syntax: TIMEOUTS = <number>

WAIT parameter

This parameter is used to prevent SYCOM from going to "end-of-job when it receives a "disconnect message". SYCOM will disconnect and lapse into a "offline/idle" state.

Syntax: WAIT

12 Sep 1978

Software Operational Guide

Page - 119

Section IV Operating Procedures

(Original Version)

NOFS parameter

SYCOM will not send a "FS1" or "FS2" (file separator) to the other system when sending card images (relevant to 7.0 RJE).

Syntax: NOFS

NOFIP parameter

This parameter inhibits File Transfers.

Syntax: NOFTP

ETX parameter

Entering an ETX in the parameter string terminates the parameter input process.

Syntax: <sycom job #> AX <parameter string> (etx)

SYCOM Commands from The SPO

The system operator is permitted to communicate with SYCOM using the MCP "AX" facility. By entering certain commands (examples below), the operator is also permitted to communicate with application programs in a Server System; thru the "virtual terminal" facility of SYCOM. A typical command session might go as follows; SPO input is flagged with "-->" and indented lines are output reply messages:

--> <sycom job #> AX.WHAT(etx)

% SYCON =<job #> SYCOM ONLINE; IDLE % SYCOM =<job #> THIS STATION OFFLINE, LOCAL HOST = USER1 % REMOTE HOST = SERVER1

--> <sycom job #> AX.PUT DMPALL(etx)

% SYCOM =<job #> FILE DMPALL START TFANSMISSION % SYCOM =<job #> FILE DMPALL END TRANSMISSION B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 120 Section IV Operating Procedures (Original Version) --> <sycom job #> AX\_CONNECT(etx) Z SYCOM =<job #> ".CONNECT" INITIATED --> <sycom job #> AXEX DMPALL;FI OUTP.FILE INV 2(etx) % SYCOM =<job #> DMPALL =<job #> BGJ. PP=9, MP=9 TIME =... % SYCOM =<job #> % DMPALL =<job #> ENTER SPECS % SYCON =<job #> DMPALL =<job #> ACCEPT. --> <sycom job #> AX<dmpall job #>AXCOPY DSKLST SMCS/TRACE 120 9(etx) X SYCOM =<job #> DNPALL =<job #> "OUTPUT" = "BACKUP.PRT/11 % SYCON =<job #> DMPALL =<job #> "BACKUP.PRT/11" RELEASED X SYCOM =<job #> X DMPALL =<job #> ENTER SPECS % SYCON =<job #> DMPALL =<job #> ACCEPT. --> <sycom job #> AX(etx) % SYCON =<job #> DMPALL =<job #> E0J. TIME = 12:35:33.6 --> <sycom job #> AX.DISCONNECT(etx) % SYCOM =<job #> STATION DISCONNECTED --> <sycom lob #> AX.FETCH BACKUP.PRT/11(etx) % SYCON = < job #> FILE BACKUP.PRT/11 START TRANSMISSION X SYCOM = < job #> FILE BACKUP.PRT/11 RECEIVED X SYCOM =<job #> FILE BACKUP.PRT/11 END TRANSHISSION

### Systematic SYCOM Shutdown

When SYCON is running as a self-sustaining entity (without the SMCS), the ".EOJ" command provides the means by which an orderly shutdown can occur. Once this command has been entered, all subsequent commands entered are rejected. An appropriate "close message" is sent to the other SYCOM after the completion of all in-process file transfers. SYCOM then sends a "disconnect message" to the Network Controller and goes to "end-of-job". To initiate an orderly shutdown of SYCOM, the system operator must enter the following:

--> <sycom job #> AX.EOJ(etx)

12 Sep 1978

Software Operational Guide

Page - 121

Section IV Operating Procedures

(Original Version)

The following message is displayed immediately on the User SPO.

% SYCOM =<job #> ECJ INITIATED

No further processing requests are accepted. Any file transfers queued to process are allowed to complete. At the point in time when there is no longer any SYCOM activity, SYCOM goes to "EDJ". The standard MCP "EOJ message" is displayed for this action:

SYCOM = < job #> E0J. TIME = 16:21:55.1

When SYCOM is running in a self-sustaining environment and it is necessary to shutdown the system "immediately", then the ".STOP" command provides this means for termination. Upon entry of this command, all in-process activity is aborted and all scheduled events are purged. A "close message" is then sent to the other SYCOM, a "disconnect message" sent to the Network Controller, and the program goes to "end-of-job". The system operator enters the following to initiate the immediate shutdown of SYCOM:

--> <sycom job #> AX.STOP(etx)

The following messages are then displayed on the system SPD:

% SYCOM = <job #> STOP INITIATED SYCOM = <job #> E0J. TIME = 16:21:55.1

When SYCOM is running in conjunction with the SMCS, the "STOP" command entered to the SMCS in turn causes SYCOM to be terminated automatically. When this occurs, the following message will be displayed on the SPO:

SYCOM =<job #> DEATH IN FAMILY. TIME 12:58:01.1

A note of importance: If there are any requests queued for processing, they are lost when SYCOM shuts down under these circumstances.

12 Sep 1978

Software Operational Guide

Page - 122

Section IV Operating Procedures

(Original Version)

Reactionary Hints and Pointers

During the course of SYCOM operations, there are a few unusual system "conditions" which could occur. In order for the system operator to correctly respond to these conditions, there must be sufficient information available. That information is provided here in the form of problem descriptions, "hints", "pointers", and responses to these conditions.

A. If at any time SYCOM appears to be "hung-up", a "<job #> WY" command must be entered at the system SPO. If the following message is displayed:

SYCOM =<job #> WAITING MAGNETIC TAPE <tape label>

SYCOM has been automatically stopped waiting for the required magnetic tape to become available; at which time SYCOM operations will automatically resume.

This condition is resolved by the operator making a scratch magnetic tape available to the system. brought down with a MCP "QC" command. (A system includes the SMCS, Network Controller, CANDE, and all programs associated with the Network Controller.)

If the following message is displayed:

SYCOM =< job #> EXECUTING

and SYCOM has just been executed, then the problem lies in the fact that SYCOM is attempting to establish a line connection to the remote system. It will continue this cycle until the connection is made or the program is DS'd. No input commands will be entertained during this initial establishment cycle.

B. If the following message is displayed (by the MCP) on the system SPO during the initialization phase:

INSUFFICIENT DISK WHILE OPENING QUEUE-FILE "SYCOM"

It means that there is not sufficient space on disk to open the "SYCOM/LINK.Q" queue. Affirmative action by the system operator would be to make available a sufficient amount of disk space (approximately 2000 segments).

12 Sep 1978

#### Software Operational Guide

Page - 123

Section IV Operating Procedures

(Original Version)

TERMINAL PROCEDURES

In this sub-section, basic terminal operating procedures are provided in the form of examples. For a SYCOM oriented system however, terminals may be owned by either SYCOM or the SMCS. Because of these possibilities and the associated implications, specific operating procedures are included for both "SYCOM controlled" and "SMCS controlled" terminals.

### TERMINAL FUNCTIONS

Terminal functions are those software features found in a TD820, and/or TD830 type Video Display Terminal (VDT) devices. These terminal features would included such things as; tab stops, lower/upper case, scrolling, etc.. These functions are fully described in the Supervisory Message Control System (SMCS) Software Operational Guide (see documentation reference).

## TERMINAL OPERATIONS (SYCOM CONTROLLED)

When a local terminal is under the direct control of SYCOM, it is also permanently "attached" to it. Therefore, a terminal user has immediate access to SYCOM functions. A terminal can however, be temporarily "attached" to either SYCOM or the SMCS in a Server System. Thru Server System's SMCS, the terminal can also be "attached" to an appropriate interactive application program in the Server System. The terminal user is not limited to just a few SMCS commands, a user may invoke any command which the SMCS has in its repertoire; with the exception of those restricted to input from the SPC.

## Example Interactive Terminal Session

Terminals at this point are attached to the User System's SYCOM. In order to utilize the SYCOM capabilities, a terminal user need only enter the cesired SYCOM commands. Among these SYCOM commands are ".CONNECT" and ".DISCONNECT". These are used to gain access to any appropriate interactive application program in the Server System. The examples which follow explore all possible terminal connections as cetailed in "Message Path Assignments". B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page = 124 Section IV Operating Procedures (Original Version) In the examples below, terminal input is identified by the flag "-->". All other lines are reply messages from the program to which the terminal is currently inputting to. First some basic local SYCOM commands are shown. All SYCOM commands are entered directly as exampled below: --> .WHAT (User SYCOM) SYCOM ONLINE; IDLE THIS STATION OFFLINE, LOCAL HOST = USERI REMOTE HOST = SERVER1 --> .PUT DMPALL (User SYCOM) FILE CMPALL START TRANSMISSION --> WHAT (User SYCOM) SYCOM ONLINE; 00 FILES QUEUED; SENDING DMPALL(47/85) THIS STATION OFFLINE; LOCAL HOST = USER1, REMCTE HOST = SERVER1 FILE DMPALL END TRANSMISSION -CONNECT (User SYCOM) --> ".CONNECT" INITIATED (User SYCOM) --> .WHAT SYCOM ONLINE; IDLE THIS STATION UNLINE, LOCAL HOST = USER1, REMOTE HOST = SERVER1 (Server SMCS) JOBS --> ACTIVE: SERVER/NC =< job #> PP=15, MP=15 SMCS =<job #> PP=13, MP=14 SERVER/SYCOM =< iob #> PP=12, NP=12 SCHEDULED: BOTH SCHEDULES EMPTY. MIX LIMIT=16 It is necessary that the terminal user be aware of what "signal

character" the Server System's SMCS is currently using. To find out, the following command is entered to query that character:

--> SIGNAL

(Server SMCS)

SIGNAL CHARACTER= \$

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 125 Section IV Operating Procedures (Original Version)

Now the terminal user can "sign on" to an application program in the Server System; and alternately enter commands to the Server System's SMCS. A series of following interactions exemplify this concept: (The application program "SET/TABS" used in the example is assumed to "set" specified tab stops in the user's terminal.)

--> SIGN CN SET/TABS

(Server SMCS)

SIGNED ON TO "SET/TABS", SIGNAL="\$"

(Server SMCS)

--> \$J085

ACTIVE: SERVER/NC =<job #> PP=15, MP=15 SERVER/SYCOM =<job #> PP=12, MP=12 SET/TABS =<job #> PP=4, MP=4 SMCS =<job #> PP=13, MP=13 SCHEDULED: BOTH SCHEDULES EMPTY. MIX LIMIT=16

--> 10,20,30,40,50,60,70

(Server AP)

(Server SMCS)

(tab stops are now set) REMCTE FILE CLOSED BY 'SET/TABS', STATION SIGNED OFF

--> EX DMPALL;FI OUTP.FILE INV 2

DMPALL =<job #> BOJ. PP=9, MP=9 TIME = 11:35:21.8 % DMPALL =<job #> ENTER SPECS DMPALL =<iob #> ACCEPT.

--> <dmpall job #> AX COPY DSKLST SHCS/TRACE 120 9

% DMPALL =<job #> ENTER SPECS DMPALL =<job #> ACCEPT.

--> <dmpall job #> AX

DMPALL =< job #> EDJ. TIME = 11:36:58.1

At the point in time when a user is finished entering commands to SYCOM and wishes to terminate the session, the following command is entered; however, the terminal is still "attached" to SYCOM.

--> .DISCENNECT

(User SYCOM)

STATICN DISCONNECTED

12 Sep 1978

Page - 126

Software Operational Guide

Section IV Operating Procedures

### (Original Version)

### TERMINAL OPERATIONS (SMCS CONTROLLED)

When a terminal is permanently attached to, and owned by, the SMCS, the terminal user must gain access to SYCOM functions thru the use of explicit SMCS commands. These commands are exampled below. By gaining access to SYCOM, the SMCS will temporarily "attach" the requesting terminal to SYCOM for the duration of a session. However, a terminal user is not limited to just a few SMCS commands, a user may invoke any command which the SMCS has in its repertoire; with the exception of these restricted to input from the SPD.

Example Interactive Terminal Session

The terminal at this point is attached to the User System's SMCS. In order to gain access to those nifty SYCOM capabilities, the terminal user must make use of the SMCS "SIGN ON" and "SIGN OFF" commands. The sample session which follows, shows the terminal user just how easy it is to "get on" and "get off" SYCOM: (input from the terminal is identified by the flag "-->".)

--> SIGN ON SYCOM

(User SMCS)

SIGNED ON TO "SYCOM", SIGNAL= \*\*\*

When the "signed-on" message is returned to the terminal, that terminal is then attached to SYCOM. Therefore, SYCOM commands are entered directly as exampled below:

--> .WHAT

(User SYCON)

SYCOM ONLINE; IDLE THIS STATION OFFLINE, LOCAL HOST = USER1 REMOTE HOST = SERVER1

--> .PUT DNPALL

FILE DMPALL START TRANSMISSION

--> . WHAT

(User SYCOM)

(User SYCOM)

SYCCM ONLINE; OO FILES QUEUED; SENDING DMPALL(47/85) THIS STATION OFFLINE; LOCAL HOST = USER1, REMOTE HOST = SERVER1 FILE DMPALL END TRANSMISSION BIROO/BITOO SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page = 127 Section IV Operating Procedures (Original Version) (User SYCOM)

--> .CONNECT

".CONNECT" INITIATED

- WHAT -->

(User SYCOM)

SYCCM ONLINE; IDLE THIS STATION CNLINE, LOCAL HOST = USER1. REMOTE HOST = SERVER1

\*REPORT > -->

(User SMCS)

>D1 (SPU) : SIGNED ON TO 'SYCOM' 1 STATION REPORTED AT 11:42.6 CE/22/78

--> JOBS (Server SMCS)

ACTIVE: SERVER/NC =<job #> PP=15, MP=15 SMCS =<iob #> PP=13, MP=14 SERVER/SYCOM =<iob #> PP=12, MP=12 SCHEDULED: BOTH SCHEDULES EMPTY. MIX LIMIT=16

It is necessary that the SMCS in a Server System have a "signal character" that is different than that of the SMCS in the User System. To ensure this, the following command is entered to query that signal character: (If it "is not" different, then change it; "SIGNAL \$".)

--> SIGNAL (Server SMCS)

SIGNAL CHARACTER= \* \$ \*

Now the terminal user can "sign on" to an application program in the Server System; and alternately enter commands to the Server System's SMCS. A series of following interactions exemplify this concept: (The application program "SET/TABS" used in the example is assumed to "set" specified tab stops in the user's terminal.)

SIGN ON SET/TABS -->

(Server SMCS)

SIGNED ON TO 'SET/TABS', SIGNAL='\$'

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page - 128
Section IV Operating Procedures (Origi	inal Version)
> \$JOBS	(Server SMCS)
ACTIVE: Server/NC = <job #=""> PP=15, MP=15</job>	
SERVER/SYCOM = <job #=""> PP=12, MP=12 SET/TABS =<job #=""> PP=4, MP=4</job></job>	
SMCS = <job #=""> PP=13, MP=13 SCHEDULED:</job>	
BOTH SCHEDULES EMPTY. MIX LIMIT=16	
> *REPORT >	(User SMCS)
D1 (SPD) # SIGNED ON TO "SYCOM" 1 STATION REPORTED AT 11:44.2 08/22/78	n an an Star an Anna An Star an Anna An Star an Anna Anna
> 10,20,30,40,50,60,70	(Server AP)
(tab stops are now set) REMCTE FILE CLOSED BY 'SET/TABS', STATIO	N SIGNED OFF
> .DEBUG	(User SYCOM)

DEBUG OFF

--> .DISCONNECT

STATION DISCONNECTED

At the point in time when a User is finished entering commands to SYCOM and wishes to terminate the session, the following command is entered: (This is a SMCS command.)

--> \*SIGN OFF

(User SMCS)

(User SYCOM

The response from the SMCS will be as follows:

DETACHED FROM \*SYCOM\*

At this point, the terminal has been "detached" from SYCOM and is now "re-attached" to the SMCS.

SECTION V

GENERATING THE NETWORK CONTROLLERS

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 130 Section V Generating the Network Controllers (Original Version)

SECTION V

GENERATING THE NETWORK CONTROLLERS

# GENERAL

This section is provided as a user oriented guideline to the generation of a Network Controller for the SYCOM system. The controller is generated by the Network Definition Language (NDL) compiler using a prepared set of NDL specifications. The system user is required to determine these NDL specifications according to the system configuration and environment under which the Network Controller is to function.

The following guidelines are offered for a target system that is to provide the full range of SYCOM services; including terminals. They are intended to aid the user in generating a Network Controller with correct functional attributes. For a complete description of the NDL specifications and definitions, refer to the "Network Definition Language reference manual" (associated documentation).

# DETERMINING THE SPECIFICATIONS

The following sample Network Controller NGL specifications are presented in a sequence similar to that of the NDL reference manual. This allows the user a logical reference for evaluating the sample NDL specifications for application to the installation requirements. They identify suggested parameters which produce an operational Network Controller program for the SYCOM system configuration as specified.

12 Sep 1978

Software Operational Guide

Page - 131

Section V Generating the Network Controllers (Original Version)

OPERATIONAL PROCEDURES

The following procedures provide the methods by which the user prepares the input to the NDL compilation process, generates a Network Controller, and induces modifications into the resident NDL compiler, if required.

Preparation of Input

The NDL "source" input file can be prepared in the form of either a card file or a NDL type disk file. The cisk file could be created and maintained using CANDE or another text editor of similar capabilities.

The "Sample NDL Parameters File" is furnished for the purpose of creating this file with the most amount of accuracy in the least amount of time. Any adjustments to this scurce file for number of lines, terminals, and other such attributes is the responsibility of the user which is attuned to the system configuration and its operational environment.

Compilation Process

Prior to initiating the NDL compilation process, the user should ensure the presence of a "7.0 NDL/LIBRARY" file on disk, of a 07/24/78 or later vintage). Then, there are several methods by which the NDL compilation (generative) process can be invoked. Following are two of these methods:

1. Card Input:

Following is a primitive example of a NDL source card input deck: (The <I> in the example below represents an "invalid" character as required by the MCP.)

<I> CO USER/NC NCL LI <I> DATA CARDS

NDL source images NDL library calls

<I> END

B1800/B1700 SYSTEMS	COMMUNICATIONS	MODULE	12 Sep 1978
Software Operational	Guide		Page - 132

Section V Generating the Network Controllers (Original Version)

2. Disk Input:

This method is used to initiate a NDL compiler process when the source input file resides on disk. Enter the following MCP command using the system SPO:

--> CO USER/NC NDL LIFFI CARDS NAM USERNC/SOURCE DSK DEF

From this point on, the compilation procedures are the same. When the MCP executes the NDL compiler, a normal "BOJ" message is displayed. Upon successful completion of the compile, the normal "EOJ" message is displayed. However, if an error message such as the following is displayed:

#### NDL : USER/NC =<job #> --EVALUATION/PROGRAM POINTER STACK OVERFLOW DS OR DP

Then the procedure "NDL Compiler Modifications" must be utilized to fix the problem.

NDL Compiler Modifications

Should the NDL compilation fail because of a lack of sufficient internal work space, the following procedure can be invoked to expand the NDL compiler itself to overcome the problem:

--> EX SDL/SSC(etx) SDL/SSC =<job #> BOJ. PR=4 TIME = 07:28:48.1 X SDL/SSC =<job #> SDL/SSC, MARK VII.0 OF 10/21/77 21:19 X SDL/SSC =<job #> X SDL/SSC =<job #> ENTER FILENAME OF BLANK TO TERMINATE SDL/SSC =<job #> ACCEPT.

--> <sdl/ssc job #>AXNDL(etx)

% SDL/SSC =<job #> \*\*\*\*\* FILE NAME : NDL % SDL/SSC =<job #> ENTER SCRATCHPAD NUMBER SDL/SSC =<job #> ACCEPT.

--> <sdl/ssc job #>AXO(etx)
12 Sep 1978

#### Software Operational Guide

Page - 133

Section V Generating the Network Controllers (Original Version)

% SDL/SSC =<job #> % SDL/SSC =<job #> \*\*\*\*\* SCRATCHPAD: 0 % SDL/SSC =<job #> VALUE : 27000 % SDL/SSC =<job #> NAME : 200 % SDL/SSC =<job #> DISPLAY : 192 % SDL/SSC =<job #> CONTROL : 40 % SDL/SSC =<job #> EVALUATION : 40 % SDL/SSC =<job #> PROGRAM POINTER : 50 % SDL/SSC =<job #> DYNAMIC : 20000 % SDL/SSC =<job #> \*\*\*\*\* TOTAL CORE: 63112 % SDL/SSC =<job #> ENTER STACK ID SDL/SSC =<job #> ACCEPT.

The value "300" used in the example below is a known acceptable integer; a smaller value may also suffice but would be dependent upon the size and complexity of the Network Controller being generated.

--> <sdl/ssc job #>AXE 300(etx)

% SDL/SSC =<job #> ENTER \* \* TO ENC CHANGES SDL/SSC =<iob #> ACCEPT.

--> <sdl/ssc job #>AX (etx)

(a updated list of program parameters is displayed)

SDL/SSC =<job #> E0J. TIME = 07:29:50.4

If any other error messages are displayed along this vein, they should contain sufficient information to allow the modification of the associated program parameter; in the manner shown above.

#### Print The USER/NCNIF File

Once the compilation/generation of the Network Controller is complete, it is suggested that the "Network Information File" be printed. This is accomplished by executing the following program; and providing the correct file name (if different than the one suggested):

--> EX NIF/ANALYZER; FI NIF NAM USER/NCNIF(etx) NIF/ANALYZER =<job #> B0J. PR=4 TIME = 11:05:27.1 NIF/ANALYZER =<job #> E0J. TIME = 11:07:12.9

12 Sep 1978

Software Operational Guide

Page = 134

Section V Generating the Network Controllers (Original Version)

SAMPLE NDL SOURCE FILE (LCCAL USER SYSTEM)

Following is a sample NDL source file to be used as input to the NDL compiler for the generation of a Network Controller oriented to a User System. This example provides a Network Controller which services a local site of 3 terminals, 2 virtual terminals, and a inter-system connection. The user must tailor this sample Network Controller specifications to the needs and desires of the installation (e.g., more/less terminals, more/less virtual terminal positions, etc).

This sample NDL input file is comprised of both NDL statements and parameter specifications. The resulting Network Controller will provide the correct functional attributes necessary for the operation of the SYCOM system; in testing mode.

\$ CONTROL \$ LIBINFO X number of entries; for IOLOG \$ NSSIZE 20 % number of bits; \$ VSSIZE 150000 for IOLOG 7 % Z 7 Declaration Section Z z DECLARATION: NIF = "USER"/"NCNIF". % NIF file name; user option MAX FILES = 16. MAX BUFFERS = 2. MAX MESSAGES = 20. \$ LIBRARY SYCOMDECL % Z 2 7 **Requests and Controls** Ľ Z \$ LIBRARY SYCONREQ \$ LIBRARY CANDEPOLTD 2 \* \* **\$** LIBRARY CANDEFSLTD %\*\* \$ LIBRARY SYCONCTL \$ LIBRARY CANDETDCTL 2 \* \*

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Page - 135 Software Operational Guide Section V Generating the Network Controllers (Original Version) 2 Z Z Network Configuration Z 7 X 7 Z % Terminal Section Z % Z TERMINAL DEFAULT SYCOMDEFAULT: REQUEST = SYCOMREQ:RECEIVE, SYCOMREQ:TRANSMIT. TRANSMISSION = 1. ADDRESS = 2.X Number of polling address characters BUFFERSIZE = 3840.2 2 x 1920 bytes TYPE = 44. % Required for virtual terminal msg's TERMINAL SYCOMTERNINAL: DEFAULT = SYCCMCEFAULT. TERMINAL DEFAULT TOFLT: X \* \* REQUEST = CANDEPOLTD:RECEIVE, X++ CANDEFSLTD:TRANSMIT.X\*\*  $TRANSMISSION = C_{-}$ 2\*\* ADDRESS = 2. % \* \* TYPE = 44. 2 \* \* BUFFERSIZE = 2250. 2 \* \* TERMINAL TDE24: % \* \* DEFAULT = TDFLT.2 \* \* % z 2 Station Section Z 2 2 STATION DEFAULT SYCOMDEF: FREQUENCY =  $250 \cdot 250$ . MCS Controller capability?? 2 MYUSE = INPUT, CUTPUT. TERMINAL = SYCOMTERMINAL. RETRY = 25. STATION DEFAULT TD8DF: %\*\* FREQUENCY = 249, 249.7 \* \* Non-Remote SPO MYUSE = INPUT, CUTPUT. % \* \* TERMINAL = TD8?4.% \* \* RETRY = 4. 7 \* \*

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 136 Section V Generating the Network Controllers (Original Version) % STATION CONTROLST1: 7 lsn 001 | 2 DEFAULT = SYCONDEF. ADDRESS = "00". \* 2 STATION SPOST2: X lsn 002 1 DEFAULT = SYCCMDEF. 2 ADDRESS = "01". z 1.00 Must be grouped. z STATION CARDST3: 7 lsn 003 > -----------DEFAULT = SYCONDEF. Z % ADDRESS = "02". Don't have to be 2 first stations % STATION PRINTERST4: defined. lsn 004 1 X DEFAULT = SYCONDEF. ADDRESS = "03". 2 x STATION FILESTS: % Lsn 005 1 DEFAULT = SYCOMDEF.7 1 ADDRESS = "04". Z 1 X 2-----Z STATION USERT1: % lsn 006 \* Virtual terminal DEFAULT = SYCOMDEF. z (SPO/terminals) ADDRESS = "Aa", "aA". X 2 STATION USERT2: lsn 007 \* Virtual Terminal z DEFAULT = SYCONDEF. 7 (Program) ADDRESS = "Pa", "aP". Z 2 ----7 2---------X\*\* \_\_\_\_\_ STATION D1: tsn 008 X\*\* 1 DEFAULT = TD8DF. %\*\* ADDRESS = "D1". %\*\* FREQUENCY = 249, 250.2\*\* Remote SPO I % \* \* STATION D2: %★★ lsn 009 1 DEFAULT = TD8DF. > Local terminal %\*\* ADDRESS = "D2". Z\*\* FREQUENCY = 249, 250.X \* \* Remote SPO 1 % \* \* STATION D3: % \* \* lsn 010 DEFAULT = TD8DF. %\*\* ADDRESS = "D3". 7 \* \* FREQUENCY = 249, 250.% \* \* Remote SPO 1 X \* \*

ftware Operational Guide		Page - 13
ction V Generating the Net	ork Controllers (0	riginal Version
Line	Section	
	.===================================	***************
NE DIRECTO1:		
CONTROL = SYCCMCTL.		
STATION = CONTROLST1 · SPOST	12 CARDST3 PRINTERS	ST4, FILEST5,
ADDRESS = 1:0:6.	% (line, adapte	r, port; system
NE DEFAULT L'INCOLLT.	•	
CONTROL = CANDETDCTL.	% = = X * *	and a state of the
ALTOPOLL = 15.	X** 5 x max stati	ions/lines
NE MULTIC4:	Z** adapter 4	
DEFAULT = LINEDFLT.	X**	
$STATION = D1 \cdot D2 \cdot D3 \cdot D3 \cdot D3 \cdot D00 \cdot C5 - 1 \cdot 0 \cdot C \cdot C5 - 1 \cdot 0 \cdot C \cdot C5 - C5 - C5 - C5 - C5 - C5 - C5 $	X** X** (ling, adapt/	s. sost: systa
ADDRESS - 1.0.4.	A== ((inep auapt)	sr ports syste
LE LINK:	Z for SYCOM	
FAMILY = CONTROLST1, SPOST	2. CARDST3. PRINTERS	14, FILESTS,
USER II.	2 rsn's UCL+ UU 2 005, 000	<i>LP</i> UUSP UU4P 5
LLE RFUSER: FAMILY = USERT2.	% For program-to % rsn 001	p-program I/O.
ILE TERMINALS:	X** This file is a X** when SYCCM is	named "TERNINAL
	%** SMCS. It must	t be named
	X** "MC SREMOTE" wi	nen SYCOM is ru
$FAMILY = D1_{\bullet} C2_{\bullet} D3_{\bullet}$	X** rsn's 001. 002	2. 003

FINI.

Source configuration notes:

If the User System does not have "local terminals" then all lines in the above example with comments indicated with "Z\*\*" can be omitted from the source file.

12 Sep 1978

Software Operational Guide

Page - 138

Section V Generating the Network Controllers (Original Version)

SAMPLE NDL SOURCE FILE (REMOTE SERVER SYSTEM)

Following is a sample NDL source file to be used as input to the NDL compiler for the generation of a Network Controller oriented to a Server System. This example provides a Network Controller which services one remote site of no terminals, an inter-system connection, and a SMCS. The user must tailor these sample test Network Controller specifications to the needs and desires of the installation (e.g., local terminals, more/less virtual terminal positions, etc).

This sample NDL input file is comprised of both NDL statements and parameter specifications. The resulting Network Controller will provide the correct functional attributes necessary for the operation of the SYCOM system; in a test mode.

S CONTROL \$ LIBINFO \$ NSSIZE 20 % number of entries; for IOLOG \$ VSSIZE 150000 % number of bits; for IOLOG % % X Declaration Section Z 7 Z DECLARATION: NIF = "SERVER"/"NCNIF". X NIF file name; user option MAX FILES = 16. MAX BUFFERS = 2. MAX MESSAGES = 20. \$ LIBRARY SYCOMOECL 7 2 2 **Requests and Controls** 2 Z % \$ LIBRARY SYCOMREQ **\$ LIBRARY SYCOMCTL** 

B1800/B1700 SYSTEMS CUMMUNICATIONS MODULE 1 Software Operational Guide P Section V Generating the Network Controllers (Origina	2 Sep 1
Software Operational Guide P Section V Generating the Network Controllers (Origina	
Section V Generating the Network Controllers (Origina	age –
	l Versi
X = = = = = = = = = = = = = = = = = = =	========
7. Y Terrinal Section	
7	•
X====================================	
ILEMENAL DEFAULT SYCOMPERATES	
TRANSMISSION = 1.	
ADDRESS = 2. X Number of polling address	charact
BUFFERSIZE = $3840$ . $\%$ 2 x 1920 bytes	
TYPE = 44. Z Required for virtual termi	nal msg
TERMINAL EVERNITERMINAL -	
IERMINAL STUUMIERMINAL: OFFAILT = SYCONOFFAILT	
ULINULI - JIUUNULINULI	a transformer e
X=====================================	=======
$\mathbf{x}$ , but the end of the constant of the product of the constant of the co	
% Station Section	
<b>X</b>	• .
	======
STATION DEFAMILT SWOONDEF.	
FREQUENCY = 250, 250, 2 MCS Controllor canab	il it v 77
NYUSE = INPUT, CUIPUT.	
TERMINAL = SYCOMTERMINAL.	artigan. An an
RETRY = 25.	
N	
<u> </u>	
χ	
STATION CONTROL STI: Z lsn 001 I	
DEFAULT = SYCOMDEF. Z	· · ·
ADUKESS = "00". Z	1.11
7 II	
STATION SPOST2: 7 len 002 1	н. С. 4
STATION SPOST2: X lsn 002 1 DEFAULT = SYCCMDEF. X l	000000
STATION SPOST2:ZIsn 002 1DEFAULT = SYCCMDEF.Z1ADDRESS = "01".Z1	giuupeu
STATION SPOST2:XLsn 002 1DEFAULT = SYCCMDEF.X1ADDRESS = "01".X1X11	grouped
STATION SPOST2:ZLsn 002 IDEFAULT = SYCCMDEF.ZIADDRESS = "01".ZISTATION CARDST3:ZIsn 003 >	91 Uupeu
STATION SPOST2:ZLsn 002 IDEFAULT = SYCCMDEF.ZIADDRESS = "01".ZISTATION CARDST3:ZIDEFAULT = SYCCMDEF.ZI	91 04 peo
STATION SPOST2:ZLsn 002DEFAULT = SYCCMDEF.ZIADDRESS = "01".ZISTATION CARDST3:ZIDEFAULT = SYCCMDEF.ZIADDRESS = "02".ZIDon't haveI	sve to b
STATION SPOST2:       Z       Lsn 002 I         DEFAULT = SYCCMDEF.       Z       I         ADDRESS = "01".       Z       I         STATION CARDST3:       Z       I         DEFAULT = SYCCHDEF.       Z       I         ADDRESS = "02".       Z       I         STATION CARDST3:       Z       Isn 003 >         DEFAULT = SYCCHDEF.       Z       I         ADDRESS = "02".       Z       I         STATION PRINTCRST4.       Z       I	ave to t
STATION SPOST2:ZLsn 002 IDEFAULT = SYCCMDEF.ZIADDRESS = "01".ZISTATION CARDST3:ZIDEFAULT = SYCCMDEF.ZIADDRESS = "02".ZISTATION PRINTERST4:ZIsn 004 IDEFAULT = SYCCMDEF.ZIDON'T have a synthesized and a synthesized	ave to t tations
STATION SPOST2:       Z       Lsn 002 I         DEFAULT = SYCCMDEF.       Z       I         ADDRESS = "01".       Z       I         STATION CARDST3:       Z       I         DEFAULT = SYCCMDEF.       Z       I         ADDRESS = "02".       Z       I         STATION PRINTERST4:       Z       I         DEFAULT = SYCOMDEF.       Z       I         ADDRESS = "02".       Z       I         STATION PRINTERST4:       Z       Isn 004 I         DEFAULT = SYCOMDEF.       Z       I         ADDRESS = "03"       Y       I	yrouped ave to b tations
STATION SPOST2: DEFAULT = SYCCMDEF. ADDRESS = "01".ZLsn 002 I ZSTATION CARDST3: DEFAULT = SYCCMDEF. ADDRESS = "02".ZISTATION PRINTERST4: DEFAULT = SYCOMDEF. ADDRESS = "03".ZISTATION PRINTERST4: DEFAULT = SYCOMDEF. ADDRESS = "03".ZI	ave to b tations
STATION SPOST2: DEFAULT = SYCCMDEF. ADDRESS = "01".XIsn 002 ISTATION CARDST3: DEFAULT = SYCCMDEF. ADDRESS = "02".XISTATION PRINTERST4: DEFAULT = SYCOMDEF. ADDRESS = "03".XIsn 004 ISTATION FILEST5:XIsn 005 I	ave to b tations
STATION SPOST2: DEFAULT = SYCCMDEF. ADDRESS = "01".ZI sn 002 i ZSTATION CARDST3: 	ave to b
STATION SPOST2:       Z       Isn 002         DEFAULT = SYCCMDEF.       Z       I         ADDRESS = "01".       Z       I         STATION CARDST3:       Z       Isn 003         DEFAULT = SYCCHDEF.       Z       I         ADDRESS = "02".       Z       I         STATION PRINTERST4:       Z       I         DEFAULT = SYCOMDEF.       Z       I         ADDRESS = "03".       Z       I         STATION FILEST5:       Z       Isn 005         DEFAULT = SYCOMDEF.       Z       I         ADDRESS = "03".       Z       I	ave to tation

%

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 140 Section V Generating the Network Controllers (Original Version) STATION SERVERT1: X \* Virtual terminal lsn 006 DEFAULT = SYCCMDEF. 2 (SPO/terminals) ADDRESS = "aA", "Aa". Z 2 STATION SERVERT2: % \* Virtual terminal lsn 007 DEFAULT = SYCONDEF. X (Program) Z ADDRESS = "aP", "Pa". Z X 2 % Line Section 2 Z % LINE DIRECTO1: CONTROL = SYCOMCTL. STATION = CONTROLST1, SPOST2, CARDST3, PRINTERST4, FILEST5, SERVERTI, SERVERT2. ADDRESS = 1:0:6. 2 (line, adapter, port; system option) FILE LINK: X for SYCOM FAMILY = CONTROLST1, SPOST2, CARDST3, PRINTERST4, FILEST5. rsn's 001, 002, 003, 004, 005 z FILE RESERVER: Z For program-to-program I/O. FAMILY = SERVERT2.z rsn 001 z Note: There is only 1 device 2 defined here because the sample test program is a 2 X single user program. FILE MCSREMOTE: 2 This file is used for the FAMILY = SERVERT1. 2 virtual terminals assigned 7 to Server System's SMCS. Z rsn 001 FINI. Source configuration notes: \_\_\_\_\_ The Network Controller source file for a Server System "with

local terminals" is configured by appropriately adding the NDL statements identified in the User System example with "Z\*\*" as comments.

SECTION VI

APPENDICES

B1800/B1700 SYSTEM	S COMMUNICATION	IS MODULE		L2 Sep 1978
Software Operation	nal Guide			Page - 142
Section VI Appendi	ices		(Origina	al Version)

DESCRIPTION

During command processing various errors may occur; the response error messages and their meanings are shown below. There are six classes of these error messages:

- . Command Error Messages
- . Command Informational Messages
- System Parameter Error Messages
- System Error Messages
- System Condition Messages
- System Fatal Messages

The following error message groups describe all messages that may be displayed in each of the above identified message classes:

12 Sep 1978

Software Operational Guide

Page - 143

Section VI Appendices

(Original Version)

Command Error Messages

The following error messages are displayed as the result of the improper use of SYCOM commands. Generally, an error can occur for one of two reasons; they are:

- 1. Invalid command syntax
- 2. Invalid parameter data

Although the reason for each error message is recognizable, there has been no attempt to group the messages into one of the two categories. The following command error messages are however, grouped alphabetically, by command, to make locating easier for the user.

COMMAND MESSAGE REASON

UNRECOGNIZED COMMAND <x ... x>

This simply means that the last SYCOM command entered was not a valid command.

• A BORT :

THERE IS NO 'PUT' IN PROGRESS TO ABORT

An ABERT command was entered with a specified option of of "PUT" and no file is being transmitted at this time.

THERE IS NO "FETCH! IN PROGRESS TO ABORT

An ABORT command was entered with a specified option of of "FETCH" and no file is being received at this time.

UNRECOGNIZED OBJECT OF "ABORT" <x ... x>

The command was entered with an option keyword which was neither "PUT" or "FETCH". Try again...

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE12 Sep 1978Software Operational GuidePage - 144

(Original Version)

Section VI Appendices

Command Error Messages (continued)

COMMAND MESSAGE REASON

### .CL:

UNRECOGNIZED COMMAND CL XX>

The .CL command was entered either without an option or the option was found to be invalid.

# -CONNECT:

"CONNECT FAILED": VIRTUAL TERMINAL TABLE FULL

This message displayed indicates that all allocated "virtual terminal" positions are currently in use. The terminal user must try again at a later time.

### STATION ALREADY CONNECTED

This only means that a ".CONNECT" command was entered from a terminal (or SPO) that was already "connected" to a Server System; no problem.

### -COPY:

INVALID USERCODE SYNTAX

The "=" equal sign delimiter is missing between the keyword "USER" and the assumed usercode character string.

CAN'T SPECIFY USERCODE TWICE

The "usercode" entered is already logged-on to SYCOM, therefore, a second user with the same identifier is not permitted.

81	800/81700	SYSTEMS	COMMUNICATIONS	MODULE	12 :	Sep	1978

Software Operational Guide

Section VI Appendices

(Original Version)

Page - 145

Command Error Messages (continued)

COMMAND MESSAGE REASON

•COPY (continued):

INVALID USERCODE NAME

The "usercode" entered was rejected because it contains non-alphabetic characters.

USERCODE MAY NOT HAVE MORE THAN 17 CHARACTERS

The length of the "usercode" entered exceeded a maximum allowable length of 17 characters.

#### INVALID PASSWORD SYNTAX

The "=" equal sign delimiter is missing from between the keyword "PASSWORD" and the assumed password character string.

#### CAN'T SPECIFY PASSWORD TWICE

The "password" entered has already been specified by another user; it was rejected for this reason.

#### INVALID PASSWORD NAME

The "password" entered was rejected because it contains non-alphabetic characters.

#### PASSWORD MAY NOT HAVE MORE THAN 17 CHARACTERS

The length of the "password" entered exceeded a maximum allowable length of 17 characters.

#### INVALID FILE ATTRIBUTE: <token>

The command syntax expected a keyword of "US", "USER", or "PASSWORD" which was not located. However, what was entered is displayed in the message as "token".

Page - 146

Software Operational Guide

Section VI Appendices

(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

.COPY (continued):

INVALID FILE ATTRIBUTE DELIMITER: <token>

To be defined.

#### INVALID TERMINATION OF FILE ATTRIBUTE LIST: <token>

The command syntax calls for the "usercode-password" to be delimited by a ")" close paren which was not located in the command string. The character (or characters) that was found is displayed in the error message.

#### FILE NAME MISSING

The command syntax requires a "file name" specified that is the subject of the file transfer; it was not given or was not recognized.

KIND DEFINED TWICE

The "KIND" option keyword has been previously specified in the command string.

MISSING "="

The command syntax requires a "=" equal sign delimiter between the keyword "KIND" and the kind option keyword; which as missing.

#### INVALID KIND ATTRIBUTE

The kind option keyword was not "PACK", "PETAPE", "TAPE", "TAPE7", or "TAPE9" which makes what was entered invalid.

Software Operational Guide

12 Sep 1978

Page - 147

Section VI Appendices

(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

.CCPY (continued): ========

HOST ATTRIBUTE DEFINED TWICE

To be defined.

#### MISSING "=" AFTER HOST ATTRIBUTE

The required "=" equal sign delimiter used between the keyword "HDST" and the host "system name" is missing.

INVALID HOST NAME

The host system name entered was rejected because it contained non-alphabetic characters.

HOST NAME HAS MORE THAN 17 CHARACTERS

When a host system name is entered with more than 17 characters (maximum allowable), it will be rejected.

#### UNRECOGNIZABLE VOLUME ATTRIBUTE

This message is somewhat misleading in that it is only an indication that a keyword of "KIND" or "HOST" was expected and not found.

#### MISSING "," BETWEEN VOLUME ATTRIBUTES

An attribute separator character of "," is required by the command syntax to indicate the end of the attribute "KIND" and its associated option keyword. It is not there and it is wanted.

12 Sep 1978

Page - 148

Software Operational Guide

Section VI Appendices

(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

.COPY (continued):

NO VOLUME ATTRIBUTES DEFINED

Neither the "KIND" or the "HOST" keyword attributes were specified in the command string and "HOST" is required as a minimum.

VOLUME HAS MORE THAN 17 CHARACTERS

The "volume name" (or pack-id) entered exceeded the maximum allowable length of 17 characters. This message could relate to either "volume-1" or "volume-2" names.

DESTINATION HOST NOT ONLINE

The Destination Host Name specified in the COPY command string is not a "local" or "remcte" system identifier. Therefore, the request cannot be accepted as is.

INVALID DESTINATION HOST NAME

The "destination" system name specified is not the same as the "local" system name. Therefore, because the name is not known to SYCOM, it cannot process the request.

SOURCE HOST NOT DEFINED OR REMOTE HOST NOT ONLINE

This message can be displayed for any one of several reasons; they are: 1) the system identifier specified as "source host" is neither the "local" system or the current "remote" system, 2) the required syntactical keyword "FROM" is missing, 3) SYCOM is not connected to the source host specified, or 4) required parens are is missing.

12 Sep 1978

Software Operational Guide

Page - 149

(Original Version)

Section VI Appendices

Command Error Messages (continued)

COMMAND MESSAGE REASON

.CCPY (continued):

INVALID SOURCE HOST NAME.

The system identifier specified as a "source host" does not match the "local" system name, the "remote" system name, and is not blanks. The reduest is not processed.

CANNOT FETCH WHEN REMOTE SYCOM OFFLINE

This ressage is indication that an inter-system line connection does not presently exist and the "FETCH" request cannot be accommodated.

SOURCE AND HOST NAMES MAY NOT BE DEFINED IN A FETCH

Use of the shorthand version (FETCH) of the COPY command implies the source and destination systems, therefore, they cannot be specified.

FILE <file name> IN USE AT DESTINATION--PUT ABORTED

This message is displayed when SYCCM is attempting to "send" a file to the other system and the specified file is currently in use on that system.

FILE <file name> IN USE AT DESTINATION--FETCH ABORTED

The file in question has been received and when closing it (which removes existing named file), finds that the file to be replaced is in use. The FETCH is aborted.

FILE <file name> NOT FOUND

The object file for a PUT file transfer was not found in the disk directory (disk) or was not online (tape or disk pack). B1800/B1700 SYSTEMS COMMUNICATIONS MODULE12 Sep 1978Software Operational GuidePage = 150Section VI Appendices(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

.COPY (continued):

FILE <file name> LOCKED

The object file for a PUT file transfer was found to be "locked" in the disk directory (disk).

FILE <file name> FILE TYPE CANNOT BE TRANSMITTED

The disk file specified for the file transfer is not one of the file types which SYCOM can transfer. (No code files disk-tape or tape-disk, only data files.)

FILE <file name> SOURCE OR DESTINATION FILE NAME MISSING

This message indicates that a subject file name was missing from the command string which was entered as either a "COPY", "PUT", or "FETCH" from another system.

FILE <file name> SOURCE OR DESTINATION FILE NAME INVALID

The specified file name was found to contain characters other than those which are considered valid for a file identifier; or, a multi-file identifier was indicated ("/") but was not included.

FILE <file name> FILE NAME WAS IDENTIFIED WITH MORE THAN 10 CHARACTERS

The name of the file to be transmitted was found to contain an identifier in excess of 10 characters. Its the maximum allowable.

ONTO-PACK CR USER-CODE NOT FOUND--PUT ABORTED

This message would occur when SYCOM was attempting to setup for a disk (or pack) file to be received and the destination "disk pack" was not found to be online. After the message is displayed the transfer is aborted.

12 Sep 1978

Software Operational Guide

Page - 151

(Original Version)

Section VI Appendices

\_\_\_\_\_

Command Error Messages (continued)

COMMAND MESSAGE REASON

•COPY (continued):

FILE <file name> ALREADY EXISTS AT DESTINATION - PUT ABORTED

When SYCOM is setting up to "receive" a disk (or pack) file, it checks to disk directory. If the specified file already exists, this message is displayed and the transfer is aborted.

FILE <file name> RECORD SIZE TOO BIG

The record size for the disk or tape file specified is too large for SYCOM to transfer with the present "Record Size" setting. The file cannot be transferred.

FILE <file name> TOO MANY RECORDS (>99999)

The disk (or pack) file specified for the file transfer contains an excessive number of records; outside SYCOM limitations. The file cannot be transferred.

FILE FAMILY NOT ALLOWED

As suspected, in the specified file name, an equal sign ("=") was found which represented either a "pack-id", "multi-file-id", or "file-id". This is illegal; the file named must be specific.

•DEBUG:

ERROR - "ON" OR "OFF" EXPECTED

The option keyword entered with the DEBUG command is not valid. Please try again.

B1800/B1700 SYSTEMS	COMMUNICATIONS	MODULE	12 Sep 1978
Software Operational	Guide		Page - 152
Section VI Appendice	2 <b>S</b>		Driginal Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

DISCONNECT:

\_\_\_\_\_

STATION ALREADY DISCONNECTED

This only means that a "DISCONNECT" command was entered from a terminal that was already disconnected; it was ignored.

•E0J:

STOP OR WAIT IN PROGRESS.

This means that a ".WAIT" command has been entered and SYCOM is responding to this request; the ".EOJ" command is ignored.

FETCH:

CANNOT FETCH WHEN REMOTE SYCOM OFFLINE

This message is indication that an inter-system line connection does not presently exist and the "FETCH" request cannot be accommodated.

SOURCE AND HOST NAMES MAY NOT BE DEFINED IN A FETCH

Use of the shorthand version (FETCH) of the COPY command implies the source and destination systems, therefore, they cannot be specified.

FILE <file name> IN USE AT DESTINATION--FETCH ABORTED

The file in question was received and when attempting to close it (which removes any existing file copy), finds that the file to be replaced is currently in use. The FETCH is aborted and the received file is lost; it must be requested again.

12 Sep 1978

Software Operational Guide

Page - 153

Section VI Appendices

(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

.LIST:

UNRECOGNIZED OPERAND "<token>"

This simply means that an incorrect or misspelled option keyword was entered; and rejected.

• OFFLINE:

STOP OR WAIT IN PROGRESS

This ressage indicates that SYCOM is in the process of servicing a ".WAIT" command which in effect produces an "offline" condition; the ".OFFLINE" command is ignored.

•ONLINE

This command does not cause an error message.

• P HONE :

#### INVALID PHONE NUMBER

This message is returned when the phone number entered is either more than 20 digits in length or contains alphabetic characters.

•PUT:

CANNOT PUT WHEN REMOTE SYCOM OFFLINE

This message means that an inter-system connection does not presently exist, therefore, a "PUI" cannot be done.

Software Operational Guide

Page - 154

Section VI Appendices

(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

.PUT (continued):

SOURCE OR DESTINATION HOST NAMES MAY NOT BE DEFINED IN PUT

Use of the shorthand version (PUI) of the COPY command implies the source and destination systems, therefore, they cannot be specified in the command string.

FILE <file name> IN USE AT DESTINATION--PUT ABORTED

This message occurs when SYCOM is attempting to "send" a file to the other system and the specified file is currently in use on that system. The PUT is aborted.

# •QS:

ERROR: NO "=" IN ". QS" COMMAND

This message occurs when a ".QS" command is entered and the required equal sign "=" was missing. The function requested by this command was not done.

ERROR: ".QS" OPERAND NOT NUMERIC

The ".QS" command is entered with a non-numeric queue size integer field and was ignored. It is necessary to re-enter the command string in correct format.

ERROR: "QS" OPERAND MAY NOT EXCEED 15

The queue size integer entered was greater than 15 and was summarily rejected. It must be remembered with an appropriate integer value.

ERROR: "OS" OPERAND MUST BE AT LEAST 4

The queue size integer entered was less than the value 4 and was rejected. The command must be re-entered with a legitimate integer value.

Software Operational Guide

12 Sep 1978

Page - 155

Section VI Appendices

(Original Version)

Command Error Messages (continued)

CUMMAND MESSAGE REASON

•RY:

UNRECOGNIZED COMMAND <RY xx>

The command string contained an urrecognizable option keyword; not "CP", "LP", or "CR".

•SIZE:

ERROR: NO "=" IN ".SIZE" COMMAND

This message is clear, the command syntax "requires" an equal sign (=) prefixing the size integer.

ERROR: ".SIZE" OPERAND NOT NUMERIC

The integer entered with the command string was comprised of at least one non-digit character. The integer must be all digits, no exceptions.

ERROR: ".SIZE" CANNOT EXCEED 400

This message is displayed when a size integer is entered which exceeds the maximum message block size established for the system; four hundred (400) bytes.

ERROR: ".SIZE" CANNOT BE LESS THAN 78

This message is displayed when the size integer entered is less than the minimum allowable message block size; for this system the minimum size is 78 bytes.

12 Sep 1978

Page - 156

Software Operational Guide

Section VI Appendices

(Original Version)

Command Error Messages (continued)

COMMAND MESSAGE REASON

# •STOP:

This command cannot directly cause an error message.

# •SV:

UNRECOGNIZED COMMAND <SV <xx>>

The .SV command was entered either without an option or the option entered was found to be invalid (not "LP" or "CP").

### .USER:

=====

INVALID USERCODE NAME

The usercode used to log-on with has been found to be invalid. Please log-off and log-on again with correct usercode/password combination.

INVALID PASSWORD NAME

The password which the user signed-on with has been found to be invalid. Please log-off and log-on again with a correct usercode/password combination.

12 Sep 1978

Software Operational Guide

Page - 157

Section VI Appendices

(Original Version)

Command Informational Messages

COMMAND MESSAGE REASON

SYCOM IS CLOSING; CANNOT ACCEPT INPUT

This message is returned in response to any new command entered "after" an ".EOJ" command is entered but prior to the actual shutdown of SYCOM. (Once shutdown has been initiated, commands are no longer accepted.)

.READ:

REQUEST IGNORED: CARD FILE STILL OPEN

This ressage is displayed if a ".READ" command is entered when an input card file is currently active. The entire command string is rejected and discarded.

REQUEST IGNORED: NOT LOGGED ON

This message could mean the card reader is not ready.

FILE "RJECARDS" NOT PRESENT, TRY AGAIN

To be defined.

.COPY:

======

CAN'T SCHEDULE NOW; TRY AGAIN LATER

To be defined.

REQUEST SCHEDULED; WILL SEND FILE <file name> ASAP

To be defined.

REQUEST SCHEDULED -- REMOTE SYCOM OFFLINE

To be defined.

12 Sep 1978

Software Operational Guide

Page = 158

Section VI Appendices

(Original Version)

System Parameter Error Messages

The following error messages could occur during the parameters input process. The problem indicated applies to both SPO input as well as card input. In either case, SYCON must be DS'd, the problem(s) corrected, and the parameters reentered as required.

MESSAGE REASON

% SYCOM =<job #> INVALID PARAMETER: <token>

This message is displayed when the parameter keyword entered did not match one of the acceptable keywords.

% SYCOM =<job #> "=" EXPECTED BUT GOT <token>

The parameters which include a value must be separated by the "=" character. In one of the parameters entered this word separator was missing and SYCOM is unforgiving in these matters.

% SYCOM =<job #> INTEGER EXPECTED BUT GOT <token>

The BUFFER and RECORD parameters require an integer which specifies the BUFFER/RECORD size in bytes. As this is not what was entered, what was found is included in the message.

X SYCOM =<job #> BUFFER SIZE EXCEEDS NAXIMUM ALLOWABLE

The maximum allowable BUFFER size is 1950 bytes. What was specified exceeded this limit.

% SYCOM =<job #> RECORD SIZE EXCEEDS MAXIMUM ALLOWABLE

The maximum allowable RECORD size is 1950 bytes. What was entered exceeded this limit.

Z SYCOM =<job #> INVALID SYSTEM ID

The "system id" entered has exceeded the parameter limit of 17 bytes. Please shorten it up some and reenter it.

12 Sep 1978

Software Operational Guide

Page - 159

Section VI Appendices

(Original Version)

System Parameter Error Messages (continued)

MESSAGE REASON

% SYCOM =<job #> RECORD SIZE EXCEEDS BUFFER SIZE

The integer entered for the parameter "RECORD" exceeds the current "BUFFER" size. Do it again.

X SYCOM =<job #> INVALID PARAMETERS FILE, BEING REMOVED

This message occurs when the "PARAMETERS" file is opened, the parameters are read in, and the "maximum buffer size" was found to exceed the system maximum of 1950 bytes. When this condition is encountered, the "SYCOM/PARAMETERS" file is removed and new parameters must be entered.

12 Sep 1978

Software Operational Guide

Page - 160

(Original Version)

Section VI Appendices

System Warning Messages

Warning messages displayed on the User SPO indicate a problem at the system level. They have been rectified by SYCOM so that operations can continue without operator intervention. They are not generally considered to be major or extensive in nature and therefore, SYCOM has elected to take immediate corrective action.

MESSAGE REASON

% SYCON =<job #> REMOTE FILE OVERFLOW, MSG DISCARDED % SYCOM =<job #> <60 or less characters of message>

Self explanatory.

% SYCOM =<job #> INQ OVERFLOWP FOLLOWING MSG DISCARDED % SYCOM =<job #> <60 or less characters of message>

Self explanatory.

% SYCOM =<job #> TERMINALS FILE OVERFLOW, MESSAGE DISCARDED % SYCOM =<job #> <60 or less characters of message>

Self explanatory.

% SYCOM =<job #> ERROR: MESSAGE TYPE = <type code>

This message indicates that the input message type received on the remote file "LINK" is incorrect and the message has been shunted off.

% SYCOM =<job #> ERROR: INCOMPLETE IO ON LINK FILE

To be defined.

% SYCOM = < job #> LINK QUEUE OVERFLOW, MESSAGE DISCARDED

To be defined.

% SYCOM =<job #> QUEUE RJEQ FULL, MESSAGE DISCARDED

To be defined.

12 Sep 1978

Software Operational Guide

Page - 161

Section VI Appendices

(Original Version)

System Warning Messages (continued)

MESSAGE REASON

% SYCOM =<job #> NUMBER OF TIMEOUTS = <number>

This message is displayed automatically when "timeouts" keep re-occurring on the inter-system line connection. However, it is only output on an incremental basis to reduce SPD output activity.

X SYCOM = < job #> LOSS OF DATA-SET-READY

Display of this message occurs each time the indicated event is detected. A count is maintained of these occurrences and SYCOM goes into the state of "initialization". In progress file transfers are aborted and lost.

2 SYCOM =<job #> HOST SYSTEM CARD READER HALTED

When this message occurs it is to inform the system operator of a condition that presently exists on the remote system. No action is required by the operator.

% SYCOM =<job #> HOST SYSTEM CARD READER ABORTED

This message is used to inform the system operator that card reader input from the remote system has been aborted. No action is required of the operator.

2 SYCON =< job #> HOST SYSTEM CARD READER READY

A message that is used to inform the local system operator that card input from the remote system has resumed.

% SYCOM = < job #> SYCOM WILL CLOSE WHEN NEXT IDLE

Self explanatory.

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE12 Sep 1978Software Operational GuidePage = 162Section VI Appendices(Original Version)

System Warning Message (continued)

MESSAGE REASON

Z SYCOM =< job #> SYCOM CLOSING ON REQUEST FROM REMOTE SYCOM

Self explanatory.

% SYCOM =<job #> ERROR: MESSAGE TYPE = <message type>

An input message from the remote file "LINK" was found to be invalid. The message type code is included in the displayed error message for reference.

2 SYCOH = < job #> REMOTE SYCOM HAS GONE DOWN AND COME BACK UP

This message is self explanatory.

12 Sep 1978

Software Operational Guide

Page - 163

Section VI Appendices

(Original Version)

System Condition Messages

Any of the following messages displayed on the User SPD indicate an abnormal system condition of some type. As a rule, when one of these conditions is encountered, the display of an appropriate message also elicits a response from the system operator.

\_\_\_\_\_

MESSAGE REASON

% SYCOM =<job #> CARD PUNCH EXCEPTION % SYCOM =<job #> PLEASE ENTER ".RY CP" WHEN DEVICE IS READY OR ".SV CP" TO SAVE THE FILE

This message is displayed when SYCOM is attempting to punch a card and an "exception" condition is encountered. Generally this indicates that the card punch has gone "not ready" for any one of several reasons. It is necessary to respond to SYCOM with the appropriate command when action has been taken to ready the unit.

% SYCOM =<job #> LINE PRINTER EXCEPTION % SYCOM =<job #> PLEASE ENTER ".PY LP" WHEN DEVICE IS READY OR ".SV LP" TO SAVE THE FILE

This message is displayed when SYCOM is attempting to print a line and an "exception" condition is encountered. Generally this indicates that the line printer has gone "not ready" for any one of several reasons. It is necessary to respond to SYCOM with the appropriate command when action has been taken to ready the unit.

% SYCOM =<job #> PARITY ERROR ON READ PLEASE READ LAST CARD AGAIN

This message is displayed when SYCOM is attempting to read the card file and an "exception" condition is encountered. Generally this indicates that the card reader has gone "not ready" for any one of several reasons. It is necessary to respond to SYCOM with the appropriate command when action has been taken to ready the unit.

12 Sep 1978

Page - 164

Software Operational Guide

Section VI Appendices

(Original Version)

¥.

System Condition Messages (continued)

MESSAGE REASON

% SYCOM =<job #> PUNCH ERROR

The card punch has either detected an error in the last card punched or has degraded into a "not ready" state. In either case, a "suspend" control message has been sent to the Server System.

% SYCOM =< job #> READER ERROR

The card reader has either detected an error in the last card read, gone into a "not ready" state, or has detected an invalid character in a card column; other than column 1. If there was a read error, the last card must be put back into the card hopper. The card reader is then made physically "ready"; all cases. SYCOM must then be informed of this new status condition with the entry of a ".READY" command.

% SYCOM =<job #> PRINT ERROR

The line printer has either detected an error on the last line printed or has gone into a "not ready" state. In both cases a "suspend" control message is sent to the Server System. When everything is again ready with the printer, a ".READY" command must be entered to inform SYCOM.

X SYCOM =< job #> CARD READER CLOSED

This simply means the card input has been completed and the card file closed.

% SYCOM =<job #> QC DONE

In doing a read on the remote file "LINK" an "EOF" condition was detected. This condition is caused by the NCP when a "QC" command is entered; which shuts down all programs associated with the Network Controller.

12 Sep 1978

Page - 165

Software Operational Guide

Section VI Appendices

(Original Version)

System Condition Messages (continued)

MESSAGE REASON

X SYCOM =<job #> ERROR: INCOMPLETE IO ON LINK FILE

the state of the second state of the

In doing a read on the remote file "LINK" an "incomplete io" condition was detected. This message werely reports on the situation.

% SYCOM = < job #> PHONE NUMBER REJECTED BY NC <phone number>

To be defined.

% SYCOM = <job #> INVALID CONTROL MESSAGE RECEIVED

When SYCOM receives a system "control ressage" from a remote system and the "type code" in this message is not within the 00 = 23 number range, then the above message is displayed. The input message is discarded and no further processing is done on its behalf.

% SYCOM =<job #> BUG IN NETWORK CONTROLLER. VTT INDEX = <vtt>

A "close" message has been received from the system Network Controller for a program known to SYCCH which contained a "lsn list". This list included a Virtual Terminal Table Index (VTT INDEX) entry for a terminal not signed-on to that program, thus the reason for this message.

% SYCOM =<job #> LOST CONNECTION TO REMOTE SYSTEM SYCOM

This message is self explanatory also.

Software Operational Guide

Page - 166

Section VI Appendices

(Original Version)

System Fatal Messages

These messages are of the type which preclude the automatic shut down of SYCOM. There are considered "fatal" in the outlook of continuing operations and therefore, SYCCM is shut down.

MESSAGE REASON

% SYCOM =<job #> FATAL ERROR AT <program sequence #>

The display of this message on the User SPO could be the result of any number of reasons. The SYCON source sequence number displayed is the only clue to the problem. SYCON initiates a "memory dump" at this point and automatically goes to EOJ after certain disconnect functions have been satisfied.

% SYCON =<job #> REMOTE VERSION OF SYCOM IS INCOMPATIBLE % SYCON =<job #> HIS VERSION="<x>", MINE="4"

This message identifies the problem at hand. SYCOM goes to EOJ immediately at this point.

% SYCOM =<job #> ANOTHER SYCOM WITH SAME MULTI-FILE-ID IS RUNNING

The display of this message indicates that SYCOM cannot gain access to the "SYCOM/FIP.QUEUE" file (cn disk) as it is presently in use by another program.

Z SYCOM =<job #> PRINT.PUNCH.QUEUE OVERFLOW: PROGRAM ABORT

To be defined.

X SYCOM =<job #> REMOVE FTP.QUEUE OR CHANGE ITS NAME IF 7.0 FILE

To be defined.

12 Sep 1978

Software Operational Guide

Page - 167

Section VI Appendices

(Original Version)

APPENDIX B: CHARACTER SFT

The following printable characters define the character set acceptable to SYCCM for input messages from terminals. They are arranged by columns in internal collating sequence with "blank" low and "9" high:

\_\_\_\_\_

bl	ank	8	- :,	?	9	h	0	<b>V</b>	<b>C</b>	J	0	₩	3
	C	]	1	<b>:</b>	b	i	p	" <b>H</b>	D	K	R	X	4
	•	\$	a	#	С	j	Q	x	E	L	N	Y V	5
	<	*	•	3	d	k	r	<b>y</b>	F	M	S	2	6
	C 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	)	7.	•	e	L	<b>S</b> .	2	G	N	T	0	7
	<b>+</b>	;	-	=	f	en.	t	A	Н	0	U	1	8
	ŧ.	-	>	•	9	<b>n</b>	u	8	I	P	V	2	5

It is recommended that the standard "signal character" of "\*" be retained by the SYCOM user for use in message transfers to the SMCS. However, if this is not suitable for any reason, one of the following special characters are suggested:

E & E % ] # @ +

Special data communications characters recognized by SYCOM are as follows:

SOH EOT DLE ENQ STX EIX RS FS

Mr. N. B.

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
Software Operational Guide	Page - 168
Section VI Appendices (Origin	al Version)

APPENDIX C: RESERVED WORDS

----------

The reserved words recognized by SYCON are dependent upon the position type format. Hence the following list of reserved words are divided into the following types:

Type 1 - Commands
Type 2 - Specifications

Words which are immediately followed by a word in parentheses are interchangeable alternates for that word.

		Type 1		
-ABORT	•CP	LIST	•PHONE	•SIZE
•BYE	• DEBUG	.LOG	PUT	.STOP
•CL	• DISCONNECT	LP	• Q S	•SV
.CONNECT	•E0J	.OFFLINE	-READ	.US (USER)
-COPY	•FETCH	DNLINE	- R Y	-WHAT

÷		2	1	y	ρ	e		Z		
	1	-	÷	۵	-	••	•	-	•	-

AS	FROM	OFF	PASSWORD	TAPE7 USER		
СР	HOST	ÓN	RETAPE	TAPE9 USERS		
CR	KIND	ONTO	PUT	TERMINALS		
DISK	LP	PACK	SCHEDULE	TO		
FETCH	NR	PARAMETERS	TAPE	US		
81800/81700	SYSTEMS	COMMUNICA	TIUNS MODULE	- 1. 1. 1. - 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	12 Sep	1978
--------------	-------------------	------------------------	--------------	--	--	--
	· · · · · ·	a far e a strategio de				100 N.
Software «Op	e <b>r</b> ationa	l Guide			Page -	169
					and the second	

Section VI Appendices

(Original Version)

APPENDIX D: COMMAND SUMMARY . Charles Barris

and the second second

الحاري والم

The following command summary list is presented in alphabetical sequence for easy reference. Refer to the command descriptions for command syntax and option details.

COMMAND	STATION	FUNCTION
ABORT	any	Abort a PUT or FEICH file transfer.
BYE	any	Station log-off from SYCOM.
CL	SPC	Clear peripheral device.
CONNECT	any	Request establishment of a terminal to a Server System.
COPY	any	File transfers (PUT or FETCH).
CP	SPO	Inform SYCOM of "not ready" peripheral.
DEBLG	SPC	Start/stop input/output message tracing.
DISCONNECT	any	Request for terminal disconnection from the Server System.
EOJ	SPO	Cause SYCOM into an orderly shutdown.
FETCH	any	File transfers (fetch from other system)
LIST	any	Cisplay selective system information.
LUG	SPC	Display a list of cata comm error counts
LP	SPO	Inform SYCOM line printer "not ready".
OFFLINE	SPO	Cause SYCOM to break the connection to a Server System.
ONLINE	SPC	Establish a Server System connection.
PHONE	SPO	Establish or clear phone number.

B1800/B1700 SYSTEMS	COMMUNICATIONS M	ODULE 12 Sep 19	78
Software Operationa	l Guide	Page - 1	70
Section VI Appendic	es	(Original Versio	n)

COMMAND	STATION	FUNCTION
PUT	any	File transfers (put to other system).
0S	SP 0	Change Network Controller queue depth.
READ	SPO	Open and transmit card input file.
RY	SPO	Inform SYCOM of available peripheral.
SIZE	SPO	Change maximum message block size.
STOP	SPO	Cause SYCOM to shutdown immediately.
SV	SPO	Stop transmitting and "save" file.
USER	any	Log a user onto SYCOM.
WHAT	any	SYCOM status inquiry.

12 Sep 1978

Software Operational Guide

Page - 171

Section VI Appendices

(Original Version)

APPENDIX E: COMMUNICATIONS CUNTROL MESSAGES

× 11 11

#### GENERAL

-----

As there are several message formats used for inter-SYCOM data communications, each format is specified in detail. These message formats are presented in a sequence dictated by the message type code. A list of message types are shown below, followed by individual message format descriptions.

### MESSAGE TYPES

----

次々 課金という

The message type codes are derivatives from the "DA1"/"DA2" (Device Address) field. The message type codes apply to message transmissions in either direction and are compatible between system types. Following then are the device address conventions used by SYCOM; all other values are considered invalid:

DA1	DA2	Usage			
0	C C	System con	trol		
0	2	Card reade	r/punch dat	a messages	
0 0 - 23 - 20	у 4% м. т. т. 20 т.	File trans	fer data me	ssages ssages	
8	2	Packed car	d punch dat	a messages	
a thr	A A	User Syste "receive",	m "transmit virtual te	", Server rminal mes	System sages
Z	2				
jAj≋ ≜∵thr	ส์ ละ 5-4. น.ศ. 197	User Syste	m "receive" • virtual t	<ul> <li>Server S</li> <li>erminal me</li> </ul>	y stem ssages
2					

Device addresses are advisory only. The receiving system can direct a message to any appropriate peripheral device, based on its own requirements.

12 Sep 1978

Software Operational Guide

Page - 172

Section VI Appendices

(Original Version)

Note that the line printer data is allowed in both directions. When a system receives line printer data, it can print it on any available printer, independently of whether the system is termed a User System or Server System.

MESSAGE FORMATS

Basic Data Communication Messages

As per the line protocol, the below message format adheres to the criteria of standard RJE message formatting:

SATSDD			EB
ORMTAA<	 TEXT	 >	TC
HM#X12			XC

The individual characters in this message function as follows:

- A. SOH This is the standard ASCII character which is used to indicate data transmission as opposed to connection establishment communication.
- B. ARM The "Acknowledge Received Kessage" character which is used to accept or reject the last message block received.
- C. TM# The "Transmission Number" character used in the acknowledgement of a message. It represents one of two values (alternating characters) selected for implementation by SYCOM. The characters "5" and "&" are used by the User System SYCOM whereas the characters "F" and "U" are used by the Server System SYCOM.
- D. STX A standard control character used to delimit the preceding header characters from the message.
- E. DA1/DA2 Two characters representing a "Device Address" value.
- F. TEXT The message text field usually containing data (record images or a message).

12 Sep 1978

Software Operational Guide

Page - 173

Section VI Appendices

(Original Version)

C. EfX - the standard "end-of-text" control character.

H. BCC - A Block Check Character which is comprised of a longitudinal parity bit configuration that is used, by the hardware, to verify that the message has been received correctly.

System Control Message

System control messages are used to exchange control type data between systems. A complete descriptive list follows the message format which is a variation of the basic message format:

> SATSDDCC EB GRMTAADD<---- variables (text) ---->TC HM#X1212 XC

"00"

Here we have the addition of the "CD1"/"CD2" characters. These two characters represent the system message "type" code; which may or may not be followed by variable control data.

SPC Messages

The SPO Message format uses the basic format without alteration. There are no provisions for blocking SPO Messages, therefore, the rule is: one SPC Message per control message block. The format of this type message is as follows:

SATSDD		5 E B
ORMTAA<	TEXT .	 >TC
HM#X12		XC

"01"

가지 않는 것은 것은 것은 것은 것은 것은 것이다. 가지 않는 것은 것은 것은 것이다. 같은 것은 것은 것은 것은 것은 것이다. 것은 것은 것은 것은 것은 것은 것이다. 것은 것은 것은 것은 것은 것이다.	
11800/B1700 SYSTEMS COMMUNICATIONS MODULE	12 Sep 1978
oftware Operational Guide	Page - 174
ection VI Appendices	(Original Version)
Card Reader/Punch Message	
The format for card reader/punch data image "PCC" character represents a "printer contr chis case is "O". The "RS" (Record Separat between card images provides the means for	s is as follows: The cl character" which in or) character used blocking.
SATSDDP	<b>ΕB</b>
ORMTAAC< card image>S< HM#X12C	card image>TC XC
<b>"020"</b>	
ine Printer Nessage	
ata lines between systems:	ransmit Line Printer
SATSDDP	EB
ORMTAAC< line image>S< HM#X12C	line image>TC XC
"03x"	
This format is the same as the Card Reader/ that the "PCC" character is utilized. The	Punch format except value of this field
ban be end et end terrenting bedebe	

PCC Function -----0 Print, no spacing Print, single space 1 2 Print, double space Skip to channel 1 (top of page) A 8 -2 C 3 -Ð m 4 E 5 . F = G 7 -8 9 H ee, -I J 10 K -11 L 12 (bottom of page) Ŷ.

12 Sep 1978

Software Operational Guide

Page - 175

Section VI Appendices

(Original Version)

File Transfer Message

The File Transfer message format is also the same as that used to transfer card images. Although the "PCC" character field is included, it is only there to maintain message conformity:

> SATSDD R EB ORMTAA<---- record data ---->S<---record data ---->TC HM#X12 XC

"04"

Virtual Terminal Message

The message format utilized for virtual terminal messages is the basic message format, with one exception: the "device address" for a virtual terminal is comprised of two letters (1 lower, 1 upper "transmit": 1 upper, 1 lower "receive") and will vary by terminal. The format is as follows: (Terminal and SPO messages are "not" blocked.)

SATSDDEBURMTAA<</td>message textHM#X12XC

"xX" (transmit) "Xx" (receive)

File Separator Messages

This message format is utilized by SYCOM when it is necessary to transfer a "symbolic file header", indicate the "beginning of a file", or indicate the "end of a file". This applies equally to the "Card Reader/Punch", "Line Printer", and "File Transfer" operations. To accomplish this, there are three minor variations to the File Transfer Message format which are identified with the following message formats. B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 176 Section VI Appendices (Original Version) Begin File Message The format for the Begin File message is as follows: SATSDDF EB ORMTAAS1<---- file name ---->TC HM#X12 XC "02" \*03\* \*04\* End File Message The format for the "end file" message is as follows: SATSDDF EB **ORMTAAS2TC** HM#X12 XC Program-to-Program Messages The following message format pertains to SYCON's "program-to-Program" facility; thru a "Remote File": (10 characters) % Message "header". REMOTE.KEY - RSN ( 3) Z Relative Station Number for % "virtual terminal" as defined % for the program-to-program % remote file; usually "001". - LENGTH ( 4) % Length of message text. TYPE (3) % Message type; usually "000". MESSAGE TEXT (nn characters) % Maximum is "buffersize".

Program-to-SYCOM Messages

To be specified when implemented.

12 Sep 1978

Software Operational Guide

Page - 177

Section VI Appendices

(Original Version)

APPENDIX F: RESTRICTED MCP COMMAND SUMMARY

All input commands that are prefixed with a "?" are sent directly to the MCP exactly as entered. The MCP then generates appropriate responses which are sent to the terminal via the Network Controller. Security restrictions have imposed certain constraints on MCP commands from remote terminals. Also, certain commands, such as those which alter system options, purge, or clear system peripherals, have been disallowed for a remote terminal. The following list represents the MCP command subset permitted from tser System terminals and the User SPO.

The following Command Subset is listed in the order as described in the B1800/B170C Systems System Software Operational Guide (associated documentation). Any command listed below which is suffixed with "US" requires the command to be entered with a valid "usercode"/"password" combination.

Library Maintenance Instructions

•	CH	Change (US)
•	A D	Add (US)
•	LC	Load (US)
•	DU	Dump (US)
•	UN	Unload (US)
•	RE	Remove (US)

Program Control Instructions

- CO Compile (US)
- DY Dynamic
- EX Execute (US)
- MH Modify Header (US)
- . MO Modify (US)

Software Operational Guide

Page - 178

Section VI Appendices

		-		· •		•								•				
		- 11	-		~	•	- 62	-		•	~	-	-	•	-	-	•	
		- EL			•													۲
	-		•	•	- 22	•		-	-		•	•						
					-													

Program Control Instruction Attributes

•	AF	After
•."	AN	After - Number
•	CA	Conditional
•	CG	Charge (certain restrictions apply
•	DS	Dynamic.Spaces
•	FI	File
•	FR	Freeze
• -	HO	Hold
•	IN	Interp interpreter
• .'	IT	Intrinsic.Name
•	ID	Intrinsic.Directory
•	ME	Hemory
•	O V O	Override
•	PR	Priority
•	SC	Schedule.Priority
•	SW	Switch
•	TH	Then
• •	UC	Unconditional
•	UF	Unfreeze
• .	UV	Unoverride
•	VI	Virtual.Disk

System Control Instructions

•	AX	Response to ACCEPT
•	BF.	Display BACKUP files (US)
•	CP	Compute
•	CU	Core usage
•	DF	Date of file (US)
	DM	Dump memory and continue
•	DP	Dump memory and discontinue
•	DS	Discontinue program
•	FN	Disolay internal file name
•	FR	Final reel of unlabeled tape file
• •	FS	Force from schedule
•	GO	Resume stopped program
•	HS	Hold in waiting schedule
•	HW	Hold in waiting schedule until job FOJ
•	MR	Close output file with purge
•	MX	Display MIX
•	OF	Optional file response
•	OK	Continue processing
•	OL	Display peripheral status
• •	PD	Display directory
•	PM	Print memory dump
•	PR	Change priority

Page - 179

Software Operational Guide

Section VI Appendices

#### (Original Version)

System Control Instructions (continued)

•	PS	Prod schedule
•	QF	Query file (US)
•	QP	Query program
•	ន ម	Remove backup files (US)
•	IS M	Remove duplicate disk file
•	RS	Femove jobs from schedule
•	SP	Change schedule priority
•	ST	Suspend processing
•	SW	Set switch
•	T D	Time and date
•	TI	Time interrogation
•	TS	Test switches
•	UL	Assign unlabeled file
•	WD	Display MCP date
•	WM	Display current MCP and interpreter
•	WS	Display schedule
•	WT	Display MCF time
•	WY	Program status interrogation

Job Spawning Control Attributes

OU Queue command
US Usercode command
ZQ Zip queue command

MIX related MCP commands may refer only to jobs initiated by the same remote terminal where the initial command was input. All MCP commands must be entered in UPPER CASE.....

Software Operational Guide

Section VI Appendices

(Original Version)

\_\_\_\_\_

Page - 180

APPENDIX G: PARAMETERS FILE ATTRIBUTES

The PARAMETERS file is a disk file, data type, with one physical record of 180 characters. This parameters record is formatted as shown below.

Record Format

Field (i	Length n bits)	Default Value	Description
Filler	105	<b>97</b> 98	Reserved for identifier.
CURRENT.DATE	72		Run date as "dd mm yy"; (9 bytes).
STATION.ID	136	47 98	System identifier; 17 bytes.
STATION.ID.LENGTH	24	0	Length of system identifier.
MAX.BUFFER.SIZE	24	400	File Transfer buffer size.
REC.SIZE	24	180	Naximum record size for RJE message block transmissions.
TIMEOUT.INTERVAL	24	· · · · · · · · · · · · · · · · · · ·	A value for message display intervals.
AUDITING	<b>1</b>	0	DEBUG on/off switch. Default of "O" inhibits trace list.
QUE.P	1	n	When "set", this parameter permits "program-to-SYCOM" communications.
NOFTP.P	1	0	To be specified.
NOFS	1	0	To be specified.
WAIT.P	1	0	To shutdown or not shutdown when receiving a "disconnect message" from another SYCOM. 0 = shutdown; 1 = idle.

Software Operational Guide

Section VI Appendices

	• • •	
	inal Ve	r 510D)

	Length (in bits)	Default Value	Description
Filter	3	000	Reserved for future use.
MY . NAME	240		SYCCM's name; 30 bytes.
MY • JOB • NO	32		SYCOM'S "job #"; 4 bytes.
Filler	128		Reserved for identifier.
COMPILE.DATE	72	99 99 99 10 10 10 10 10 10 10 10 10 10 10 10 10	Compile date as "dd mmm yy"; (9 bytes).
Filler	128	** **	Reserved for identifier.
COMPILE.TIME	08	<b>17 11</b>	Compile time as "hh:mm:ss.t" (10 bytes).
Filler	334		Reserved for future use

12 Sep 1978 Page - 181

B1800/B1700 SYSTEMS COMMUNICATIONS	MODULE 12 Sep 1978
Software Operational Guide	Page - 182
Section VI Appendices	(Original Version)
	이는 것은 가지 않는 것 같은 것이 있었다. 가지 않는 것 같은 것을 가지 않는 것을 가 있다. 가지 않는 것을 수가요. 이렇게 있는 것을 가지 않는 것을 것을 수가요. 이렇게 있는 것을 것을 수가요. 같은 것을 것을 수 있는 것을 것을 것을 것을 것을 것을 것을 것을 것을 수 있는 것을

APPENDIX H: SUPMARY OF FILES

Program	Internal Name	External Name	Device	Use	Size
SYCOM	RJECARDS	RJE/CARDS	Reader	I	80/1
	RJEPUNCH	RJE/PUNCH	Punch*	0	80/1
	RJEPRINT	RJE/PRINT	Printer*	0	132/1
	LINK	LINK	Remote	1/0	455/1
	LINE	LINE	Printer*	0	132/1
	IN	<file name="">**</file>	Disk/Tap	eI	varies
	OUT	<file name="">**</file>	Disk/Tap	e 0	varies
	TERMINALS	TERMINALS	Remote	1/0	1980/1
	INQ	SYCOM/INQ	Queue	1/0	360/1
	OUTQ	SYCOM/OUTQ	Queue	1/0	360/1
	LINK.Q	SYCOM/LINK.Q	Queue	1/0	2000/1
	FTP-QUEUE	SYCON/FTP.QUEUE	Disk(R)	1/0	354/1
	RJE.Q	SYCOM/RJE.Q	Queue	1/0	455/1
	FP	SYCOM/PARAMETERS	Disk	1/0	180/1
	CARDS	CARDS	Reader	I	80/1
SYCOM/NC	LINK	LINK	Remote	I/0	455/1
SMCS	MCSQUEUE	NCSQUEUE	Remote	I/0	455/1

\* Note: Printer or printer backup disk file.

\*\* Note: The actual file name as specified in the COPY command.

12 Sep 1978

Software Operational Guide

Page - 183

Section VI Appendices

(Original Version)

APPENDIX I: SAMPLE APPLICATION PROGRAMS

This program was designed basically to test "program-to-program" communication functions of SYCOM. What this sample application program provides is the mechanics of what is necessary for this communication process. The user can expand upon this sample to create a program with the desired capabilities as required. The program is written in UPL which is available for this purpose.

The only data communication type functions required for this type of application program is to "read" input messages from a Remote File and "write" reply messages into the same Remote File. This message interfacing architecture has been established for this purpose and must be, therefore, adhered to at all times.

## PROGRAM-TO-PROGRAM (User System)

COMPILE USER/TEST UPL LI;

7 Th 7 co 7	is application unmunication u	n program ising SYCD	tests pr 4; in the	ogram-to "USER S	-program YSTEM"	1 · · ·
8 DETAIL 8 SIZE 2 8 EXPA 8 PASS_E	ND_DEFINES X	(REF XMAP	7	(user o	ptional)	1999 - San San San San San San San San San San San San San San San
FILE	RFUSER (DEV REN	VICE = REM NOTE_KEY+	OTE, NUMB Records =	ER_CF_ST 80/1, 8	ATIONS = UFFERS =	= 1, = 2);
DEFINE	CH		AS #CHA	RACTER#;		·
DECLARE	TEXT		CH (80) CH (80)	;		

Software Operational Guide

Page - 184

(Original Version)

Section VI Appendices

DECLARE 1	RMT_KEY 2 RMT_RSN 2 RMT_LENGTH 2 RMT_TYPE	CH (10) CH ( CH ( CH (	3), 4), 3),
1	READ_KEY 2 READ_RSN 2 READ_LENGTH 2 READ_TYPE	CH (10) CH ( CH ( CH ( CH (	<ul> <li>3 ),</li> <li>4 ),</li> <li>5 );</li> </ul>
RMT_RSN RMT_LENG RMT_TYPE	:= "001"; GTH := "0080"; E := "000";	% First st % Standar % MCS type	tation in "family". d message length. e code.
OPEN RFL	JSER INPUT, OU	TPUT;	
DO TEST	LOOP FOREVER	andra andra anglesis. Produktional anglesis anglesis anglesis anglesis anglesis ang anglesis ang anglesis ang ang ang ang ang ang ang	
ACCI	EPT TEXT;		
% Could c	heck for speci	al input message	e for EOJ stuff.
WRI	TE RFUSER ERM	T_KEY] (TEXT);	<pre>7 Cutputs messages 7 Cutputs messages 7 directly to the 7 Network Controller 7 for delivery to the 7 cther system.</pre>
REA	D RFUSER ERE	AD_KEY] (REPLY)	; % Obtains input directly % for the Network % Controller.
ON	EOF STOP;		% In case of "QC"
DIS	PLAY REPLY;		
END TES	T_L00P;	andra ann an Aonaichte Ann an Aonaichte ann an Aonaichte Ann an Aonaichte ann an Aonaichte	
% The abov FINI;	e program Loop	b will never end	; it must be DS <sup>1</sup> d.
	*====		

	Uperational G	uide			Page - 18
Section V	/I Appendices			(Ariain	al Version
PROGRAM-	D-PROGRAM (Se	rver System	)		
*********			========================	. == = = = = = = = = = = = = = = = = =	. = = = = = = = = = = = = = = = = = = =
OMPILE	SERVERITEST	UPL LT;			an An an
K Thi	is sample appl	ication pro	gram tests	program-to	-program
CO	nmunication us	ing SYCOM;	in the "SEI	RVER SYSTEM	1*
B DETAIL					
B PASS_EI	ND			1	
-		NICC - 8540		6C CT.T.T.C.	16 - 1
ilt	REPERVEN	MOTE_KEY R	ECORDS = 8	C/1, BUFFE	$r_{2} = 1$
DEFINE	СН	A	S #CHARACT	ER#;	
DECLARE	TEXT	Ci	H (80).		
	1 RMT KEY	n .	H. (10).		
	1 RMT_KEY 2 RMT_RSN	CI	H (10), CH ( 3),		
	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT	CI H	H (10), CH ( 3), CH ( 4),		
	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE	CI H	H (10), CH ( 3), CH ( 4), CH ( 3),		
	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY	CI H CI	H (10), CH ( 3), CH ( 4), CH ( 3), H (10),		
	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN	CI H CI	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3),		
	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG	CI H CI	H (10), CH ( 3), CH ( 4), CH ( 3), CH ( 3), CH ( 3), CH ( 4),		
	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG 2 READ_TYPE	CI H CI	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3), CH ( 3);		
RMT_R	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG 2 READ_TYPE SN := "CO1";	CI H CI TH % Fir	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3), CH ( 3); st station	in "famil	
RMT_R RMT_L	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "CO8G	CI TH Z Fir Z Sta	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3), CH ( 3); st station ndard mess	in "famil age length	
RMT_R RMT_L RMT_L RMT_T	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "OOBG YPE := "OOD";	CI H CI TH X Fir X Sta X MCS	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3); St station ndard mess type code	in "family age length	
RMT_R RMT_L( RMT_T OPEN	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "CO8G YPE := "CO1"; RFSERVER INPUT	CI TH Z Fir Z Sta Z MCS	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3), CH ( 3); st station ndard mess type code	in "famil age length	
RMT_R RMT_L RMT_L RMT_T OPEN	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_KEY 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "OOBG YPE := "OOD"; RFSERVER INPUT	CI TH TH Z Fir Z Sta Z MCS F OUTPUT;	H (10), CH ( 3), CH ( 4), CH ( 3), CH ( 3), CH ( 3), CH ( 4), CH ( 3); st station ndard mess type code	in "famil age length	
RMT_R RMT_L RMT_T UPEN DD T	1 RMT_KEY 2 RMT_RSN 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_KEY 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "OOBO YPE := "OGO"; RFSERVER INPUT EST_LOOP FOREV	CI TH Z Fir Z Sta Z MCS F OUTPUT; ZEK;	H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3), CH ( 3); st station ndard mess type code	in "famil age length	
RMT_R RMT_LI RMT_T UPEN DD T R	1 RMT_KEY 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_KEY 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "OOBG YPE := "OOBG YPE := "OOBG RFSERVER INPUT EST_LOOP FOREW EAD_RFSERVER	CI TH 2 Fir 2 Star 2 MCS 7, OUTPUT; 2 EREAD_KEY]	<pre>H (10), CH ( 3), CH ( 4), CH ( 3), CH ( 3), CH ( 3), CH ( 3); st station ndard mess type code (TEXT); 2</pre>	in "family age length Input fro	n User AP
RMT_R RMT_L RMT_L RMT_T UPEN DO T R G	1 RMT_KEY 2 RMT_RSN 2 RMT_RSN 2 RMT_TYPE 1 READ_KEY 2 READ_RSN 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "OOBO YPE := "OOD"; RFSERVER INPUT EST_LOOP FOREW EAD_RFSERVER N EOF STOP; RITE RESERVER	CI TH Z Fir Z Star Z MCS Z MCS I OUTPUT; ZER; EREAD_KEY] ERMT KEY]	<pre>H (10), CH ( 3), CH ( 4), CH ( 3), H (10), CH ( 3), CH ( 3), CH ( 3); st station ndard mess type code (TEXT); % %</pre>	in "family age length Input fro In case o Output to	m User AP f "9C" User AP
RMT_R RMT_L RMT_T UPEN DD T R G W END T	1 RMT_KEY 2 RMT_RSN 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_KEY 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "CO8G YPE := "CO1"; RFSERVER INPUT EST_LOOP FOREV EAD_RFSERVER N EOF STOP; RITE RFSERVER EST_LOOP;	CI TH Z Fir Z Star Z MCS OUTPUT; Z MCS (EREAD_KEY) ERMT_KEY]	<pre>H (10), CH ( 3), CH ( 3), CH ( 3), CH ( 3), CH ( 3), CH ( 3); st station ndard mess type code (TEXT); % CH ( TEXT); %</pre>	in "family age length Input fro In case o Output to	m User AP f "qc" User AP
RMT_R RMT_L RMT_T UPEN DO T R G W END T	1 RMT_KEY 2 RMT_RSN 2 RMT_RSN 2 RMT_LENGT 2 RMT_TYPE 1 READ_KEY 2 READ_KEY 2 READ_LENG 2 READ_LENG 2 READ_TYPE SN := "CO1"; ENGTH := "OOBG YPE := "OOD"; RFSERVER INPUT EST_LOOP FOREW R EOF STOP; RITE RFSERVER EST_LOOP;	CI TH Z Fir Z Stan Z MCS F OUTPUT; Z READ_KEY] ERMT_KEY]	<pre>H (10), CH ( 3), CH ( 4), CH ( 3), CH ( 3), CH ( 3), CH ( 3); st station ndard mess type code (TEXT); % % (TEXT); %</pre>	in "famil age length Input fro In case o Output to	n User AP f "AC" User AP

SECTION VII

GLOSSARY

81 8007 8170	O SYSTEMS	COMMUNIC	CATIONS	MODULE		12 Sep	1978
Software C	perational	Guide	an a			Page	- 187
			· · · · ·				

Section VII Glossary of Terms and Acronyms (Original Version)

SECTION VII

GLESSARY OF TERMS

This glossary defines SYCOM terms and acronyms used throughout this software operational guide. For ease of use, the glossary is separated into two sections; Terms and Acronyms. Each section is sequenced alphabetically for guick reference.

## TERMS

----

TERM

DEFINITION

ACTIVE STATION:

A work station (terminal) that is currently eligible for entering or accepting messages.

AMERICAN STANDARD CODE FOR INFORMATION INTERCHANGE (ASCII):

This code, established by the American Standards Association, defines codes for a set of characters to be used in the interchange between business equipment over telephone and telegraph circuits. The code consists of 128 graphic and control characters.

APPLICATION PROGRAM:

A program that processes data and is unique to one type of application. It is sometimes referred to as a "user program".

RIGOOVRILOO SARIEWS CO	MMUNICATIONS MUDULE 12 Sep	14/8
Software Operational G	uide 📍	188
Section VII Glossary o	f Terms and Acronyms (Original Vers	ion)
TERM	DEFINITION	
AUTO CALL UNIT:	A device furnished by the communicatio common carrier which allows a User Sys to automatically establish a dialed li to a Server System over a communicatio network.	ns tem nk ins
BEGINNING OF JOB (BOJ):	The execution of a single program unit be performed by the system.	to
BINARY CARD:	A punched card with multiple punches i any one or more columns which represen non-ASCII character.	n nt a
8IT:	The smallest unit of information in a binary system.	
BITS PER SECOND (bps):	The expression bits-per-second is used express the speed at which bits of dat are transmitted over a communication channel or internally moved within a computer system.	i to :a
BLOCK:	A group of characters/bits transmitted a single unit of data. Also, see Bloo Check Character.	d as :k
BLOCK CHECK CHARACTER (BCC):	The Block Check Character is a charac appended to data blocks transmitted fo the purpose of error detection and par control.	ter br rity
BLOCKING FACTOR:	The maximum number of characters allo per single transmission.	ved
BROADCAST:	The simultaneous dissemination of information to a number of work stati (terminals).	ons

12 Sep 1978

Software Operational Guide

Page - 189

Section VII Glossary of Terms and Acronyms (Original Version)

TE	RM
----	----

DEFINITION

#### BUFFER:

This is temporary storage that is used during the transfer of data from one device to another. It is also used to compensate for variations in bit rates or timing of events within the system.

CARFIAGE CONTROL:

Controls the vertical paper motion for a line printer.

CHANNEL:

A device that directs the flow of data between a computer's main memory and input/output devices.

CHARACTER:

COMMON CARRIER:

(Letters can be upper or lower case.)

A cigit, letter, or special character .

A company, such as Bell Telephone, that provides communication lines as a public service.

COMMON CARRIER EXCHANGE:

A defined area, served by a communications carrier, within which the carrier provides service.

COMMUNICATION LINE:

Any medium, such as a wire or a telephone circuit, that connects a terminal with the computer.

CONDITIONED LINES:

These are private and leased transmission lines that are conditioned to reduce data distortion while providing transmission of data at lower error rates.

CONNECTION:

An established data transmission path between two or more computer systems. It is considered either a temporary or permanent connection, depending on whether it is established with/without switching facilities, respectively.

B18CO/B1700 SYSTEMS COMMUNICATIONS NODULE 12 Sep 1978 Page = 190 Software Operational Guide Section VII Glossary of Terms and Acronyms (Original Version) DEFINITION TERM CONTROL CARD: A punched card which contains information which identifies a job or describes its requirements to the MCP. The first column must contain an invalid character in 80 column cards, a question mark "?" in 96 column cards, or a rull character 2002 for remote iobs. CONTROL STATE: The term "control state", for this system, refers to a program that can assume control of the system's processor with privileged operands. As a rule, the type of control state program suggested here infers an "operating system" or MCP. An operational mode in which message data CONVERSATIONAL MODE: (full or partial messages) is used in lieu or, or in addition to, control characters as replies for message information. DATA: Data is any representation of information, such as characters, to which meaning might be assigned. Data communication is a method of DATA COMMUNICATIONS: transferring encoded data by means of electrical transmission between two or terminals and/or systems. DATA COMMUNICATIONS SYSTEM: A data communications system consists of two or more computer systems that are linked by means of a communication line(s) for the purpose of data exchange. DATA LINK: A composite of computer systems and the interconnecting network, together with control procedures, operating in a particular mode that permits information

to be exchanged between these systems.

Software Operational Guide

## 12 Sep 1978

Page - 191

Section VII Glossary of Terms and Acronyms

(Original Version)

#### TERM

#### DEFINITION

DATA SET:

A device that provides an interface between a data communications system and a data communications line. Most data communication lines use a modulated carrier for data transfer. The data set provides the logic to modify the data from being received or transmitted between terminals and the system.

DATA SET READY:

A signal provided by the Data Set.

DATA TRANSMISSION:

Data transmission consists of transmitting data between two or more computer systems in a system network.

DISCONNECT (DEOT):

The term (and mnemonic) for the "DLE EUT" communication control sequence which is used to signal that a "disconnect" of a switched circuit must be initiated.

SIAL-OUT:

The use of an Auto Call Unit to establish a "switched" connection between two (2) computers.

DIAL-UP:

The use of a dial or touchtone data set to establish a "switched" connection between two (2) computers.

DISK CARTRIDGE:

A secondary data storage device much the same as a "disk pack" but usually smaller in size. It can be moved on-line or off-line.

DISK DIRECTORY:

A disk-resident table that contains the name and type of file, together with a pointer to the disk file header or subdirectory for all permanent files which reside on disk.

DISK PACK:

A secondary data storage device that can be moved on-line or off-line.

B1800/B1700 SYSTEMS CO	MMUNICATIONS MODULE	12 Sep 1978
Software Operational G	uide	Page - 192
Section VII Glossary o	f Terms and Acronyms (	Original Version)
TERM	DEFINITION	
DISK-PACK-IDENTIFIER dp-id):	An identifier which is th to a disk pack or disk ca	e name assigned rtridge.
EXTENDED BINARY CODED DECIMAL INTERCHANGE		
CODE (EBCDIC):	A code consisting of 8-bi to represent 256 unique l and special characters.	ts which is used etters, numbers,
ERROR:	Any received character or characters that does not transmitted are considere	sequence of conform to those d to be in error.
ERROR RECOVERY PROCEDURES:	These are data communicat procedures used to restor operations to a data link has occurred.	ion control e normal after an error
FAMILY NAME:	A identifier used as a finame assigned to identify subdirectory entries.	ile name, or the a main file with
FILE SECURITY:	The procedures or special prevent access to or use programs without authoriz	l devices used to of data or zation.
FILE IDENTIFIER:	All disk file-identifiers system must be unique to file-names. A file ident comprised of any combinat following file-identifier	s used on the prevent duplicate tifier can be tion of the r options:
	<ul> <li>file-identifier</li> <li>family-name/file</li> <li>dp-id/family-name</li> </ul>	-identifier e/file-identifier
HALF-DUPLEX CHANNEL:	This is a communication the signaling may be in but not in both direction	channel in which either direction, ns simultaneously.

81	800/81700	SYSTEMS	COMMUNICATIONS	MODULE

12 Sep 1978

Software Operational Guide

Page - 193

Section VII Glossary of Terms and Acronyms (Original Version)

TERM

#### DEFINITION

in combination.

HALF-DUPLEX TRANSMISSION:

This is a type of data transmission where information is sent in only one direction at a time. It cannot send data in both directions simultaneously.

HARD COPY:

Generally, the printed copy of some material in a readable form is considered to be "hard copy". This would include computer printed media such as Reports, Listings, and Documents like this.

A word consisting of from one to seventeen

alphabetic, numeric, or special characters

**IDENTIFIER:** 

INTEGER:

A whole number as opposed to a fraction; that is, a number that contains the unit of (one) an exact number of times.

INTER-SYSTEM COMMUNICATIONS:

The ability for a terminal or program in one system to communicate with a program in another system.

INTRA-SYSTEM COMMUNICATIONS:

The restricted ability for a terminal or program in a system to communicate with another program in the same system.

JOB:

A "job" to this system is any program that is executable under the control of the MCP; SYCOM is a job.

JOB NUMBER:

A unique number assigned by the systems MCP to each job that is executed under its control.

B1800/B1700 SYSTEMS COMMUNICATIONS MODULE 12 Sep 1978 Software Operational Guide Page - 194 Section VII Glossary of Terms and Acronyms (Original Version) TERM DEFINITION -----\_ \_ \_ \_ \_ \_ سر من خبر من خبر خبر خبر خبر آن آن آن خبر خبر من من من من JOB SPAWNING: The ability for a user to execute jobs under program control and maintain an adequate level of control over these jobs during their execution. KE YBOARD: A device for the encoding of data by key depression, which causes the generation of the selected code element. LEASED LINE: A data transmission line reserved for the exclusive use of a customer (data center). A "like system" is a system with physical characteristics similar to those of a LIKE SYSTEM: target system. For a 81800/81700 system, a "like" system would be either a B1800 or 817001 A data communications line linking two (2) LINE: computers or a computer and its associated terminals. A line adapter consists hardware and/or LINE ADAPTER: software logic used to interface a data set or communications line into a multiline or single-line I/D control unit. LINE SWITCHING: The technique of temporarily connecting two communications lines together so that two computers and/or a computer and its associated terminals can exchange data. LOG-OFF: The procedure by which a user ends a data exchange session either between two computers or between a computer and a terminal. LOG-ON: The procedure by which a user establishes a data exchange session between two computers or a computer and a terminal.

<u>i</u> 1	DAULTONS SACIENC	COMMENTOATTONS	KODIUE
2,1	LCANNUTIAN DISICUS	COMMUNICATIONS	HUDULL

Software Operational Guide

Page - 195

12 Sep 1978

Section VII Glossary of Terms and Acronyms

(Original Version)

TERM

#### DEFINITION

LEGICAL STATION NUMBER

(LSN): The identification number assigned by the Network Controller to a given Station (Terminal). The Logical Station Number has no relationship to file names, but is determined by the order in which a station is defined in the Network Controller.

MASTER CONTROL PROGRAM

(MCP): The Master Control Program is the master "control state" program (operating system) used in the B1800/B1700 series computers for system control.

MESSAGE:

A sequence of characters arranged in a form suitable for the purpose of conveying information. A message contains the data to be conveyed (the text) and may, in addition, contain communication characters to aid in the routing or handling of the message.

MESSAGE FORMAT:

A message format defines the placement of message elements within the message text. These elements would be such items as; header, control characters, and text data.

MESSAGE HEADER:

Every message moved between a terminal and the SMCS, or an application program for that matter, is prefixed with a "header". This header contains station attributes necessary to identify the station when the time comes to return a reply message.

MESSAGE NUMBERING:

MODEM:

Message numbering is the sequencing of messages transmitted by the communications system to ensure all messages sent are in turn received.

Same as Data Set.

ottware Operational (		Page - 196
ection VII Glossary o	of Terms and Acronyms (O	riginal Version)
IERM	DEFINITION	
(MF-ID):	A generalization for a gro the same "family name" but file names.	up of files with with different
MULTI-LINE CONTROL (MLC):	The (MLC) is an I/O contro provides the functional co computer hardware and the	l device that introl between line adapters.
	The control is an integral multi-line communications	part of a sub-system.
NE TWORK:	A number of data communica connecting the User System work stations (terminals).	ition lines with remote
NETWORK CONTROLLER:	The NDL generated data com program used by the SYCOM	munication to interface
	with the work stations (te other systems.	erminals) and
LANGUAGE (NDL):	A descriptive, free-form I defining and implementing	anguage, for a data
	communications network. 1	The NDL compiler
	analyzes the input stateme generates a custom Networl	ents and Controller.
NORMAL STATE:	The term "normal state",	for this system,
	facilities to seize control	of a system's
	processor; as opposed to '	"control state".
	(e.g., SYCOM is a normal s	state program.)
OFFLINE:	There are two facets to th	his term for the
	STUUM SYSTEMJ L) a system	IS "OTTLINE"
	exists between it and a h	ost system, and
	2) a systems' terminal, Si	PO, or program is
	"offline" when it is not	logically
	connected to an application	on program in a
	птаг аузгана	

Software Operational Guide

12 Sep 1978

Page - 197

Section VII Glossary of Terms and Acronyms

(Original Version)

TERM

#### DEFINITION

ONLINE:

There are two meanings for this term in a SYCOM system; 1) a system is "online" when a communications link currently exists between it and a host system that provides processing services, and 2) a terminal, SPO, or program in a User System is "online" when it is logically connected to an application program in a host system. (The system must be online in order for a terminal, SPO, or program to be online.)

PARTICIPATE:

This term refers to a function in the Supervisory Message Control System (SMCS) which allows it to "participate" in all input/output message traffic between the Network Controller and an application program. In reference to a SYCOM system, the SMCS "is not permitted" to participate in SYCOM's message traffic.

PASSWORD:

A unique string of characters that a program, computer operator, or user must supply to meet security requirements before gaining access to data/files. The password is confidential as opposed to the "usercode".

POINT-TO-POINT CONNECTION:

This is a configuration in which a network connection is established between two, and only two, computers. The connection may include a switching facility.

PRIVATE LINE:

This would be a data communications line which has been furnished to the user for exclusive use.

PSEUDO CARD READER:

In a B1800/B1700 system, a pseudo card reader is in fact a cisk file. The disk file may be created via cards, program output, or via a terminal. When a program requires an input card file, the pseudo card reader can be substituted; as that file can be appropriately labeled.

B1800/B1700 SYSTEMS COM	IMUNICATIONS MODULE	12 Sep 1978
Software Operational Gu	ide	Page - 198
Section VII Glossary of	Terms and Acronyms	(Original Version)
TERN	DEFINITION	• • • • • • • • • • • • • • • • • • •
WUEUC #	transmission or process	ing.
REMOTE JOB:	Submission of control s	tatements and data
	causing the job describ	ed to be scheduled
	and executed at another	system.
REMOTE JOB ENTRYCRJE:	The Remote Job Entry pr	ogram is a method
	be which a remote compu	ter executes or
	compiles programs sent	to it by other insidered terminal
	devices.	
CONTROL STATION (RSC):	This is an input/output	device located on
	a User System (remote t	o a Server System)
	to provide the operator	with the ability
	to control jobs and to	request information
	trom the Server System.	
RJE/TERMINAL:	The official name for t	the software system
	capabilities.	SU ILS DASIC
RJE STATIUN:	FOR THIS SYSTEMP THE TE	orms "KJL Station" ouivalent. The new
	term is used to denote	a system with many
	additional capabilities	s and potential.
	therefore, the term "R.	JE Station" is no
	tonger appropriate.	
RELATIVE STATION		
NUMBER (RSN):	Is the position within	the Remote File
	Tamity assigned at get	1398 GLIVII LING¢
SEMANTICS:	The relationship betwee	en symbols and their
	<u>meantid</u> ⊅•	
SERVER SYSTEM:	A Server System is a p	assive system which
	provides computing ser	vices to the "User
	лания развания и с взя мая в формация и разва. Жайна	मध्य के साथ प्रमाण के मिन्द्र के सिंह क

1

12 Sep 1978

Software Operational Guide

Page - 199

Section VII Glossary of Terms and Acronyms (Original Version)

#### TERM DEFINITION

SESSION:

\_\_\_\_\_

The period of time during which a user engages in dialog between a terminal and a computer or between two computers. (The elapsed time betweer when the user logs-on the system and when the log-off occurs.

SESSION NUMBER:

A unique number assigned by a originating system to each session.

SINGLE LINE CONTROL (SLC):

This is an I/O control device that provides the functional control between a computers main frame and a line adapter. This control is an integral part of a single line communications sub-system.

SITE ID:

A unique 17-character identifier sent in a special message to a server system prior to being logged-on to provide a positive means of identification, independent of any other system security.

SPO:

The accepted term for the 81800/81700 systems "supervisory console" (VDT or Teletype). This term has been derived from the obsolete acronym for "Supervisory Printer Output". Usage in this document is for an abbreviation for User Spo.

SPO LOG:

A disk file that contains all input/output messages sent to and received from the User SPO.

SPOOLING:

STATION:

The reading and writing of input/output data streams on secondary storage devices. in a format convenient for later output operations and/or processing.

Same as "TERMINAL".

B1800/B1700 SYSTEMS C	OMMUNICATIONS MODULE	12 Sep 1970
Software Operational	Guide	Page - 200
Section VII Glossary	of Terms and Acronyms	(Original Version)
TERM	DEFINITION	
SWITCHED LINE:	A communications line connection between a L Server System is estat	in which a temporary lser System and a blished by dialing.
SWITCHING:	This is an operation is connecting of circuits establish a temporary between two or more sy	involving the inter- s in order to communication link ystems.
SYNCHRONOUS TRANSMISSION:	A data transmission pr sending and receiving	rocess in which the devices are
	the same frequency and means of correction, relationship.	d are maintained, by in a desired phase
SYSTEM DISK:	A disk pack or disk ca initialized as a systepack is under control or more must be present the MCP to function. is always considered	artridge that is em pack. A system of the MCP and one nt on the system for Head-per-track disk system disk.
TERMINAL:	This is an input/output both receive and send environment associated performed. Further, sending entries to, and from, the system of w	ut device designed to source data in an d with the job to be it is capable of nd obtaining output hich it is a part.
TEXT:	A sequence of charact information to be con	ers forming to veyed.
TR ANSMISSION:	Same as Data Transmis	sion.
TRANSMISSION BLOCK:	A transmission block part of a message blo block is independent	may contain all or ck. A transmission of the message(s)

B1800/B1700 SYSTEMS CO	MMUNICATIONS MODULE	12 Sep 1978
Software Operational G	uide	Page - 201
Section VII Glossary o	f Terms and Acronyms	(Original Version)
TERM	DEFINITION	
TRANSPARENT MODE:	An operational mode i combinations of eight allowed as message te	n which all coded -bit characters are xt characters.
UNIT DEVICE TDENTIFIER		
(udi):	The mnemonic identifi peripheral device att	er assigned to each ached to the system.
USER SPO:	The User System's Sup station (VDT).	ervisory Control
USER SYSIEM:	A user System is the system providing loca	active participant al service to the user
	wishing to utilize ar resource; locally or	other computing remotely.
USERCODE:	A symbol which consis eight characters to i	sts of from one to identify the user.
VIRTUAL TERMINAL:	A logical communication permits a terminal S	ions capability which SPOP or program in one
	system to "interact" program in another sy	' with an application stem.
V0 T •	Video Dicolay Termin	1 A video type york
	station (terminal) su	ich as the TD820 and
	IDCSU LEIBINAL GEVIC	
WAITING SCHEDULE:	Contains the names an	d run-time attributes
	the Active Schedule.	
ZIP:	A method by which a communicates with the specific MCP function	crogram such as SYCOM, E MCP for certain rs.

B1800/B1700 SYSTEMS COMMUNICATIONS	MODULE 12 Sep 1978
Software Operational Guide	Page - 202
Section VII Glossary of Terms and	Acronyms (Original Version)

## ACRENYMS

ACU	•	Auto Call Unit
AP	•	Application Program
ASCII		American Standard Code for Information Interchange
BCC	-	Block Check Character
80J	•	Beginning Of Job
bpi	-	Bits Per Second
DIS	-	Data Set
DDD	•	Direct Distance Dialing
dp-id		Disk Pack IDentifier
EOJ		End Of Job
1/0	•	Input/Output
lsn		Logical Station Number
MCP	•	Master Control Program
NC	-	Network Controller
NDL	-	Network Definition Language
NIF	-	Network Information File
RJE	-	Remote Job Entry
RSN	•	Relative Station Number
SMCS	• •	Supervisory Message Control System
SOG	•	Software Cperational Guide
SPO	-	Supervisory Printer Output
SYCOM		Systems Communications Module
udi		Unit Device Identifier
VDT	-	Video Display Terminal

# SANTA HARBARA PLANT

GOLETA+ CALIFORNIA

Address Comments Concerring

This Document To:

Dennis F. Ehrenberg Data Communications Group Computer Systems Division 630G Hollister Rd Goleta, California 93G17 805/964-6881

10/2\$/78 See Peri A

#### DOCUMENTATION ON ENHANCEMENT TO SYCOM

1) When SLAVE/BACKUP is running and the all backup files will be sent to the HUB and printed automatically. These backup files will then be removed from the sending system. Thus to print and remove backup . files no operator/user intervention is required.

If you do not want a backup file printed be sure to set the "NO ATP" attribute for the file.

Example:

CO HUB LSDL LI: FI LINE NO ATP; FI CARDS NAME HUB/SOURCE DSK;

The PUT, FETCH and COPY syntax has been extended as follows: 2)



The attribute LIST implies that the file will be printed at the receiving This attribute is applicable to Backup files only. end.

The attribute REMOVE will cause the file to be removed from the sending system.

The default attributes for Backup files are (LIST, REMOVE). The default attribute for non Backup file is SAVE. The LIST attribute does not apply to non Backup files.

The PUT, COPY, FETCH systax is unchanged except for the above additions.

Examples:

(SPO)/#4170 (SAVE) ONTO BACKUP.PRT/4170 .PUT

.PUT HUB (REMOVE) ONTO SYCOM3

.PUT HUB (US=SPO/X, REMOVE) AS XYZ
- Program to SYCOM communication has been fixed so that a program can initiate .CONNECT.
- Many changes have been made to the SYCOM request section to prevent garbage messages and lost messages.

5) Compile late 10 27 SLAVE BACKUP 10/25/25 or later SYCOM 10/24/78 or later Network controller A listing of SLAVE-BACKUP will be kept 6) on the system SLAVE3. Documentation de SLAUE/BACKUP options is in the history